#### SECURITIES AND EXCHANGE COMMISSION

## FORM 8-K

Current report filing

Filing Date: **2024-08-21** | Period of Report: **2024-08-16** SEC Accession No. 0001140361-24-037974

(HTML Version on secdatabase.com)

#### **FILER**

#### BridgeBio Pharma, Inc.

CIK:1743881| IRS No.: 000000000 | State of Incorp.:DE | Fiscal Year End: 1231

Type: 8-K | Act: 34 | File No.: 001-38959 | Film No.: 241226900

SIC: 2834 Pharmaceutical preparations

Mailing Address 3160 PORTER DR. SUITE 250 PALO ALTO CA 94304 Business Address 3160 PORTER DR. SUITE 250 PALO ALTO CA 94304 (650) 391-9740

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

#### FORM 8-K

#### **CURRENT REPORT**

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): August 16, 2024

## BridgeBio Pharma, Inc.

(Exact name of registrant as specified in its charter)

Delaware
(State or other jurisdiction of
incorporation)

per share

001-38959 (Commission File Number) 84-1850815 (IRS Employer Identification No.)

3160 Porter Dr., Suite 250
Palo Alto, CA
(Address of principal executive offices)

94304 (Zip Code)

(650) 391-9740 (Registrant's telephone number, including area code)

Not Applicable (Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):			
☐ Written communications pursuant	to Rule 425 under the Securities	s Act (17 CFR 230.425)	
Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
Securities registered pursuant to Section	n 12(b) of the Act:		
Title of each class	Trading Symbol(s)	Name of each exchange on which registered	
Common Stock, par value \$0.001	BBIO	The Nasdag Global Select Market	

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§ 240.12b-2 of this chapter).	
Emerging growth company □	
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.□	
	•

#### Item 1.01. Entry into a Material Definitive Agreement.

On August 16, 2024, BridgeBio Pharma, Inc. (the "Company"), on the recommendation of a special committee of independent and disinterested directors of the Company, entered into a transaction agreement (the "Transaction Agreement") providing for the formation and funding by certain third party investors (the "Investors") of GondolaBio, LLC, a Delaware limited liability company ("GondolaBio"), as a joint venture for the purpose of researching, developing, manufacturing and commercializing pharmaceutical products, including those contributed to GondolaBio by the Company (the "Transaction"). The Investors providing financing to GondolaBio consist of an investor syndicate, including Viking Global Investors LP, Patient Square Capital, Sequoia Capital, Frazier Life Sciences, Cormorant Asset Management, Aisling Capital and an entity owned by Neil Kumar, the Company's Chief Executive Officer. The Transaction was consummated (the "Closing") on the date of signing the Transaction Agreement.

Pursuant to the Transaction Agreement, the Company contributed to GondolaBio certain early-stage clinical and pre-clinical stage programs in the areas of Erythropoietic Protoporphyria, Alpha-1 Antitrypsin Deficiency, and Tuberous Sclerosis Complex. The Investors have committed \$300 million of tranched financing to GondolaBio. The Company's initial interest in GondolaBio will be approximately 45% and will be subject to reduction as additional tranches of capital contributions are funded.

Also on August 16, 2024, BridgeBio Pharma LLC, a wholly owned subsidiary of the Company, became a party to the Amended and Restated Limited Liability Company Agreement of GondolaBio (the "LLC Agreement"). The LLC Agreement sets forth, among other things, the economic and governance rights of the members of GondolaBio, including governance rights, economic preferences, privileges, restrictions and obligations of the members.

The foregoing agreements and the transactions were approved and recommended to the Board of Directors of the Company by a special committee of independent and disinterested directors of the Company, with the assistance of Goldman Sachs & Co. LLC as financial advisor to the special committee, Goodwin Procter LLP and Richards, Layton & Finger P.A. as legal advisors to the special committee, and Skadden, Arps, Slate, Meagher & Flom LLP as legal advisor to the Company.

The foregoing summary does not purport to be complete and is qualified in its entirety by reference to the full text of the Transaction Agreement and the LLC Agreement, copies of which are filed as Exhibit 10.1 and 10.2, respectively, to this Current Report on Form 8-K and are incorporated herein by reference.

#### Item 9.01. Financial Statements and Exhibits.

Description

#### (d) Exhibits.

E-bibit

EXHIDIU	Description
<u>10.1</u>	Transaction Agreement, dated as of August 16, 2024, by and among BridgeBio Pharma, Inc., Viking Global Opportunities
	Illiquid Investments Sub-Master LP, Viking Global Opportunities Drawdown (Aggregator) LP, Patient Square Bravo
	Aggregator, LP, SC US/E GROWTH FUND X MANAGEMENT, L.P., SC US/E Venture Fund XVIII Management, L.P.,
	Frazier Life Sciences XI, L.P., Frazier Life Sciences Public Fund, L.P., Frazier Life Sciences Public Overage Fund, L.P.,
	Cormorant Private Healthcare Fund IV, LP, Cormorant Private Healthcare Fund V, LP, Cormorant Global Healthcare Master
	Fund, LP, Aisling V Bridge Splitter LP, Kumar Haldea Revocable Trust and GondolaBio, LLC.*
<u>10.2</u>	Amended and Restated Limited Liability Company Agreement of GondolaBio, LLC, dated as of August 16, 2024*
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

<sup>\*</sup>Certain of the exhibits and schedules to this Exhibit have been omitted in accordance with Regulation S-K Item 601(a)(5).

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

BridgeBio Pharma, Inc.

Date: August 20, 2024 By: /s/ Brian C. Stephenson

Brian C. Stephenson, Ph.D., CFA

#### **EXECUTION VERSION**

#### TRANSACTION AGREEMENT

dated as of August 16, 2024

by and among

BRIDGEBIO PHARMA, INC., VIKING GLOBAL OPPORTUNITIES ILLIQUID INVESTMENTS SUB-MASTER LP, VIKING GLOBAL OPPORTUNITIES DRAWDOWN (AGGREGATOR) LP, PATIENT SQUARE BRAVO AGGREGATOR, LP, SC US/E GROWTH FUND X MANAGEMENT, L.P., SC US/E VENTURE FUND XVIII MANAGEMENT, L.P., FRAZIER LIFE SCIENCES XI, L.P., FRAZIER LIFE SCIENCES PUBLIC FUND, L.P., FRAZIER LIFE SCIENCES PUBLIC OVERAGE FUND, L.P., CORMORANT PRIVATE HEALTHCARE FUND IV, LP, CORMORANT PRIVATE HEALTHCARE FUND V, LP, CORMORANT GLOBAL HEALTHCARE MASTER FUND, LP, AISLING V BRIDGE SPLITTER LP, KUMAR HALDEA REVOCABLE TRUST, PATIENT SQUARE BRAVO AGGREGATOR, LP, VIKING GLOBAL OPPORTUNITIES ILLIQUID INVESTMENTS SUB-MASTER LP & VIKING GLOBAL OPPORTUNITIES DRAWDOWN (AGGREGATOR) LP, as Representatives of the **Investors** 

> and BRIDGEBIOX PHARMA, LLC

#### TABLE OF CONTENTS

		<u>Page</u>
	ARTICLE I	
	DEFINITIONS	
Section 1.1	Definitions	2
	ARTICLE II	
	PRE-CLOSING RESTRUCTURING	
Section 2.1	Pre-Closing Restructuring	15
Section 2.2	Non-Assignment of Certain Acquired Assets	16
	ARTICLE III	
	CLOSING	
Section 3.1	Closing	17
Section 3.2	Other Closing Deliveries	18
Section 3.3	Withholding Rights	19
	ARTICLE IV	
	CERTAIN REPRESENTATIONS AND WARRANTIES	
	OF BRIDGEBIO	
Section 4.1	Organization	19
Section 4.2	Power and Authorization; Enforceability	20
Section 4.3	No Prior Operations or Liabilities	21
Section 4.4	Consents and Approvals; No Conflicts	21
Section 4.5	Acquired Subsidiary	22
Section 4.6	Assets; Sufficiency of Assets	22
Section 4.7	Litigation	22
Section 4.8	Compliance with Laws	22
Section 4.9	Permits	23
Section 4.10	Taxes	23
Section 4.11	Opinion of Financial Advisor; Special Committee Recommendation; BridgeBio Board Approval	24
Section 4.12	Brokers	25
Section 4.13	Absence of Changes	25
Section 4.14	Financial Statements; No Undisclosed Liabilities	25
Section 4.15	Intellectual Property	26
Section 4.16	Industry Regulatory Matters	27
Section 4.17	Employee Benefits	27
Section 4.18	Labor and Employment	28
Section 4.19	In-Scope Contracts	29
Section 4.20	Affiliate Arrangements	30

Section 4.21	Sanctions	30
Section 4.22	No Other Representations or Warranties	30
	ARTICLE V	
	CERTAIN REPRESENTATIONS AND WARRANTIES OF INVESTORS	
Section 5.1	Organization	31
Section 5.2	Power and Authorization; Enforceability	31
Section 5.3	Consents and Approvals; No Conflicts	31
Section 5.4	Compliance with Laws	32
Section 5.5	Litigation	32
Section 5.6	Availability of Funds	32
Section 5.7	Qualified Client; Accredited Investor	32
Section 5.8	Investment Experience and Ability to Bear Risk; Suitability; Disclosure of Information	32
Section 5.9	Transfers and Transferability	33
Section 5.10	Publicly Traded Company	34
Section 5.11	No Investment Company Issues	34
Section 5.12	Certain ERISA Matters	34
Section 5.13	No Intent to Distribute	35
Section 5.14	Sanctions	35
Section 5.15	Residence	35
Section 5.16	Certain Regulatory Matters	35
Section 5.17	Brokers	35
Section 5.18	No Other Representations or Warranties	35
	ARTICLE VI	
	ADDITIONAL AGREEMENTS	
~		
Section 6.1	Further Assurances; Cooperation	36
Section 6.2	Insurance	36
Section 6.3	Tax Treatment	36
Section 6.4	Transfer Taxes	37
Section 6.5	Tax Returns	37
Section 6.6	Intercompany Accounts and Arrangements	37
Section 6.7	Director and Officer Indemnification	37
Section 6.8	Shared Contracts	38
Section 6.9	Misallocated Assets; Third-Party Claims	39
Section 6.10	Employee Matters	40
Section 6.11	Transitional Trademark License	42
Section 6.12	Intellectual Property Cross-Licenses	43
Section 6.13	Non-Solicitation	43
Section 6.14	Non-Competition	44
Section 6.15	Release	45
Section 6.16	Prepaid Expenses	46

ii

	ARTICLE VII	
	SURVIVAL; INDEMNIFICATION	
Section 7.1	Survival	46
Section 7.2	Indemnification by BridgeBio	46
Section 7.3	Indemnification by the Company	46
Section 7.4	Limitations on Indemnification	46
Section 7.5	Procedure	47
Section 7.6	Exclusive Remedies	48
	ARTICLE VIII	
	MISCELLANEOUS	
Section 8.1	Expenses	48
Section 8.2	Amendment; Waiver	49
Section 8.3	Assignment; Successors and Assigns	49
Section 8.4	No Third-Party Beneficiaries	49
Section 8.5	Governing Law; Venue; Waiver of Jury Trial	49
Section 8.6	Specific Performance	50
Section 8.7	Severability	50
Section 8.8	Counterparts; Electronic Delivery	51
Section 8.9	Descriptive Headings; Interpretation	51
Section 8.10	Schedules	52
Section 8.11	Notices	52
Section 8.12	Entire Agreement	56
Section 8.13	Public Announcements	56
Section 8.14	Investors' Representative	57
Section 8.15	Privileges; Conflicts	58
Section 8.16	Non-Recourse	60

iii

#### INDEX OF EXHIBITS, SCHEDULES AND ANNEXES

<b>Exhibits</b>	
Exhibit A	Form of Amended and Restated LLC Agreement
Exhibit B	Certificate of Formation
Exhibit C	Transition Services Agreement
Exhibit D	Form of Pre-Closing Assignment and Assumption Agreement
Exhibit E	Form of Intercompany Termination Agreement
Schedules	
Schedule 1.1(a)	Acquired Subsidiaries
Schedule 1.1(b)	In-Scope Programs
Schedule 1.1(c)	In-Scope Program Employees
Schedule 1.1(d)	In-Scope Scientific Equipment
Schedule 1.1(e)	In-Scope Regulatory Approvals
Schedule 1.2(a)	Knowledge of BridgeBio
Schedule 1.2(b)	Knowledge of Investors
Schedule 3.1	Initial Company Interests
<u>Annexes</u>	
Annex A	Index of Defined Terms
BridgeBio Disclosure Letter	

iv

#### TRANSACTION AGREEMENT

This TRANSACTION AGREEMENT (this "Agreement"), dated as of August 16, 2024, is made by and among Viking Global Opportunities Illiquid Investments Sub-Master LP, Viking Global Opportunities Drawdown (Aggregator) LP, Patient Square Bravo Aggregator, LP, SC US/E GROWTH FUND X MANAGEMENT, L.P., SC US/E Venture Fund XVIII Management, L.P., Frazier Life Sciences XI, L.P., Frazier Life Sciences Public Fund, L.P., Frazier Life Sciences Public Overage Fund, L.P., Cormorant Private Healthcare Fund IV, LP, Cormorant Private Healthcare Fund V, LP, Cormorant Global Healthcare Master Fund, LP, Aisling V Bridge Splitter LP, Kumar Haldea Revocable Trust (each, an "Investor" and collectively, the "Investors"), BridgeBio Pharma, Inc., a Delaware corporation ("BridgeBio" and together with the Investors, the "Members" and each a "Member"), Patient Square Bravo Aggregator, LP, Viking Global Opportunities Illiquid Investments Sub-Master LP and Viking Global Opportunities Drawdown (Aggregator) LP, solely in their capacity as Investors' representatives (each, an "Investors' Representative" and, collectively, the "Investors' Representatives"), and BridgeBioX Pharma, LLC, a Delaware limited liability company (the "Company"). Each of BridgeBio, the Investors, the Investors' Representatives and the Company are referred to herein as a "Party" and, collectively, the "Parties." Capitalized terms used but not otherwise defined herein shall have the respective meanings ascribed to such terms in Article I.

WHEREAS, pursuant to the terms and subject to the conditions set forth in this Agreement, the Members desire to make certain contributions to the Company and operate the Company as a joint venture for the purpose of researching, developing, manufacturing and commercializing pharmaceutical products, including those within the In-Scope Programs;

WHEREAS, the BridgeBio Board has established the Special Committee and delegated to the Special Committee authority to, among other things, consider, review, evaluate and negotiate (on behalf of BridgeBio) with the Investors the terms of such potential transaction (including corporate opportunity waiver related thereto) and alternatives thereto and to make a recommendation to the full BridgeBio Board as to whether BridgeBio should enter into such potential transaction;

WHEREAS, the Special Committee has (a) determined that the terms of this Agreement and the transactions contemplated hereby are fair to and in the best interests of BridgeBio and its stockholders and (b) resolved to recommend that the BridgeBio Board (i) declare advisable the execution, delivery and performance of this Agreement and the transactions contemplated hereby and (ii) approve this Agreement and the transactions contemplated hereby (the "Special Committee Recommendation");

WHEREAS, the BridgeBio Board, acting upon the Special Committee Recommendation, has (a) determined that the terms of this Agreement and the transactions contemplated hereby are fair to and in the best interests of BridgeBio and its stockholders, and (b) approved and declared advisable the execution, delivery and performance of this Agreement and the transactions contemplated hereby;

1

WHEREAS, in connection with the foregoing, BridgeBio formed the Company on June 5, 2024 as a Delaware limited liability company pursuant to and in accordance with the Delaware Limited Liability Company Act (as amended from time to time, the "Act") and entered into the original limited liability company agreement of the Company as of June 5, 2024 (the "Original LLC Agreement");

WHEREAS, as of the date hereof, BridgeBio has, and has caused its Subsidiaries to, effect the Pre-Closing Restructuring pursuant to Section 2.1 (the "Pre-Closing Restructuring"), pursuant to which, among other things, the Acquired Assets and Assumed Liabilities have been transferred to or assumed by, as applicable, an In-Scope Entity (to the extent not already owned by such In-Scope Entity) or Sub23, Inc., a Delaware corporation and a wholly owned Subsidiary of BridgeBio formed in connection with the Pre-Closing Restructuring, as applicable (such formed Subsidiary, the "Formed Entity" and together with the In-Scope Entities, the "Acquired Subsidiaries"); and

WHEREAS, pursuant to the terms and subject to the conditions set forth in this Agreement, at the Closing: (a) after giving effect to the Pre-Closing Restructuring, BridgeBio will assign and transfer (or cause to be assigned and transferred) the capital stock of the Acquired Subsidiaries owned by BridgeBio or any of its Subsidiaries to the Company, (b) each Investor will make cash contributions to the Company at Closing and commit to make future cash contributions to the Company and (c) the Parties will amend and restate the Original LLC Agreement in its entirety by entering into the Amended and Restated Limited Liability Company Agreement of the Company in substantially the form attached hereto as <a href="Exhibit A">Exhibit A</a> (the "Amended and Restated LLC Agreement"); and (d) the Investors will be admitted as members of the Company in connection therewith.

**NOW, THEREFORE**, in consideration of the mutual promises and covenants made herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the undersigned, intending to be legally bound, hereby agree as follows:

#### ARTICLE I DEFINITIONS

#### Section 1.1 Definitions.

"Acquired Assets" means the following assets, rights and interests of BridgeBio and its Subsidiaries (but, for the avoidance of doubt, excluding the Excluded Assets):

- (a) the In-Scope Program Patents, In-Scope Program Know-How and In-Scope Program Other IP ("Acquired IP Assets")
- (b) all IT Assets owned or purported to be owned by BridgeBio or any of its Subsidiaries and exclusively used or held for use in connection with each In-Scope Program ("Acquired IT Assets");
- (c) all scientific equipment and the inventory materials contained therein, and other tangible materials consisting of cell lines, animal models and compound inventory, in each case of the foregoing, that is exclusively used in connection with any of the In-Scope Programs (and which scientific equipment is set forth on Schedule 1.1(d));

- (d) the In-Scope Program Contracts;
- (e) any rights and benefits under Shared Contracts to the extent allocated or assigned to or retained by the Company or any of its Subsidiaries pursuant to Section 6.8;
- (f) (A) all current and historical books, records, files and physical materials of any type, including supplier and consultant lists, distribution lists, development plans and life cycle management data, correspondence (including e-mails), Transferred Employee Records (to the extent not prohibited by applicable Law) (but excluding all other personnel records), financial records, audit reports, batch records, and laboratory notebook records (in all cases, in any form or medium, including electronic or physical) (collectively, "Records"), in each case, to the extent exclusively related to the In-Scope Programs, Acquired Assets or Assumed Liabilities; provided that BridgeBio shall have the right to redact or otherwise extract any information from the Acquired Records that is not in any way related to the In-Scope Programs, Acquired Assets or Assumed Liabilities; provided that BridgeBio may retain copies of any such Records for bona fide business purposes (collectively, the "Acquired Records");
- (g) all rights to causes of action, lawsuits, judgments, claims, counterclaims, rights of recovery and demands to the extent related to or arising from any of the In-Scope Programs, any Acquired Asset or any Assumed Liability (including rights to obtain damages, injunctive relief, and any other remedies for past, present, and future infringement, misappropriation, or other violation thereof);
- (h) all (A) applications, submissions, registrations or notifications submitted to a Regulatory Authority for purposes of obtaining, updating or maintaining of any Regulatory Approval to the extent exclusively relating to any In-Scope Program, (B) correspondence with or to Regulatory Authorities (including minutes and official contact reports relating to any communications with any Regulatory Authorities) to the extent exclusively related to any In-Scope Program, (C) non-clinical, clinical and other files, writings, notes, studies, reports and other documents contained or referenced in or supporting any of the foregoing or any Regulatory Approval, in each case, that were acquired, developed, compiled, collected or generated by BridgeBio or any of its Subsidiaries, or by any Third Party on behalf of BridgeBio or any of its Subsidiaries, in each case, to the extent exclusively related to any In-Scope Program, and (D) all regulatory or legal rights in any of clauses (A) to (C) (collectively, the "Acquired Regulatory Documentation");
  - (i) all Permits primarily used or held for use in connection with the In-Scope Programs, to the extent transferable;
- (j) any assets primarily related to the In-Scope Programs conducted by Portal and Sub21 that are reflected on the "Total assets" line item of the Financial Statements for each of Portal or Sub21 or that arise after the date of the Financial Statements and are of a type that would be reflected in the "Total assets" line item on a balance sheet of Portal or Sub21 in the Financial Statements had they existed at June 30, 2024, but excluding (A) the Excluded Intercompany Receivables (such assumed intercompany receivables (the "Assumed Intercompany Receivables") and (B) Cash; and

(k) all Contracts with respect to each Acquired Subsidiary Plan.

"Acquired Subsidiary Plan" means each Employee Plan that is sponsored or maintained solely by an Acquired Subsidiary for the benefit of In-Scope Program Employees or other current or former individual service providers of an Acquired Subsidiary.

"Affiliate" means, in respect of a Person, any entity or person which directly or indirectly controls, is controlled by or is under common control with that Party and for these purposes the term "control" shall mean (a) the right to exercise or cause the exercise of fifty percent (50%) or more of the voting rights of such entity or person, or (b) the right to exercise management control by agreement; provided, that, after the Closing, the Company or any of its Subsidiaries (including the Acquired Subsidiaries) shall not be deemed or treated as an Affiliate of BridgeBio.

"Assumed Liabilities" means the following (and only the following) Liabilities of BridgeBio or its Subsidiaries, in each case whether accrued or unaccrued, fixed, known or unknown, liquidated or unliquidated, absolute or contingent, matured or determined or determinable, but in each case, excluding the Excluded Liabilities:

- (i) any Liabilities to the extent related to or arising out of the ownership or use of any Acquired Asset, to the extent arising from facts, conditions or circumstances first existing or first occurring on or after the Closing;
- (ii) any Liabilities to the extent related to or arising out of any In-Scope Program Contract, to the extent arising from facts, conditions or circumstances first existing or first occurring on or after the Closing;
- (iii) any Liabilities relating to Shared Contracts to the extent transferred to the Company or one of the Acquired Subsidiaries pursuant to <u>Section 6.8</u> and to the extent arising from facts, conditions or circumstances first existing or first occurring on or after the Closing;
- (iv) any Liabilities, up to a cap of \$1,200,000, to the extent incurred after June 30, 2024 but prior to the Closing that would be reflected in the "Total liabilities" line item of the balance sheet for either of Portal or Sub21 prepared in accordance with GAAP (the "Assumed Balance Sheet Liabilities") (including any Liabilities of each of Portal and Sub21 to the extent incurred after June 30, 2024 but prior to the Closing for (A) accounts payable ("Assumed Accounts Payable") and (B) intercompany payables owed to BridgeBio or its Subsidiaries (other than the Acquired Subsidiaries) excluding the Excluded Accounts Payable ("Assumed Intercompany Payables"));
- (v) any Taxes payable by the Acquired Subsidiaries other than any Taxes described in <u>clauses</u> (ix) or (x) of the definition of Excluded Liabilities;
- (vi) except as otherwise expressly set forth in <u>Section 6.10</u>, all Liabilities relating to, arising out of or in connection with the In-Scope Program Employees who become Transferred Employees, to the extent arising from facts, conditions or circumstances first existing or first occurring on or after the Transfer Date; and

(vii) all Liabilities relating to, arising out of or in connection with the Acquired Subsidiary Plans.

"BridgeBio Board" means the Board of Directors of BridgeBio.

"BridgeBio Material Adverse Effect" means any change, event, development, effect or occurrence (each, an "Effect") that, individually or in the aggregate, has, or would reasonably be expected to have, a material adverse effect on (a) the business, financial condition, assets, liabilities, or results of operation of the Acquired Subsidiaries, In-Scope Programs, the Acquired Assets or the Assumed Liabilities, taken as a whole, or (b) the ability of BridgeBio or the Company to perform its obligations under this Agreement or the Transaction Documents or the consummation by BridgeBio or the Company of the transactions contemplated hereby; provided that, for purposes of clause (a), none of the following, and no Effect that results from or arises in connection with the following, either alone or in combination, shall be deemed to constitute a BridgeBio Material Adverse Effect or be taken into account in determining whether a BridgeBio Material Adverse Effect has occurred: (i) general economic or regulatory, legislative or political conditions (or changes therein), including any actual or potential stoppage, shutdown, disputed election, default or similar event or occurrence affecting a national or federal government, or securities, credit, banking, financial or other capital markets conditions (including changes generally in prevailing interest rates, currency exchange rates, credit markets or equity price levels or trading volumes), in each case, in the United States or elsewhere in the world, (ii) conditions (or changes therein) generally effecting the industries in which BridgeBio and its Subsidiaries (in each case, with respect to the In-Scope Programs) operate, (iii) any change in applicable Law or GAAP, (iv) geopolitical conditions, the outbreak or escalation of hostilities, any acts or threats of war (whether or not declared), sabotage, terrorism, or any escalation or worsening of any of the foregoing, (v) any epidemic, pandemic, disease outbreak or other public health-related event ((including the COVID-19 pandemic) (or escalation or worsening of any such events or occurrences, including, in each case, the response of Governmental Authorities)), hurricane, tornado, flood, fire, volcano, earthquake or other natural disaster, (vi) the failure, in and of itself, of BridgeBio or any of its Subsidiaries (in each case solely with respect to the In-Scope Programs) to meet any internal or external projections, forecasts, estimates or predictions before, on or after the date hereof (it being understood that the underlying facts giving rise or contributing to such failure or change may be taken into account in determining whether there has been a BridgeBio Material Adverse Effect if such facts are not otherwise excluded under this definition), (vii) any change in BridgeBio's stock price or trading volume (but not underlying facts to the extent not otherwise excluded herein), (viii) any change resulting directly from the announcement, pendency or performance of this Agreement and the transactions contemplated hereby, including the identity of the Investors or their respective Affiliates (provided that this clause (viii) shall not apply with respect to any representation or warranty that is intended to address the consequences of the execution and delivery of this Agreement or the announcement, pendency or consummation of the transactions contemplated hereby), (ix) actions taken as expressly required by this Agreement, (x) any action taken by BridgeBio, the Company or any of its Subsidiaries at the Investors' written request, except in the case of clause (i), (ii), (iii), (iv) or (v), to the extent that the In-Scope Programs, taken as a whole, are materially and disproportionately affected thereby as compared with other participants in the industries or markets in which BridgeBio and its Subsidiaries (in each case, with respect to the In-Scope Programs) operate (in which case the incremental disproportionate impact or impacts may be taken into account in determining whether there has been a BridgeBio Material Adverse Effect).

"Business Day" means any day that is not a Saturday or Sunday or other day on which banks in the State of New York or California are authorized or obligated to be closed.

"Certificate of Formation" means the certificate of formation of the Company filed with the Secretary of State of the State of Delaware on June 5, 2024, in the form attached hereto as Exhibit B, as may be amended or restated by the Members in accordance with the LLC Agreement and the Act from time to time.

"Closing Date" means the date on which the Closing occurs.

"Code" means the U.S. Internal Revenue Code of 1986, as amended.

"Company Interest" means a membership interest, in the form of Units, of the Company issued to a Member.

"Consent" means any approval, consent, ratification, permission, waiver or authorization from any Person other than a Governmental Authority.

"Contract" means any contract, agreement, click-through terms, purchase order, modification, obligation, instrument, promise, commitment, undertaking or arrangement (in each case, whether written, electronic or oral) that is or purports to be legally binding.

"Control" means the right or ability to grant a license (other than through a license granted pursuant to this Agreement or any other Transaction Document) of, or use of or access to (other than use of or access to pursuant to this Agreement or any Transaction Document), specified Intellectual Property in each case without violating the terms of any agreement or other arrangement with any Third Party.

"COVID-19" means SARS-CoV-2 or COVID-19, and any variants or evolutions thereof or epidemics, pandemics or disease outbreaks thereof.

"Employee Plan" means any "employee benefit plan" (as defined in section 3(3) of ERISA, whether or not subject to ERISA), and each other agreement, plan, program, fund, policy, Contract or arrangement (whether written or unwritten) providing compensation or benefits, including employment, individual consulting, pension, retirement, profit sharing, equity or equity-based, stock bonus, stock option, stock purchase, phantom or stock equivalent, bonus, incentive, commission, deferred compensation, vacation, life insurance, death benefit, sick pay, disability, severance, termination, separation, seniority pay, holiday pay, fringe benefit or other benefit or compensation plan, agreement, program, arrangement or policy (other than any of the foregoing that is statutorily required to be maintained by a Governmental Authority) that is, in each case, maintained, sponsored, contributed to or required to be contributed to by BridgeBio or any of its Affiliates or with respect to which BridgeBio or any of its Affiliates has any Liability.

"Encumbrance" means any lien, license, encumbrance, security interest, pledge, mortgage, easement, deed of trust, option, right of way, encroachment, conditional sale agreements and restrictions, right of first option or right of first refusal, preemptive rights, dragalong right, hypothecation, restriction on transfer of title or voting or similar restrictions on the full ownership and possession of a given asset, and any other claims, encumbrances or restrictions that have the same or a similar effect to the granting of security interest in such asset, whether imposed by Contract, Law, equity or otherwise, other than restrictions on transfer under applicable securities law.

"Equity Award Holder" means a holder of Equity Awards as of immediately prior to the Closing.

"Equity Awards" means, collectively, (i) the outstanding and unexercised options (whether vested or unvested) to purchase shares of common stock of Portal Therapeutics, Inc. and other equity-based awards relating to common stock of Portal Therapeutics, Inc. and (ii) the outstanding and unexercised options (whether vested or unvested) to purchase shares of common stock of Sub21, Inc. and other equity-based awards relating to common stock of Sub21, Inc., in each case, whether granted under a Stock Plan or otherwise.

"Excluded Liabilities" means any and all Liabilities of BridgeBio or its Subsidiaries or otherwise related to the Company or any Acquired Subsidiary, whether accrued or unaccrued, fixed, known or unknown, liquidated or unliquidated, absolute or contingent, matured or unmatured or determined or determinable as of the Closing Date, that are not Assumed Liabilities, including, the following:

- (i) any Liabilities to the extent related to or arising out of any Excluded Asset;
- (ii) any Liabilities of BridgeBio under this Agreement and the other Transaction Documents;
- (iii) any Liabilities to the extent related to or arising out of any In-Scope Program, to the extent arising from facts, conditions or circumstances existing or occurring prior to the Closing (including the Pre-Closing Restructuring) (other than Assumed Balance Sheet Liabilities);
- (iv) any Liabilities to the extent related to or arising out of any In-Scope Program Contract, to the extent arising from facts, conditions or circumstances existing or occurring prior to the Closing (other than Assumed Balance Sheet Liabilities);
- (v) any Liabilities arising out of or related to any Employee Plan, including the provision of continuation coverage under Part 6 of Subtitle B of Title I of ERISA, Section 4980B of the Code and any similar state Law ("COBRA") to all "M&A qualified beneficiaries" as defined in Treasury Regulation Section 54.4980B-9, Q&A-4(a), other than any Liabilities arising out of or related to any Acquired Subsidiary Plan;
- (vi) any Liabilities relating to Shared Contracts to the extent not transferred to the Company or one of the Acquired Subsidiaries pursuant to Section 6.8;

- (vii) any Liabilities to the extent reflected in the "Total liabilities" line item of the Financial Statements for either Portal or Sub21 that are not Assumed Balance Sheet Liabilities (including any Liabilities for accounts payable that are not Assumed Accounts Payable ("Excluded Accounts Payable") and any Liabilities for intercompany payables that are not Assumed Intercompany Payables);
- (viii) (A) any Liabilities, including those arising out of or related to the employment, service or retention, or the termination thereof, of any current or former employees, directors or independent contractors or service providers of BridgeBio or its Affiliates, who are not Transferred Employees, (B) any Liabilities related to any In-Scope Program Employees who do not become Transferred Employees, (C) any Liabilities related to Transferred Employees arising prior to the Transfer Date; and (D) any Liabilities assumed by BridgeBio or its Subsidiaries related to any employee leasing arrangement set forth in the Transition Services Agreement;
  - (ix) any Taxes payable by BridgeBio or its Subsidiaries (other than the Acquired Subsidiaries); and
- (x) any Taxes payable by, or with respect to, the Acquired Subsidiaries with respect to a Pre-Closing Tax Period (including any Tax of any Person under Treasury Regulations Section 1.1502-6 (or similar provision of state, local or non-U.S. Law), by contract (other than contracts entered into in the ordinary course of business, a principal purpose of which is not related to Taxes), as a transferee or successor, or otherwise).

"FDA" means the U.S. Food and Drug Administration.

"FDCA" means the Food, Drug and Cosmetic Act (21 U.S.C. §§ 301 et seq.), as amended.

"Financing Agreement" means that certain Financing Agreement, dated as of January 17, 2024, by and among BridgeBio, certain of its Subsidiaries and the other parties thereto.

"Fraud" means the making by a Party, to another Party, of an intentional misrepresentation with respect to any representation or warranty contained in Article IV or Article V of this Agreement by the first such Party that, at the time such representation or warranty was made by such Party, (a) such representation or warranty was inaccurate, (b) such Party had actual knowledge (meaning without imputed or constructive knowledge) that such representation or warranty was inaccurate, (c) in making such representation or warranty any Person with knowledge of the inaccuracy thereof had the intent to deceive such other Party and to induce such other Party to enter into this Agreement and (d) such other Party acted in reliance on such representation or warranty in entering into this Agreement and suffered or incurred damages, injury, losses or costs as a result of such reliance. "Fraud" does not include equitable fraud, promissory fraud, unfair dealings fraud or any torts (including fraud) based on negligence or recklessness, and only the Person or Persons who committed the Fraud shall be responsible for such Fraud and only to the Party established to have suffered from such Fraud.

"GAAP" means, as of any date of determination, generally accepted accounting principles in the United States as of such date, consistently applied.

"Governmental Authority" means any supranational, federal, foreign, national, state, county, local, municipal or other governmental, legislative, judicial, regulatory or administrative authority, agency, commission or other instrumentality, including any Tax Authority, any court, tribunal or arbitrator or arbitral body (public or private) with competent jurisdictions, and any of their respective subdivisions, agencies, instrumentalities, authorities or tribunals.

"Governmental Order" means any order, writ, judgment, ruling, injunction, decree, directive, stipulation, determination or award entered, rendered, issued or made by any Governmental Authority.

"Health Laws" means any Law regarding health care products applicable to the Company and the Acquired Subsidiaries the purpose of which is to ensure the safety, efficacy and quality of medicines or pharmaceuticals by regulating the research, development, manufacturing or distribution of these products, including, as applicable (i) the FDCA and the regulations promulgated thereunder (including Laws relating to Good Laboratory Practices, Good Clinical Practices, investigational use, product marketing authorization, manufacturing facilities compliance and approval, Good Manufacturing Practices, labeling, advertising, promotional practices, safety surveillance, record keeping and filing of required reports and their respective counterparts promulgated by Regulatory Authorities in countries outside the United States), (ii) the Public Health Service Act (42 U.S.C. § 262), and the regulations promulgated thereunder, (iii) all federal and state fraud and abuse Laws, including the Federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), the Stark Law (42 U.S.C. § 1395nn et seq), the Physician Payments Sunshine Act (42 U.S.C. § 1320a-7h), the False Claims Act (31 U.S.C. §§ 3729-3733), the Exclusion Laws and the Civil Monetary Penalties Law (42 U.S.C. § 1320a-7 and 1320a-7a), the Program Fraud Civil Remedies Act of 1986 (31 U.S.C. §§ 3801-3812), and the Eliminating Kickbacks in Recovery Act of 2018 (18 U.S.C. § 220), (iv) the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health, and the regulations promulgated thereunder and comparable state Laws, (v) the Controlled Substances Act, (vi) Titles XVIII and XIX of the Social Security Act and the regulations promulgated thereunder, and (vii) the Clinical Laboratories Improvement Amendment.

"In-Scope Entities" means those BridgeBio Subsidiaries set forth in Schedule 1.1(a).

"In-Scope Programs" means those programs set forth in Schedule 1.1(b).

"In-Scope Program Contracts" means all Contracts that are primarily related to the In-Scope Programs.

"In-Scope Program Employee" means each individual listed in Schedule 1.1(c), which schedule may be updated from time to time as reasonably agreed by BridgeBio and the Company in good faith or in accordance with the terms of this Agreement.

"In-Scope Program IP" means all Intellectual Property (other than Trademarks) used or held for use by BridgeBio or its Subsidiaries to conduct any In-Scope Program.

"In-Scope Program Know-How" means Know-How owned or purported to be owned by BridgeBio (or any of its Subsidiaries) as of the Closing Date to the extent primarily related to any of the In-Scope Programs.

"In-Scope Program Other IP" means all Intellectual Property owned or purported to be owned by BridgeBio (or any of its Subsidiaries) as of the Closing Date (other than Patents, Know-How and Trademarks) to the extent primarily related to any of the In-Scope Programs.

"In-Scope Program Patents" means (a) those Patents listed on Section 4.15(a) of the BridgeBio Disclosure Letter (excluding those Patents expressly identified on Section 4.15(a) of the BridgeBio Disclosure Letter as licensed to BridgeBio or one of its Subsidiaries by the University of Pittsburgh), (b) all provisional applications, continuations, continuations-in-part, divisionals, reissues, re-examinations, substitutions, and extensions of any of the Patents described in sub-clause (a), (c) any Patents that claim priority to any of the Patents in sub-clause (a) or (b), and (d) all equivalents and counterparts of any of the Patents in sub-clause (a), (b) or (c).

"Intellectual Property" means all of the following and all rights therein: (i) patents, patent applications, and all related provisional applications, priority and other applications, continuations, continuations-in-part, divisionals, reissues, re-examinations, substitutions, and extensions thereof and equivalents or counterparts of any of the foregoing ("Patents"), (ii) trademarks, service marks, names, corporate names, trade names, domain names, logos, slogans, trade dress, design rights, and other similar designations of source or origin, together with the goodwill symbolized by any of the foregoing ("Trademarks"), (iii) copyrights and copyrightable subject matter ("Copyrights"), (iv) computer programs (whether in source code, object code, or other form), algorithms, databases, compilations and data, technology supporting the foregoing, and all documentation, including user manuals and training materials, related to any of the foregoing ("Software"); (v) Know-How; (vi) all other intellectual property or proprietary rights of any nature; (vii) all causes of action and other enforcement actions with respect to any of the foregoing, including the right to sue for past, present or future infringement, misappropriation or violation of any of the foregoing; (viii) applications and registrations, and any renewals, extensions and reversions, for the foregoing; and (ix) all rights and remedies with respect to each of the foregoing, including the rights to obtain damages, injunctive relief, and any other remedies for past, present, and future infringement, misappropriation, or other violation thereof.

"Intercompany Termination Agreement" means the Intercompany Termination Agreement by and between BridgeBio, Portal Therapeutics, Inc., Sub21 Inc. and other parties thereto in substantially the form attached hereto as Exhibit E.

"Investor Material Adverse Effect" means, with respect to any Investor, any change, event, development, effect or occurrence that, individually or in the aggregate, has had, or would reasonably be expected to have, a material adverse effect on the ability of an Investor to perform its obligations under this Agreement or on the consummation by such Investor of the transactions contemplated hereby.

"IT Assets" means any tangible information technology equipment, including computers, firmware, workstations, routers, hubs, switches, data communication lines, and any documentation associated with, and any Software incorporated in, any of the foregoing.

"Know-How" means (i) all confidential or proprietary technical, engineering, scientific, regulatory or other information, results, knowledge, techniques or data, and (ii) inventions, processes, practices, methods, trade secrets, know-how, instructions, protocols, data (including biological, chemical, pharmacological, toxicological, pharmaceutical, physical and analytical, safety, quality control, and pre-clinical and clinical data), drawings, compositions, formulae, formulations, models, methodologies, schematics, and specifications; provided that Know-How does not include Patent rights to the extent claiming any of the foregoing.

"Knowledge of BridgeBio" means the actual knowledge of the individuals set forth in Schedule 1.2(a) (in each case, assuming reasonable inquiry of direct reports).

"Knowledge of Investors" with respect to each Investor, the actual knowledge of the individuals set forth opposite such Investor's name in Schedule 1.2(b).

"Law" means any applicable law (including, for the avoidance of doubt, Health Laws), act, statute, ordinance, rule, code, regulation, directive or other requirement or rule of law enacted, issued or promulgated by any Governmental Authority or any Governmental Order.

"Legal Proceeding" means any civil, criminal, judicial, administrative or arbitral actions, suits, charges, Governmental Orders, complaints, inquiries, arbitrations, hearings, litigation, proceedings (public or private), claims, investigations or audits by or before a Governmental Authority or other body of competent jurisdiction, including notice of violation or noncompliance, interferences, oppositions, cancellations or other proceedings of any nature.

"<u>Liabilities</u>" means any and all debts, liabilities, claims, guarantees, assurances, commitments and obligations of any kind, whether fixed, contingent or absolute, asserted or unasserted, matured or unmatured, liquidated or unliquidated, accrued or not accrued, known or unknown, due or to become due and whenever or however arising (including, whether arising out of any contract or tort based on negligence or strict liability).

"Losses" means any and all losses, damages, penalties, fines, Taxes, costs and expenses, and the amounts of or paid or payable in respect of, any and all Liabilities and claims (including interest, penalties, reasonable attorneys' fees and disbursements and all amounts paid in investigation, defense or settlement of any of the foregoing); provided, that "Losses" shall not include (except to the extent actually awarded to a Third Party in connection with a Third-Party Claim) any consequential, special, exemplary, incidental, indirect or punitive damages, lost profits, or diminution in value.

"Organizational Documents" means, with respect to any Person, the certificate or articles of incorporation, formation or organization and bylaws, partnership agreement, limited partnership agreement, limited liability company agreement, other operating agreement, stockholders' agreement or other similar governing documents of such Person, in each case, as amended through the date hereof.

"Permits" means all (i) Regulatory Approvals, to the extent exclusively related to any In-Scope Program (if applicable), including the Regulatory Approvals set forth in Schedule 1.1(e) and (ii) other consents, approvals, authorizations, certificates, filings, notices, permits, concessions, exemptions, clearances, registrations, franchises or licenses issued by any Regulatory Authority or other Governmental Authority.

"Permitted Encumbrances" means (a) Encumbrances of mechanics, carriers, workmen, repairmen, warehouseman, materialmen or other similar Encumbrances arising or incurred by operation of law or otherwise incurred in the ordinary course of business for security amounts (i) that are not overdue and payable and (ii) for which adequate reserves (as determined in accordance with GAAP) have been established; (b) Encumbrances for Taxes, labor claims, assessments and other governmental charges (i) which are not yet due and payable or which are being contested in good faith by appropriate proceedings and (ii) for which adequate reserves (as determined in accordance with GAAP) have been established; (c) Encumbrances imposed or promulgated by requirements and restrictions of zoning, permit, license, building and other applicable Laws, and development, site plan, subdivision or other agreements with municipalities that do not, and would not reasonably be expected to, materially detract from the value of any Acquired Subsidiary or any Acquired Assets or that do not materially interfere with the operation or use of any Acquired Subsidiary or any Acquired Assets or that do not materially interfere with the operation or use of any Acquired Subsidiary or any Acquired Assets as currently used.

"Person" means any individual, corporation, partnership, association, limited liability company, trust, estate, unincorporated organization, government, entity or agency or political subdivision thereof or other entity or any group comprised of two (2) or more of the foregoing.

"Pre-Closing Assignment and Assumption Agreement" means the Pre-Closing Assignment and Assumption Agreement by and between BridgeBio, the Company and other parties thereto in substantially the form attached hereto as Exhibit D.

"Pre-Closing Tax Period" means any taxable period (or portion thereof) that ends on, or prior to, the Closing Date. In the case of any taxable period that begins on or before but ends after the Closing Date (a "Straddle Period"), (i) property, ad valorem, and similar Taxes charged on a periodic basis allocable to the pre-Closing portion of such Straddle Period shall be equal to the amount of such Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of days during the Straddle Period that include or precede the Closing Date, and the denominator of which is the number of days in the entire Straddle Period; and (ii) Taxes other than Taxes described in clause (i) above of the Company allocable to the pre-Closing portion of such Straddle Period shall be computed as if such taxable period ended as of the end of the day of the Closing Date.

"Portal" means Portal Therapeutics, Inc., a Delaware corporation.

"Regulatory Approvals" means, with respect to a pharmaceutical product in an applicable regulatory jurisdiction, all permits, licenses, certificates, approvals, clearances, or other authorizations recognized by the applicable Regulatory Authority necessary to conduct clinical trials of, manufacture and/or commercialize, use, import, export, or otherwise exploit such pharmaceutical product in such regulatory jurisdiction in accordance with Law, including any Investigational New Drug application ("IND") or any other investigational medicinal product dossier.

"Regulatory Authority" means any applicable supranational, federal, foreign, national, regional, state, provincial, local or municipal regulatory agencies, departments, bureaus, commissions, councils or other Governmental Authority (including the FDA) and applicable institutional review boards (IRBs) regulating or otherwise exercising authority with respect to the In-Scope Programs.

"Sanctioned Party" means any Person: (i) organized under the laws of, ordinarily resident in, or located in a country or territory that is the subject of comprehensive Sanctions (which as of the date of this Agreement include Cuba, Iran, North Korea, Syria, and the Crimea, Donetsk, and Luhansk regions of Ukraine ("Restricted Countries")); (ii) 50% or more owned or controlled by the government of a Restricted Country; or (iii) (A) designated on a sanctioned parties list administered by the United States, including the U.S. Department of the Treasury's Office of Foreign Assets Control's Specially Designated Nationals and Blocked Persons List, Foreign Sanctions Evaders List, Sectoral Sanctions Identification List (collectively, "Designated Parties"); or (B) 50% or more owned or, where relevant under applicable Sanctions, controlled, individually or in the aggregate, by one or more Designated Party, in each case only to the extent that dealings with such Person is are prohibited pursuant to applicable Sanctions.

"Sanctions" means economic sanctions administered by the Office of Foreign Assets Control of the United States Department of the Treasury, His Majesty's Treasury, the European Union and the Bureau of Industry Security of the U.S. Department of Commerce, and any sanctions measures under the U.S. International Emergency Economic Powers Act, the U.S. Trading with the Enemy Act, the U.S. Iran Sanctions Act, the U.S. Comprehensive Iran Sanctions, Accountability and Divestment Act of 2010, and the U.S. Iran Threat Reduction and Syria Human Rights Act of 2012, the U.S. National Defense Authorization Act of 2012 and the U.S. National Defense Authorization Act of 2013, all as amended, and any executive order, directive or regulation pursuant to the authority of any of the foregoing.

"Special Committee" means the special committee of independent directors of the BridgeBio Board formed in connection with evaluation and negotiation of the transactions contemplated by the Transaction Documents and alternatives thereto.

"Stock Plans" means, collectively, (i) the Sub21, Inc. 2022 Stock Option and Grant Plan and (ii) the Portal Therapeutics, Inc. 2020 Stock Option and Grant Plan.

"Sub21" means Sub21, Inc., a Delaware corporation.

"Subsidiary" means, as to any Person, (a) any corporation, partnership, limited liability company or other entity a majority of the capital stock or other equity interests of which (i) ordinary voting power to elect a majority of the board of directors or other Persons performing similar functions or (ii) at least a majority of the economic interest attributable to all capital stock or other equity interests of any of the foregoing entities, is at the time owned, directly or indirectly, with power to vote, by such Person or any direct or indirect Subsidiary of such Person is a general partner or (c) a limited liability company in which such Person or any direct or indirect Subsidiary of such Person is a managing member or manager; provided, that, after the Closing, the Company shall not be deemed or treated as a Subsidiary of BridgeBio.

"Tax" or "Taxes" means, without duplication, all federal, state, provincial, local, or non U.S. income, profits, revenues, franchise, services, receipts, gross receipts, net worth, sales, use, excise, recording, real estate, real estate transfer, withholding, alternative minimum or add on, ad valorem, inventory, payroll, estimated, goods and services, employment, welfare, social security, disability, occupation, unemployment, premium, real property, personal property, capital stock, stock transfer, stamp, transfer, documentary, conveyance, production, windfall profits, pension, duties, customs duties, value added and other similar taxes, duties, levies, tariffs, imposts and assessments, including related interest, penalties, fines, and additions to tax levied by any Tax Authority.

"<u>Tax Authority</u>" means any national, federal, state, local, municipal or non-U.S. Governmental Authority exercising authority to charge, audit, regulate or administer the imposition of Taxes (including the U.S. Internal Revenue Service).

"<u>Tax Return</u>" means any return, declaration, report, claim for refund or information return, certificate, bill, statement or other written information required to be provided to or filed with any Governmental Authority relating to Taxes, including any supplement, schedule or attachment thereto, and including any amendment thereof.

"Third Party" means any Person who is not a Party or an Affiliate of a Party.

"<u>Transaction Documents</u>" means this Agreement, the Amended and Restated LLC Agreement, the Certificate of Formation, the Transition Services Agreement, Pre-Closing Assignment and Assumption Agreement, Intercompany Termination Agreement and any other certificate, instrument or other document delivered pursuant to any of the foregoing.

"<u>Transferred Employee Records</u>" means records of BridgeBio and its Subsidiaries that relate to Transferred Employees, but only to the extent that such records pertain to: (i) skill and development training, (ii) seniority histories, (iii) salary and benefit information, (iv) Occupational, Safety and Health Administration reports and records and (v) active medical restriction forms.

"<u>Transition Services Agreement</u>" means the Transition Services Agreement by and between BridgeBio Services, Inc., a Delaware corporation, on the one hand, and the Company, on the other hand, to be entered into at the Closing in substantially the form attached hereto as <u>Exhibit C</u>.

"<u>Unit</u>" means a unit of the Company representing a limited liability company interest in the Company and having the rights, preferences and privileges, and being subject to the covenants and agreements, set forth in the LLC Agreement.

"U.S." or "United States" means the United States of America.

## ARTICLE II PRE-CLOSING RESTRUCTURING

#### Section 2.1 Pre-Closing Restructuring.

- (a) Subject to Section 2.2 and applicable Law, prior to the Closing, BridgeBio shall have, and shall have caused its applicable Subsidiaries to, transfer, convey, deliver and assign to the applicable In-Scope Entity (to the extent not already owned by such In-Scope Entity) or the Formed Entity (designated by BridgeBio to hold the Acquired Assets for the Tuberous Sclerosis Complex (TSC) program (with respect to small molecule method only)) all of BridgeBio's and its Subsidiaries' respective right, title and interest in and to the Acquired Assets, free and clear of all Encumbrances (other than Permitted Encumbrances), and each In-Scope Entity or the Formed Entity, as applicable, has assumed, is responsible for and has timely discharged when due, all respective Assumed Liabilities, pursuant to the Pre-Closing Assignment and Assumption Agreement.
- (b) Each Party expressly agrees and acknowledges that the Acquired Assets will not include any assets, rights or interests of BridgeBio and its Subsidiaries that do not constitute "Acquired Assets", which excluded assets shall be deemed to include the following items (collectively, the "Excluded Assets"):
  - (i) all Intellectual Property other than the Acquired IP Assets, including, for clarity, any Trademarks that include (A) BRIDGEBIO and (B) any Trademarks containing or comprising the foregoing (A), including any Trademarks confusingly similar thereto or dilutive thereof (such Trademarks, the "BridgeBio Marks");
    - (ii) all IT Assets other than the Acquired IT Assets;
  - (iii) the portion of any Shared Contracts that are not allocated or assigned to or retained by the Company or any of its Subsidiaries, or to which BridgeBio or any of its Affiliates (other than the Acquired Subsidiaries) is or will be a party pursuant to Section 6.8 of the Agreement;
  - (iv) all (A) applications, submissions, registrations or notifications submitted to a Regulatory Authority for purposes of obtaining, updating or maintaining of any Regulatory Approval, (B) correspondence with or to Regulatory Authorities (including minutes and official contact reports relating to any communications with any Regulatory Authorities), (C) non-clinical, clinical and other files, writings, notes, studies, reports and other documents contained or referenced in or supporting any of the foregoing or any Regulatory Approval, and (D) all regulatory or legal rights in any of clauses (A) to (C), in each case of the foregoing (A) (D), other than the Acquired Regulatory Documentation;
  - (v) all Regulatory Approvals other than Regulatory Approvals exclusively related to any In-Scope Program (if applicable);
    - (vi) all rights of BridgeBio under this Agreement and the other Transaction Documents;

- (vii) (A) all real property and any buildings, improvements and fixtures thereon, and (B) all leasehold interests, including any prepaid rent, security deposits and options to renew or purchase in connection therewith, of BridgeBio or any of its Affiliates;
  - (viii) all cash and cash equivalents ("Cash");
- (ix) the portion of the "Intercompany receivables" reflected in the "Total assets" line item of the Financial Statements for each of Portal or Sub21 to the extent incurred on or prior to June 30, 2024 (the "Excluded Intercompany Receivables"); and
- (x) all insurance policies and binders and all claims, refunds and credits from insurance policies or binders due or to become due with respect to such policies or binders.

#### Section 2.2 <u>Non-Assignment of Certain Acquired Assets.</u>

(a) Notwithstanding the foregoing, this Agreement shall not constitute an agreement for BridgeBio or any of its Subsidiaries to sell, convey, assign, transfer or deliver to the Company the Acquired Assets set forth in Section 2.2 of the BridgeBio Disclosure Letter for which consent, authorization or approval of a Third Party (identified in Section 2.2 of the BridgeBio Disclosure Letter) is required but has not been obtained prior to the Closing Date because an attempted conveyance, assignment, transfer or delivery of such Acquired Asset, without the consent, authorization or approval of such Third Party, may constitute a breach or other contravention thereof or a violation of Law. For clarity, if any asset, right or interest that would otherwise constitute an Acquired Asset is not assignable or transferable as contemplated in this Section 2.2(a) (each, a "Non-Assigned Asset"), such asset shall not be deemed an Acquired Asset; provided, however, following BridgeBio's receipt of the relevant consent, authorization or approval, as applicable, BridgeBio shall, or shall cause its Subsidiaries to, promptly assign or transfer to the Company the Non-Assigned Asset, and such asset shall thereafter be deemed an "Acquired Asset" for purposes of this Agreement.

- If, on the Closing Date, any such consent, authorization or approval is not obtained, or if an attempted conveyance, assignment, transfer or delivery thereof would constitute a breach or other contravention or a violation of Law, in each case, with respect to a Non-Assigned Asset (whether or not listed on Section 2.2 of the BridgeBio Disclosure Letter), BridgeBio shall, or shall cause its applicable Subsidiaries to (as applicable), for twenty-four (24) months following the Closing Date, use commercially reasonable efforts to obtain any such consent, authorization or approval as promptly as practicable after the date hereof, and the Company shall, and shall cause each of its applicable Affiliates to, use its commercially reasonable efforts to cooperate with BridgeBio or any of its applicable Subsidiaries, as applicable, to obtain any such consent, authorization or approval, necessary for the conveyance, assignment, transfer or delivery of any such Non-Assigned Asset to the Company or its designee, and upon receipt of such consent, authorization or approval, BridgeBio shall, or shall cause its Subsidiaries to, promptly assign or transfer to the Company or its designee such Non-Assigned Asset for no additional consideration. Prior to conveying a Non-Assigned Asset as provided in this Section 2.2(b), BridgeBio shall, and shall cause its applicable Subsidiaries to, cooperate and use commercially reasonable efforts to obtain a mutually acceptable arrangement under which the Company (and/or one of its Subsidiaries) will, in compliance with Law and the terms of the applicable Non-Assigned Asset, obtain the benefits of, and assume the obligations and bear the economic burdens associated with, such Non-Assigned Asset, claim, right or benefit in accordance with this Agreement and the relevant Non-Assigned Asset, claim, right or benefit, including subcontracting or sublicensing to the Company (and/or one of its Subsidiaries), or under which BridgeBio will, or will cause one of its Subsidiaries to, (i) enforce, for the benefit of the Company (and/or its Subsidiaries) any and all of its or their rights against a Third Party (including any Governmental Authority) associated with such Non-Assigned Asset, claim, right or benefit, and (ii) promptly pay to the Company (and/or one of its Subsidiaries), when received, all monies received by it or them under any such Non-Assigned Asset, claim, right or benefit, and the Company (and/or one of its Subsidiaries) will assume the Assumed Liabilities associated therewith. In the event that BridgeBio remains unable to convey such Non-Assigned Asset after using such commercially reasonable efforts to do so for twenty-four (24) months following the Closing Date, and during such period in which BridgeBio attempts to obtain such consent, authorization or approval, BridgeBio, upon the Company's prior written request, will cooperate with BridgeBio and use commercially reasonable efforts to assist the Company in entering into a new Contract with the applicable Third Party on substantially similar terms; provided that such assistance shall not include assistance by BridgeBio with the negotiation of commercial terms between the Company or any of its Subsidiaries, on the one hand, and the applicable Third Party, on the other hand, related to such new Contract and in no event shall BridgeBio or any of its Subsidiaries be required to take any action that will result in any additional economic obligations applicable to BridgeBio or any of its Subsidiaries; provided, further, that nothing in this Section 2.2(b) shall require BridgeBio or any of its Subsidiaries to pay any consideration or make any concession with respect to any novation or assignment; provided, further, in no event shall the number of initial Units and the percentage of outstanding Units set forth on Schedule 3.1 be reduced or increased in respect of any Non-Assigned Asset.
- (c) BridgeBio provides no assurances to the Investors that any consent, authorization or approval of a Third Party contemplated by this Section 2.2 will be granted. Subject to compliance by BridgeBio with the provisions of this Section 2.2, the Parties acknowledge and agree that neither BridgeBio nor its Affiliates shall be obligated to obtain any such authorization, approval or consent hereunder and neither (i) the failure to so actually obtain any such authorization, approval or consent in connection with the consummation of the transactions contemplated by this Agreement in and of itself nor (ii) any default or termination or action commenced or threatened by or on behalf of any Person to the extent arising out of such failure to so actually obtain any such authorization, approval or consent in connection with the consummation of the transactions contemplated by this Agreement in and of itself shall be deemed to constitute a breach of this Agreement.

## ARTICLE III CLOSING

#### Section 3.1 <u>Closing.</u>

(a) General. The closing of the transactions contemplated by this <u>Section 3.1</u> (the "<u>Closing</u>") shall take place on the date hereof remotely via the electronic exchange of documents and signature pages.

#### (b) <u>Closing Contributions; Issuance of Company Interests</u>. At the Closing:

- (i) subject to Section 2.2, after giving effect to the Pre-Closing Restructuring, BridgeBio shall contribute, assign and transfer (or cause to be contributed, assigned and transferred) the Owned Shares, free and clear of all Encumbrances (other than Permitted Encumbrances), and the Company shall accept and assume from BridgeBio or its applicable Subsidiaries such Owned Shares;
- (ii) immediately following BridgeBio's contribution pursuant to the preceding clause (i), each Investor shall contribute to the Company cash in the amount equal to such Investor's Closing Date cash contribution as set forth opposite such Investor's name on <u>Schedule 3.1</u> under the column "Closing Date Cash Contributions" (the "<u>Closing Date Cash Contribution</u>") by wire transfer of immediately available funds to the account designated in writing by the Company (which account shall be designated at least two Business Days prior to the Closing Date); and
- (iii) in exchange for the Closing Date Cash Contribution of each Investor, the Company shall issue to such Investor a number of newly issued Units of the Company as set forth opposite such Investor's name on Schedule 3.1 under the column "Class A Units" (the "Initial Company Interests"). For the avoidance of doubt, concurrently with making of the Closing Date Cash Contribution by the Investors at the Closing, any and all of the equity interests or Units held by BridgeBio Pharma LLC in the Company immediately prior to the Closing shall be extinguished, forfeited and cancelled in full for no consideration, and the Units set forth opposite BridgeBio Pharma LLC's name on Schedule 3.1 under the column "Units" shall constitute BridgeBio Pharma LLC's only equity interests in the Company.

#### Section 3.2 <u>Other Closing Deliveries</u>.

- (a) BridgeBio and Company Deliveries. At the Closing, BridgeBio shall, and shall cause the Company or BridgeBio Services, Inc. to (if applicable) execute and deliver to the Investors' Representatives:
  - (i) duly executed Pre-Closing Assumption and Assignment Agreement;
  - (ii) duly executed counterparts to the Amended and Restated LLC Agreement; and
  - (iii) duly executed counterparts to the Transition Services Agreement;
  - (iv) written resignations of all members of the board of directors (or comparable governing body) of the Company and the Acquired Subsidiaries (with respect to their positions as such and not with respect to their employment, if applicable) other than those set forth on Section 3.2(a)(iv) of the BridgeBio Disclosure Letter;

- (v) termination and/or release documentation (and other customary documentation and filings, including UCC-3 financing statements), in form and substance reasonably satisfactory to the Investors, evidencing the release of all Encumbrances on the assets or properties of the In-Scope Entities created pursuant to and in connection with the Financing Agreement; and
  - (b) Investor Deliveries. At the Closing, each Investor shall execute and deliver to BridgeBio and the Company:
    - (i) duly executed counterparts to the Amended and Restated LLC Agreement; and
    - (ii) a duly executed IRS Form W-9 with respect to each Investor.
- Section 3.3 Withholding Rights. Each of the Investors and their respective Affiliates, as applicable, shall be entitled to deduct and withhold from the amounts otherwise payable or contributed pursuant to this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code and the rules and the Treasury Regulations promulgated thereunder, or any other applicable provision of federal, state, local or non-U.S. Tax Law; provided, that prior to making any such deduction or withholding, the applicable Investor shall use commercially reasonable efforts to notify BridgeBio and the Company of such withholding and provide a reasonable opportunity for BridgeBio and/or the Company to provide such forms or other evidence that would eliminate or reduce any such deduction or withholding. To the extent that amounts are so deducted or withheld and remitted to the applicable Taxing Authority, such amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction or withholding was made.

## ARTICLE IV CERTAIN REPRESENTATIONS AND WARRANTIES OF BRIDGEBIO

Except as set forth in the disclosure letter delivered by BridgeBio to the Investors' Representatives on or prior to the date of this Agreement (the "BridgeBio Disclosure Letter"), BridgeBio hereby represents and warrants to each Investor as follows:

#### Section 4.1 <u>Organization</u>.

- (a) Each of BridgeBio, the Company and the Acquired Subsidiaries is duly organized or formed, validly existing and in good standing (to the extent applicable) under the Laws of the jurisdiction of its organization or formation.
- (b) Each of BridgeBio, the Company and the Acquired Subsidiaries is duly qualified or licensed to do business and is in good standing (to the extent applicable) in each jurisdiction where the character of the properties owned, leased or licensed by it or the nature of its business makes such qualification or licensing necessary, except where the failure to be so qualified or licensed or in good standing (if applicable) would not have a BridgeBio Material Adverse Effect. Each of BridgeBio, the Company and the Acquired Subsidiaries possesses all requisite corporate, limited liability company or similar power and authority necessary to own, operate and lease and license its properties and to carry on its business as currently conducted. BridgeBio has made available to the Investors correct and complete copies of the Organizational Documents of the Company and each Acquired Subsidiary, which and are in full force and effect, and neither BridgeBio nor any of its Subsidiaries is in breach or in violation of any such Organizational Documents.

#### Section 4.2 Power and Authorization; Enforceability.

- BridgeBio has all requisite corporate or similar power and authority to execute and deliver the Transaction Documents to which it is a party in connection with the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. BridgeBio has taken all actions required to be taken by or on behalf of BridgeBio to authorize and permit the execution and delivery by BridgeBio of this Agreement and each Transaction Document required to be executed and delivered by it pursuant hereto, the performance by BridgeBio of its obligations hereunder and thereunder, and the consummation by BridgeBio of the transactions contemplated hereby and thereby. This Agreement, and each Transaction Document required to be executed and delivered by BridgeBio pursuant hereto, has been duly executed and delivered by BridgeBio and, assuming the due authorization, execution and delivery by each of the other parties hereto or thereto, constitutes the legal, valid and binding obligations of BridgeBio, enforceable against BridgeBio in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other Laws relating to or affecting creditors' rights or by general equitable principles (regardless of whether enforcement is sought at Law or in equity) (the "Enforceability Exceptions").
- (b) The Company has all requisite limited liability company or similar power and authority to execute and deliver the Transaction Documents to which it is a party in connection with the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. The Company has taken all actions required to be taken by or on behalf of the Company to authorize and permit the execution and delivery by the Company of this Agreement and each Transaction Document required to be executed and delivered by it pursuant hereto, the performance by the Company of its obligations hereunder and thereunder, and the consummation by the Company of the transactions contemplated hereby and thereby. This Agreement, and each Transaction Document required to be executed and delivered by the Company pursuant hereto, has been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery by each of the other parties hereto or thereto, constitutes the legal, valid and binding obligations of the Company, enforceable against the Company in accordance with its terms, except as such enforceability may be limited by the Enforceability Exceptions.
- (c) The Initial Company Interests, when issued, sold and delivered in accordance with the terms and for the consideration set forth in this Agreement, will be validly issued, fully paid and nonassessable and free of restrictions on transfer other than restrictions on transfer under the Transaction Documents, applicable state and federal securities laws and liens or encumbrances created by or imposed by an Investor. Assuming the accuracy of the representations of the Investors in Article V of this Agreement, the issuance of the Initial Company Interests will not require registration under applicable federal and state securities laws.

- Section 4.3 No Prior Operations or Liabilities. The Company is a limited liability company that was formed in order to facilitate and engage in the transactions contemplated by this Agreement and the other Transaction Documents. As of the date hereof, the Company (i) has not engaged in any business activities or conducted any operations other than entering into the Original LLC Agreement and this Agreement, (ii) does not own any assets, (iii) does not have any liabilities or obligations of any kind whatsoever in existence, whether accrued, contingent, absolute or otherwise, other than pursuant to its formation and this Agreement, and (iv) is not a party to any Contract other than the Original LLC Agreement, this Agreement and the other Contracts set forth in Section 4.3 of the BridgeBio Disclosure Letter.
- Section 4.4 <u>Consents and Approvals; No Conflicts</u>. Neither the execution or delivery of this Agreement or any other Transaction Document by BridgeBio or the Company nor the consummation of the transactions contemplated hereby and thereby will, directly or indirectly:
- (a) result in a breach or violation of, or default under, the Organizational Documents of BridgeBio, the Company or the Acquired Subsidiaries;
- (b) require any notice, consent, waiver, approval, order or authorization of, or registration or filing with, any Governmental Authority (collectively, "Governmental Filings and Approvals"), other than (i) to notify any Regulatory Authority of the transfer to the Company or its designee of any Regulatory Approval relating to any applicable In-Scope Program(s) (if applicable) or (ii) that, if not obtained or made, would have a BridgeBio Material Adverse Effect;
- (c) require any notice, consent or other action by any Person under, constitute a default (with or without notice, lapse of time, or both) under or breach of, or give rise to any right of termination, cancellation or acceleration of any right or obligation of BridgeBio, the Company or any Acquired Subsidiary, or to a loss of any benefit to which BridgeBio, the Company or any Acquired Subsidiary is entitled under, any In-Scope Program Contract, except for any violations, breaches, defaults, terminations, cancellations or accelerations or loss of any benefit as would not have a BridgeBio Material Adverse Effect; or
- (d) (i) constitute or result in a violation of any Law or Governmental Order to which BridgeBio, an Acquired Subsidiary or the In-Scope Programs are subject or by which any of BridgeBio's or the Company's properties or assets is bound; or (ii) result in the creation or imposition of an Encumbrance (other than Permitted Encumbrances) upon any Acquired Subsidiary or any Acquired Asset, except with respect to clauses (i) and (ii) as would not, individually or in the aggregate, reasonably be expected to (A) be material to the Company, the Acquired Subsidiaries or the In-Scope Programs, taken as a whole, or (B) materially affect, prevent or delay the ability of BridgeBio or the Company to consummate the transactions contemplated by this Agreement.

- Section 4.5 Acquired Subsidiary, Schedule 1.1(a) sets forth, as of the date hereof, the name of each Acquired Subsidiary, its jurisdiction of organization, the total amount of issued and outstanding shares (on a fully diluted basis), the amount and class of shares held by each shareholder, including BridgeBio or one of its Subsidiaries and such ownership percentages (on a fully diluted basis) (such shares owned by BridgeBio or one of its Subsidiaries, the "Owned Shares"). The equity interests of each Acquired Subsidiary have been duly authorized and validly issued and are fully paid and nonassessable. As of the date hereof, the Owned Shares are owned, free and clear of all Encumbrances (other than the Encumbrances set forth on Section 4.5(a) of the BridgeBio Disclosure Letter). None of the Acquired Subsidiaries own any equity interests in any other Person. None of the Acquired Subsidiaries has an obligation to purchase, redeem or otherwise acquire its equity interests. Except as set forth on Section 4.5(b) of the BridgeBio Disclosure Letter, there is no voting trust or agreement, stockholders agreement, pledge agreement, buy-sell agreement, right of first refusal, preemptive right or proxy relating to the equity interests of any Acquired Subsidiary. Except as set forth on Schedule 1.1(a), there are no outstanding securities convertible or exchangeable into, or the value of which is measured by reference to, equity securities of Acquired Subsidiaries, including any options, warrants, phantom stock, stock appreciation, profit participation or other equity or equity-based rights or interests. Schedule 1.1(a) sets forth, with respect to each Equity Award: (a) the name of the Equity Award Holder, (b) the total number of shares of common stock of the applicable Acquired Subsidiary that are subject to each such Equity Award, (c) if applicable, the exercise price per share of common stock of the applicable Acquired Subsidiary, (d) the grant date and, if applicable, the expiration date of such Equity Award, and (e) the vesting schedule and current vesting status for such Equity Award. Each Equity Award grant was made in accordance with the terms of the Stock Plans, applicable Law and Section 409A of the Code, and, with respect to each Equity Award which is a stock option, such the per share exercise price of such Equity Award was equal to or greater than the fair market value of a share of common stock of the applicable Acquired Subsidiary on the date of grant of such Equity Award.
- Section 4.6 <u>Assets; Sufficiency of Assets</u>. BridgeBio or one of its Subsidiaries has good, valid and marketable title to all of the Acquired Assets, free and clear of all Encumbrances, other than Permitted Encumbrances, and where the concept is applicable, the tangible Acquired Assets have been maintained in accordance with standard industry practice and are in good operation condition (normal wear and tear expected). Except as set forth in <u>Section 4.6</u> of the BridgeBio Disclosure Letter, the Acquired Assets, immediately following the Closing and when taken together with the services, rights and benefits to be provided to the Company under this Agreement, the Transition Services Agreement and the other Transaction Documents, as applicable, constitute all assets that are necessary to operate the business relating to the In-Scope Programs, in substantially the same manner as such business was conducted immediately prior to the date hereof and the twelve (12) months preceding such date; <u>provided</u> that the foregoing is not a representation or warranty with respect to infringement, misappropriation or any other violation of Intellectual Property (which is addressed in <u>Section 4.15(b)</u>).
- Section 4.7 <u>Litigation</u>. There is no, and since January 1, 2021 there has not been any, material Legal Proceeding pending or, to the Knowledge of BridgeBio, threatened against BridgeBio or the Company (with respect to the In-Scope Programs, In-Scope Program Employees, Acquired Subsidiaries or Acquired Assets), or against any of the In-Scope Program Employees (in their capacity as such). There is no, and since January 1, 2021 there has not been any, material (a) Governmental Order affecting the In-Scope Programs, In-Scope Program Employees, any Acquired Subsidiary or any Acquired Asset, or (b) investigation or suit by any Governmental Authority relating to the In-Scope Programs, any Acquired Subsidiary or any Acquired Asset.
- Section 4.8 <u>Compliance with Laws</u>. Since January 1, 2021, each of BridgeBio, its Affiliates and the Company (with respect to the In-Scope Programs, In-Scope Program Employees, the Acquired Subsidiaries and the Acquired Assets) and, to the Knowledge of BridgeBio, each In-Scope Program Employee (in its capacity as such) is not and has not been in violation of any Laws or Governmental Orders.

Section 4.9 Permits. BridgeBio or one of its Subsidiaries holds or has the right to use (and at the Closing, following the Pre-Closing Restructuring, the Acquired Subsidiaries will hold or have the right to use) all material Permits required for the development of the In-Scope Programs as now being developed. All such Permits are valid and in full force and effect, and BridgeBio and its applicable Subsidiaries are in compliance with, and not in default under or violation of, all such Permits except where the failure to be in compliance with such Permits would not, individually or in the aggregate, reasonably be expected to be material to the In-Scope Programs. BridgeBio and its Subsidiaries have not received any written notice from any Governmental Authority of any material violations, or alleged violations, of any Permit or Law applicable to the In-Scope Programs, or any cancellation, suspension, revocation, invalidation or non-renewal of any such Permit.

#### Section 4.10 <u>Taxes</u>.

- (a) All income and other material Tax Returns with respect to the Acquired Assets and all income and other material Tax Returns of the Acquired Subsidiaries required to be filed have been timely filed (taking into account any applicable extensions) and all such Tax Returns are true, correct and complete in all material respects. All income and other material Taxes with respect to the Acquired Assets and all income and other material Taxes of the Acquired Subsidiaries required to be paid or withheld and remitted on or prior to the Closing Date have been duly and timely paid or have been duly and timely withheld and remitted.
- (b) All material deficiencies for Taxes asserted or assessed in writing against any Acquired Subsidiary have been fully and timely paid or settled.
- (c) No Acquired Subsidiary is currently the subject of a Tax audit, examination or other similar proceeding with respect to any material Taxes and no such audit, examination or other proceeding has been threatened in writing that has not been resolved or is currently pending.
- (d) There are no outstanding waivers, agreements or consents extending, waiving or having the effect of extending, waiving or requesting the extension or waiver of the statutory period of limitations applicable to any claim for, or the period for the collection or assessment or reassessment of, Taxes due from any Acquired Subsidiary for any taxable period and no request for any such waiver or extension is currently pending.
- (e) There are no Encumbrances for Taxes on any of the Acquired Assets or any assets of any of the Acquired Subsidiaries, other than Permitted Encumbrances.
- (f) None of the Acquired Subsidiaries (i) has entered into, and is bound by, any Tax sharing, allocation, indemnification or similar agreement (other than agreements entered into in the ordinary course of business, a principal purpose of which is not related to Taxes) that will not terminate on or prior to the Closing Date, (ii) are or have been a member of any consolidated, combined, unitary or similar group for purposes of filing Tax Returns or paying Taxes (other than any such group of which BridgeBio or any of its Subsidiaries is parent), or (iii) has any liability for any Tax of any Person under Treasury Regulations Section 1.1502-6 (or similar provision of state, local or non-U.S. Law), by contract (other than contracts entered into in the ordinary course of business, a principal purpose of which is not related to Taxes), as a transferee or successor, or otherwise.

- (g) The Acquired Subsidiaries have, and the Company has, in respect of the Acquired Assets, in compliance with applicable Law (i) collected and remitted sale, use, value added and similar Taxes with respect to sales or leases made or services provided to its customers and (ii) for all sales, leases or provision of services that are exempt from sales, use, value added and similar Taxes and that were made without charging or remitting sales, use, value added or similar Taxes, received and retained any appropriate Tax exemption certificates and other documentation qualifying such sale, lease or provision of services as exempt.
- (h) None of the Acquired Subsidiaries have constituted either a "distributing corporation" or a "controlled corporation" in a distribution of stock intended to qualify for tax-free treatment under Section 355 or Section 361 of the Code in the two (2) years prior to the date of this Agreement.
- (i) None of the Acquired Subsidiaries have consummated or participated in, nor are any of the Acquired Subsidiaries currently participating in, any transaction which was or is a "listed transaction" as defined in Section 6707A(C)(2) of the Code and Treasury Regulation Section 1.6011-4(b) or analogous anti-avoidance provision of state, local or non-U.S. law.
- (j) No Acquired Subsidiary has engaged in a trade or business, has a permanent establishment, within the meaning of any applicable Tax treaty, office or fixed place of business, or other form of taxable nexus in any country other than in which it is organized.
- (k) None of the Acquired Subsidiaries will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any Tax period (or portion thereof) ending after the Closing as a result of any: (i) change in or improper use of a method of accounting for any Pre-Closing Tax Period; (ii) "closing agreement" as described in Section 7121 of the Code (or any comparable or similar provisions of applicable Law) executed prior to the Closing; (iii) installment sale or open transaction disposition made prior to the Closing; (iv) prepaid amount or deferred revenue received prior to the Closing; or (v) gain recognition agreement entered into prior to the Closing to which the Company is a party under Section 367 of the Code.
- (l) Each Acquired Subsidiary is, and since the day prior to the date hereof has been, validly classified as a corporation for U.S. federal and applicable state and local income Tax purposes.
  - Section 4.11 Opinion of Financial Advisor; Special Committee Recommendation; BridgeBio Board Approval.
- (a) The Special Committee has received the opinion (the "<u>Fairness Opinion</u>") of Goldman Sachs & Co. LLC (which, if initially rendered verbally, has been or will be confirmed by a written opinion), to the effect that, as of the date of such opinion, and based upon and subject to the various limitations, qualifications and assumptions set forth therein, the consideration consisting of Class A Units to be paid to the Company for the Acquired Subsidiaries pursuant to this Agreement was fair from a financial point of view to the Company.

- (b) The Special Committee has been duly authorized and constituted by resolution of the BridgeBio Board and has (i) determined that the terms of this Agreement and the transactions contemplated hereby are fair to and in the best interests of BridgeBio and its stockholders and (ii) resolved to make the Special Committee Recommendation to the BridgeBio Board.
- (c) The BridgeBio Board, acting upon the Special Committee Recommendation, has (i) determined that the terms of this Agreement and the transactions contemplated hereby are fair to and in the best interests of BridgeBio and its stockholders; and (ii) approved and declared advisable the execution, delivery and performance of this Agreement and the transactions contemplated hereby.
- Section 4.12 <u>Brokers</u>. Except for Goldman Sachs & Co. LLC, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the transactions contemplated by this Agreement based on arrangements made by BridgeBio or its Affiliates.
- Section 4.13 <u>Absence of Changes</u>. Except as set forth on <u>Section 4.13</u> of the BridgeBio Disclosure Letter, since December 31, 2023 through the date hereof, (a) there has not been a BridgeBio Material Adverse Effect and (b) BridgeBio and its Subsidiaries have conducted the business of the Company and the Acquired Subsidiaries in the ordinary course in all material respects.

#### Section 4.14 Financial Statements; No Undisclosed Liabilities.

- (a) Section 4.14 of the BridgeBio Disclosure Letter sets forth true, complete and correct copies of the unaudited consolidated balance sheets of each In-Scope Entity as of June 30, 2024, and the related unaudited consolidated statements of operations for the six (6) month period then ended (together, the "Financial Statements"). The Financial Statements have been prepared in accordance with GAAP (consistently applied during the periods involved) and are in accordance with the accounting records of BridgeBio and its Subsidiaries, and present fairly, in all material respects, the financial condition and results of operations of each In-Scope Entity, as of the applicable date or period covered thereby, subject to (i) normal year-end adjustments, (ii) the absence of footnotes, and (iii) the exclusion of Excluded Assets and Excluded Liabilities.
- (b) None of the Acquired Subsidiaries has any Liabilities of a type that would be required to be reflected on a balance sheet of the Acquired Subsidiaries prepared in accordance with GAAP, except for (i) Liabilities reflected (and reserved for in accordance with GAAP) in the Financial Statements, (ii) Liabilities that have arisen after the date of the Financial Statements in the ordinary course (none of which relates to a breach of Contract, breach of warranty, tort, misappropriation, infringement, violation of Law or a Legal Proceeding) and (iii) Liabilities that, individually or in the aggregate, would not be material to the Acquired Subsidiaries, taken as a whole.

#### Section 4.15 <u>Intellectual Property</u>.

- (a) Section 4.15(a) of the BridgeBio Disclosure Letter contains a complete and accurate list of all issuances, registrations and applications for (i) Patents, (ii) Copyrights and (iii) domain names, in each case of the foregoing ((i)-(iii)) owned or purported to be owned by BridgeBio (or any of its Subsidiaries) and included in the Acquired Assets or otherwise exclusively licensed from a third party pursuant to an In-Scope Program Contract (specifying which is owned by a third party) (collectively, "Registered Intellectual Property"). BridgeBio or one of its Subsidiaries exclusively owns all right, title, and interest in and to all Intellectual Property included in the Acquired Assets owned or purported to be owned by BridgeBio or any of its Subsidiaries, free and clear of all Encumbrances except for Permitted Encumbrances, and BridgeBio or one of its Subsidiaries has the valid right to use all other In-Scope Program IP; provided that the foregoing is not a representation or warranty with respect to infringement, misappropriation or any other violation of Intellectual Property (which is addressed in Section 4.15(b)). All Registered Intellectual Property is subsisting and with respect to issued Registered Intellectual Property, to the Knowledge of Bridge Bio, valid and enforceable. Neither BridgeBio nor any of its Subsidiaries has, since January 1, 2021, received any claims or allegations challenging the ownership, validity, or enforceability of such Intellectual Property.
- (b) To the Knowledge of BridgeBio, (i) the conduct of the business of BridgeBio and its Subsidiaries with respect to the In-Scope Programs does not, and has not since January 1, 2021, infringed, misappropriated, or violated, and none of BridgeBio or its Subsidiaries has received any written allegations from any third party of infringement, misappropriation, or violation of, third-party Intellectual Property, and (ii) no third party is, or has since January 1, 2021, infringed, misappropriated or violated any In-Scope Program IP, except in each case (i) and (ii) as would not reasonably be expected to be material to any In-Scope Program.
- (c) BridgeBio and its Subsidiaries have taken commercially-reasonable steps to protect and preserve the confidentiality of the material trade secrets, and the secrecy of confidential Know-How, included in the In-Scope Program IP and none of the material confidential Know-How included in the In-Scope Program IP has been disclosed or authorized to be disclosed to any Person, other than in the ordinary course of business pursuant to obligations of confidentiality, non-disclosure and non-use (whether by Contract or operation of Law).
- (d) Except where such Intellectual Property was assigned to or ownership otherwise vested in BridgeBio or its Subsidiaries by operation of Law, all current and former employees, consultants and contractors of BridgeBio and its Subsidiaries that have participated in the creation or development of any material Intellectual Property for or behalf of, or for the benefit of, any In-Scope Program have executed an invention assignment agreement that assigns any such rights in and to such Intellectual Property to BridgeBio or one of its Subsidiaries, as applicable.
- (e) To the Knowledge of BridgeBio, no funding, facilities or personnel of any Governmental Authority were used, directly or indirectly, to develop or create in whole or in part, any of the material Intellectual Property included in the Acquired Assets in a manner that could reasonably require BridgeBio or any of its Subsidiaries to grant such Governmental Authority any license or other rights in, to or under such Intellectual Property.

- (f) Bridge Bio and its Subsidiaries have taken commercially-reasonable efforts designed to protect the security and integrity of the IT Assets used or held for use by BridgeBio or any of its Subsidiaries and data processed thereby, (ii) to the Knowledge of BridgeBio, such IT Assets do not contain any malicious code, viruses, or other Software routines or hardware components that (A) disrupt, harm or adversely affect the functionality of any such IT Asset or (B) enable or assist any third party to access without authorization any such IT Asset, (iii) to the Knowledge of BridgeBio, no Person has gained unauthorized access to, or unauthorized use of, any of such IT Assets in a manner compromising the security of such IT Assets or the In-Scope Programs, and (iv) there has been no failure or outage of any such IT Asset that has resulted in a material disruption to the business relating to the In-Scope Programs.
- (g) BridgeBio and its Subsidiaries are and have been, solely with respect to the conduct of the business relating to the In-Scope Programs, in compliance in all material respects with all binding contractual obligations, notices and policies, in each case to the extent relating to privacy, data protection or the collection, retention, protection, transfer, use or processing of personal data, (ii) except as would not have a material adverse effect with respect to the In-Scope Programs or the Acquired Assets, there has not been, since January 1, 2021, any security breach or other unauthorized use, disclosure, modification, or access to any personal data relating to the In-Scope Programs or included in the Acquired Assets, and (iii) BridgeBio and its Subsidiaries have not since January 1, 2021 received any written notices alleging any violation of any applicable Laws, contractual obligations, or policies relating to privacy, data protection, or the retention, protection, transfer, use or processing of personal data.
- Section 4.16 <u>Industry Regulatory Matters</u>. Since January 1, 2021, (i) all tests, preclinical studies and clinical trials conducted by or on behalf of BridgeBio and the Subsidiaries in relation to the In-Scope Programs have been conducted in material compliance with all applicable Laws and, as applicable, good laboratory practices, good clinical practices and good manufacturing practices, (ii) as of the date hereof, no clinical trial conducted by or on behalf of BridgeBio or its Subsidiaries has been terminated or suspended prior to completion for safety or other non-business reasons, and (iii) neither the FDA nor any comparable foreign Regulatory Authority has commenced any action to place a clinical hold order on, or otherwise terminate or suspend, any ongoing clinical investigation conducted by or on behalf of BridgeBio or its Subsidiaries. Neither BridgeBio nor any of its Subsidiaries is debarred or otherwise excluded from participation in any government program related to medical products under any In-Scope Program.

#### Section 4.17 <u>Employee Benefits</u>.

(a) Section 4.17 of the BridgeBio Disclosure Letter sets forth a true and complete list of each Acquired Subsidiary Plan. Each Acquired Subsidiary Plan is exclusive to the Acquired Subsidiaries. With respect to each Acquired Subsidiary Plan, BridgeBio has made available, to the extent applicable: (i) the plan documents, with any amendments; (ii) the most recent summary plan description; (iii) the most recent determination or opinion letter received from the Internal Revenue Service (the "IRS"); and (iv) in the event no summary plan description exists with respect to any such Acquired Subsidiary Plan, a written description of the material terms of such Acquired Subsidiary Plan.

- (b) Each Employee Plan has been established, maintained, funded and administered in all material respects in accordance with its terms and in compliance with all applicable Laws. Each Employee Plan that is intended to be qualified under Section 401(a) of the Code has received a favorable determination or relies upon an opinion or advisory letter from the IRS except as would not reasonably be expected to result in a material Liability to BridgeBio or any Acquired Subsidiary.
- (c) No Acquired Subsidiary has any Liability (i) under or with respect to any plan, program or arrangement (A) that is subject to Title IV of ERISA or Section 412 of the Code, including any "multiemployer plan" (as defined in Section 3(37) of ERISA), or (B) that provides retiree or post-employment health or life insurance benefits (other than as required by COBRA) or (ii) by reason of at any time being considered a single employer under Section 414 of the Code with any other Person. No Acquired Subsidiary has incurred (whether or not assessed) any Liability under Section 4980B, 4980D, 4980H, 6721 or 6722 of the Code.
- (d) Neither the execution and delivery of this Agreement, nor the consummation of the transactions contemplated by this Agreement, either alone or in combination with another event, could: (i) entitle any In-Scope Program Employee or other current or former individual service provider of any Acquired Subsidiary (or any dependent or beneficiary thereof) to any payment of compensation or benefits (whether in cash, property or the vesting of property); (ii) increase the amount of compensation or benefits due or payable to any such person set forth in the preceding clause (i); (iii) accelerate the vesting, funding or time of payment of any compensation, equity award or other benefit; (iv) require a contribution by any Acquired Subsidiary to any Acquired Subsidiary Plan; (v) restrict the ability of any Acquired Subsidiary to merge, amend or terminate any Acquired Subsidiary Plan; (vi) result in the forgiveness of any employee or service provider loan; or (vii) result in the payment of any amount that could, individually or in combination with any other such payment, constitute an "excess parachute payment" (within the meaning of Section 280G(b)(1) of the Code) or result in the imposition on any person of an excise tax under Section 4999 of the Code.
- (e) Each Employee Plan (to the extent it relates to any In-Scope Program Employee) and Acquired Subsidiary Plan that constitutes in any part a "nonqualified deferred compensation plan" (within the meaning of Section 409A of the Code) has been operated, maintained and administered in all material respects in compliance with Section 409A of the Code and all regulations and guidance thereunder.
- (f) No Acquired Subsidiaries have any obligation (and no Employee Plan (to the extent it relates to any In-Scope Program Employee) or Acquired Subsidiary Plan contains any obligation) to indemnify, gross-up, reimburse or otherwise make whole any Person for any Taxes pursuant to Section 4999 or Section 409A of the Code or otherwise.

## Section 4.18 <u>Labor and Employment.</u>

(a) Section 4.18(a) of the BridgeBio Disclosure Letter sets forth each In-Scope Program Employee, and for each: (i) name or employee identification number; (ii) job title; (iii) primary work location; (iv) hourly wage or base salary (as applicable); (v) incentive compensation; (vi) exempt or non-exempt status; (vii) active or inactive status (and as applicable, type of leave and anticipated return date); (viii) full-time or part-time status; (ix) visa status (as applicable); (x) date of hire; and (xi) employing entity. Other than the In-Scope Program Employees, there are no employees of BridgeBio or its Subsidiaries who exclusively provide services to the In-Scope Program.

(b) Neither BridgeBio, its Subsidiaries, nor the Company (in each case, with respect to the In-Scope Programs, Acquired Subsidiaries, and Acquired Assets) is party to or bound by any collective bargaining agreement or other Contract with a labor union, labor organization, or other employee representative covering any In-Scope Program Employee, and no In-Scope Program Employee is represented by any labor union, labor organization, employee representative or group of employees with respect to their employment. To the Knowledge of BridgeBio, there have been no union organizing activities affecting In-Scope Program Employees since January 1, 2021. To the Knowledge of BridgeBio, there have been no allegations of sexual harassment or sexual misconduct against any director, officer, or executive at the level of Senior Vice President or above of BridgeBio, its Subsidiaries, or the Company (in each case, in their capacity as such, and with respect to the In-Scope Programs, Acquired Subsidiaries, and Acquired Assets), since January 1, 2021, that, if known to the public, would bring the In-Scope Programs, Acquired Subsidiaries, or Acquired Assets into material disrepute. To the Knowledge of BridgeBio, no In-Scope Program Employee with an annual base salary at or above \$250,000 intends to terminate his or her employment prior to the one (1) year anniversary of the Closing. To the Knowledge of BridgeBio or its Subsidiaries or (ii) owed to any third party with respect to such person's right to be employed or engaged by BridgeBio or its Subsidiaries or the Company or its Affiliates.

Section 4.19 In-Scope Contracts. Section 4.19 of the BridgeBio Disclosure Letter sets forth a list of each In-Scope Program Contract and Shared Contract (each such Contract required to be set forth on Section 4.19 of the BridgeBio Disclosure Letter, an "In-Scope Contract"). Each In-Scope Contract is valid, binding and enforceable on BridgeBio or its Subsidiaries or any of the Acquired Subsidiaries, as the case may be, and, to the Knowledge of BridgeBio, each other party thereto, and is in full force and effect. There is no violation of, or default under, any such In-Scope Contract by BridgeBio or its Subsidiaries or any of the Acquired Subsidiaries, as the case may be, and no event has occurred that, with the lapse of time or the giving of notice or both, would constitute a default thereunder by BridgeBio or its Subsidiaries or any of the Acquired Subsidiaries, as the case may be, or would permit or cause the termination thereof, in each case except as would not have a BridgeBio Material Adverse Effect. Neither BridgeBio or its Subsidiaries or any of the Acquired Subsidiaries have received written notice from any other party to an In-Scope Contract that such party intends to adversely modify, terminate, not renew or renegotiate in any material respect the terms of any such In-Scope Contract. Immediately upon consummation of the transactions contemplated hereby, each In-Scope Contract shall continue in full force and effect on substantially the same terms and conditions as of the date hereof and immediately prior to the Closing. BridgeBio has made available to the Investors true, correct and complete copies of each of the In-Scope Contracts.

- Section 4.20 <u>Affiliate Arrangements</u>. Neither BridgeBio nor any Subsidiary of BridgeBio nor any of their respective Affiliates (other than the Acquired Subsidiaries) directly or indirectly owns, uses, exploits or has any right, title, or interest in or to any asset, property, or right, tangible or intangible, which is owned by the Acquired Subsidiaries or primarily related to the In-Scope Programs. Except as set forth on <u>Section 4.20</u> of the BridgeBio Disclosure Letter, no officer, member of the board of directors (or similar governing body) or employee of BridgeBio or any of its Affiliates (including the Acquired Subsidiaries) is a party to any Contract (other than ordinary course employment arrangements) or transaction with the Acquired Subsidiaries or has any interest in any In-Scope Programs, Acquired Assets, Assumed Liabilities or any other material assets or property used by the Acquired Subsidiaries.
- Section 4.21 <u>Sanctions.</u> None of the BridgeBio, its Subsidiaries or any of respective officers, directors, employees, agents, stockholders or partners is a Sanctioned Party.

# Section 4.22 <u>No Other Representations or Warranties.</u>

- (a) Except for the representations and warranties contained in this Article IV and in the Transaction Documents, none of BridgeBio, the Company, an Acquired Subsidiary or any other Person on their behalf has made, whether express or implied, at law or in equity, with respect to BridgeBio or its businesses, the Company, the Acquired Subsidiaries, the Owned Shares, the Acquired Assets, the probability of success of any In-Scope Program, the Assumed Liabilities, the Excluded Assets or the Excluded Liabilities or with respect to the accuracy or completeness of any statements or information (written or oral) provided or made available to any Investor by or on behalf of BridgeBio or the Company (including in any data room or presentation), and BridgeBio and the Company expressly disclaim any such other representations and warranties (including with respect to any budgets, forecasts, predictions, projections, estimates or other prospective information). Without limiting the foregoing, no express or implied representation or warranty has been made or is being made herein to any Investor and there are no express or implied conditions as to merchantability, suitability or fitness for a particular purpose, or quality, with respect to any of the Acquired Subsidiaries or the Acquired Assets, or as to the condition or workmanship thereof or the absence of any defects therein, whether latent or patent, all of which are being transferred "as is," except for those representations and warranties expressly made by BridgeBio in this Article IV and in the Transaction Documents.
- (b) Notwithstanding anything contained in this Agreement to the contrary, except for the representations and warranties contained in <a href="Article V">Article V</a> and in the Transaction Documents, BridgeBio and the Company acknowledge and agree that none of the Investors or any other Person on behalf of any Investor has made, whether express or implied, at law or in equity, with respect to any Investor or their respective businesses or assets or with respect to the accuracy or completeness of any statements or information (written or oral) provided or made available to BridgeBio or the Company by or on behalf of any Investor (including in any data room or presentation), and BridgeBio and the Company expressly disclaim reliance upon any such other representations and warranties.

# ARTICLE V CERTAIN REPRESENTATIONS AND WARRANTIES OF INVESTORS

Each Investor hereby represents and warrants to BridgeBio as follows:

#### Section 5.1 <u>Organization</u>.

- (a) Each Investor is an entity, duly organized or formed, as applicable, validly existing and in good standing (to the extent applicable) under the Laws of its jurisdiction of incorporation or formation, as applicable.
- (b) Each Investor is duly qualified to do business and is in good standing (to the extent the applicable jurisdiction recognizes such concept) under the Laws of each jurisdiction in which the conduct of its business, or the character of the properties owned or leased by it, requires it to be so qualified, except where the failure to be so qualified or to be in good standing would not have an Investor Material Adverse Effect.
- Section 5.2 <u>Power and Authorization; Enforceability.</u> Each Investor has the requisite corporate or similar power and authority to execute and deliver the Transaction Documents to which it is a party in connection with the transactions contemplated hereby and thereby, and to perform its obligations hereunder and thereunder. Each Investor has taken all actions required to be taken by or on behalf of such Investor to authorize and permit the execution and delivery by such Investor of this Agreement and each Transaction Document required to be executed and delivered by it pursuant hereto, the performance by such Investor of its obligations hereunder and thereunder, and the consummation by such Investor of the transactions contemplated hereby and thereby. This Agreement, and each Transaction Document required to be executed and delivered by such Investor pursuant hereto, has been duly executed and delivered by such Investor and, assuming the due authorization, execution and delivery by each of the other parties hereto or thereto, constitutes the legal, valid and binding obligations of such Investor, enforceable against such Investor in accordance with its terms, except as such enforceability may be limited by the Enforceability Exceptions.
- Section 5.3 <u>Consents and Approvals; No Conflicts.</u> Neither the execution or delivery of this Agreement or any other Transaction Document by such Investor nor the consummation of the transactions contemplated hereby and thereby will, directly or indirectly:
- (a) result in a breach or violation of, or default under, any provision of the Organizational Documents of such Investor;
- (b) assuming the truth, correctness and completeness of the representations and warranties of BridgeBio contained herein, require any Governmental Filings and Approvals, other than Governmental Filings and Approvals that, if not obtained or made, would have an Investor Material Adverse Effect; or
- (c) assuming the truth, correctness and completeness in all material respects of the representations and warranties of BridgeBio contained herein, (i) constitute or result in a violation of any Law or Governmental Order to which such Investor is subject or by which any of its properties or assets is bound; (ii) constitute or result in the violation or breach of any term, condition or provision of, constitute (with or without due notice or lapse of time or both) a default under, or give rise to any right of termination or cancellation, or cause or permit the acceleration of or other changes to any right or obligation or the loss of any benefit for, in each case, any party with respect to any Contract of such Investor; or (iii) result in the creation or imposition of an Encumbrance upon any property or assets of such Investor, except with respect to clauses (i), (ii) and (iii) as would not have an Investor Material Adverse Effect.

- Section 5.4 <u>Compliance with Laws</u>. Since January 1, 2021, none of the Investors or any of its Affiliates that are or will be party to any Transaction Documents are in violation of any Laws or subject to or in violation of any Governmental Orders applicable to them or by which any of their respective material assets is bound or affected, except for violations that would not have an Investor Material Adverse Effect.
- Section 5.5 <u>Litigation</u>. As of the date hereof, there is no Legal Proceeding pending or, to the Knowledge of Investors, threatened against such Investor or any of its Affiliates that, if adversely determined, would have an Investor Material Adverse Effect.
- Section 5.6 <u>Availability of Funds</u>. Each Investor has at the Closing sufficient cash to pay all amounts required to be paid by such Investor at the Closing pursuant to the terms of this Agreement (including the Closing Date Cash Contribution).

#### Section 5.7 Qualified Client; Accredited Investor.

- (a) Each Investor is a qualified client within the meaning of Rule 205-3 under the U.S. Investment Advisers Act of 1940, as amended (the "Advisers Act").
- (b) Each Investor is an accredited investor within the meaning of Rule 501(a) of Regulation D promulgated under the Securities Act of 1933, as amended (the "Securities Act"). Each Investor is not disqualified from reliance on Rule 506 under Regulation D for prior "bad actor" events as set forth in Rule 506(d)(1)(i)-(viii) of Regulation D.
- (c) Each Investor acknowledges that the investment in the Company is intended to be exempt from registration by virtue of Section 4(a)(2) of the Securities Act.
  - Section 5.8 Investment Experience and Ability to Bear Risk; Suitability; Disclosure of Information.
- (a) Each Investor (i) is knowledgeable and experienced with respect to the financial, tax and business aspects of the ownership of a Unit and of the business contemplated by the Company and is capable of evaluating the risks and merits of purchasing a Unit and, in making a decision to proceed with this investment, has not relied upon any representations, warranties or agreements, other than those set forth in <a href="Article IV">Article IV</a> and in the Transaction Documents and (ii) can bear the economic risk of an investment in the Company for an indefinite period of time, and can afford to suffer the complete loss thereof.
- (b) Each Investor is aware (i) that the Company has no operating history; (ii) that the Units involve a substantial degree of risk of loss of its entire investment and that there is no assurance of any income from such Investor's investment; (iii) that any tax benefits which may be available to such Investor may be lost through the adoption of new laws or regulations, to changes to existing laws and regulations and to changes in the interpretation of existing laws and regulations; and (iv) any disposition of Units may result in unfavorable tax consequences to such Investor. Each Investor is relying solely on (x) such Investor's own conclusions or the advice of such Investor's own counsel or investment representative, including with respect to tax aspects of any investment in the Company and (y) the representations and warranties of BridgeBio set forth in Article IV and in the Transaction Documents.

- (c) Each Investor has evaluated the risks involved in investing in the Units and has determined that the Units are a suitable investment for such Investor. The aggregate amount of the investments such Investor has in, and such Investor's commitments to, all similar investments that are illiquid is reasonable in relation to such Investor's net worth, both before and after the subscription for and purchase of the Units pursuant to this Transaction Agreement.
- (d) Each Investor has been given access to, and prior to the execution of this Transaction Agreement, each Investor was provided with an opportunity to ask questions of, and receive answers from, the Company concerning the terms and conditions of the offering of the Units, and to obtain any other information which such Investor and such Investor's investment representative and professional advisors requested with respect to the Company, the In-Scope Programs and such Investor's investment in the Company in order to evaluate such Investor's investment and verify the accuracy of all information furnished to such Investor regarding the Company and the In-Scope Programs. Each Investor acknowledges that all such questions, if asked, were answered satisfactorily and all information or documents provided were found to be satisfactory.

# Section 5.9 <u>Transfers and Transferability</u>.

- (a) Each Investor understands and acknowledges that the Units have not been registered under the Securities Act or any state securities laws and are being offered and sold in reliance upon exemptions provided in the Securities Act and state securities laws for transactions not involving any public offering and, therefore, cannot be resold or transferred unless they are subsequently registered under the Securities Act and such applicable state securities laws or unless an exemption from such registration is available. Each Investor understands and acknowledges that the Company does not have any obligation or intention to register the Units for sale under the Securities Act, any state securities laws, or of supplying the information which may be necessary to enable each Investor to sell Units and that each Investor has no right to require the registration of the Units under the Securities Act, any state securities laws or other applicable securities regulations, except as set forth in the Amended and Restated LLC Agreement. Each Investor understands and acknowledges that sales or transfers of Units are further restricted by the provisions of the Amended and Restated LLC Agreement.
- (b) Each Investor understands and acknowledges that there is no public market for the Units and that any disposition of the Units may result in unfavorable tax consequences to such Investor.
- (c) Each Investor understands and acknowledges that, because of the substantial restrictions on the transferability of the Units, it may not be possible for such Investor to liquidate such Investor's investment in the Company readily, even in the case of an emergency.

Section 5.10 <u>Publicly Traded Company.</u> (a) Each Investor has neither acquired nor will any such Investor transfer or assign any Unit such Investor purchases (or any interest therein) or cause any such Units (or any interest therein) to be marketed on or through an "established securities market" within the meaning of Section 7704(b)(1) of the Code, or to the Knowledge of Investors, a "secondary market" (or the substantial equivalent thereof) within the meaning of Section 7704(b)(2) of the Code, including an over-the-countermarket or an interdealer quotation system that regularly disseminates firm buy or sell quotations, and (b) each Investor either (i) is not, and will not become, a partnership, Subchapter S corporation, or grantor trust for U.S. federal income tax purposes, or (ii) is such an entity, but was not formed and is not used with a principal purpose of permitting the Company to satisfy the 100-partner limitation of Treasury Regulations Section 1.7704-1(h)(1)(ii). Each Investor agrees that if such Investor determines to transfer or assign any of such Investor's Unit pursuant to the provisions hereof and of the Amended and Restated LLC Agreement, such Investor will cause such Investor's proposed transferee(s) to agree to the transfer restrictions set forth herein and therein and to make the representations set forth in clauses (a) and (b) above.

Section 5.11 No Investment Company Issues. If an Investor is an individual, such Investor is a qualified purchaser within the meaning of the Investment Company Act of 1940, as amended (the "Investment Company Act") (a "Qualified Purchaser"). If an Investor is an entity, (a) either (i) such Investor is a Qualified Purchaser within the meaning of the Investment Company Act or (ii) each of such Investor's beneficial owners is a Qualified Purchaser and (b) either (i) all of such Investor's outstanding securities (other than short-term paper) are beneficially owned by one Person, (ii) such Investor is not an investment company under the Investment Company Act or a "private investment company" that avoids registration and regulation under the Investment Company Act based on the exclusion provided by Section 3(c)(1) or Section 3(c)(7) of the Investment Company Act, or (iii) such Investor has delivered to the Company a representation and covenant as to certain matters under the Investment Company Act satisfactory to the Company.

Certain ERISA Matters. With respect to the purchase of a Unit by each Investor, each Investor hereby Section 5.12 represents and warrants that either (a) such Investor is not (and is not acting on behalf or using the assets of) (i) an "employee benefit plan" (within the meaning of Section 3(3) of the U.S. Employee Retirement Income Security Act of 1974, as amended ("ERISA") subject to the fiduciary responsibility provisions of Title I of ERISA, (ii) a "plan" (within the meaning of Section 4975(e)(1) of the Code)) subject to Section 4975 of the Code, (iii) an entity whose underlying assets include "plan assets" by reason of such a plan's investment in the entity (including an insurance company general account) or (iv) a "governmental plan" (within the meaning of Section 3(32) of ERISA) or other employee benefit plan subject to any federal, state, local or non-U.S. law that is substantially similar to the fiduciary responsibility provisions of Title I of ERISA or the provisions of Section 4975 of the Code ("Similar Law") or (b) (i) such Investor's purchase of a Unit will not constitute or result in a prohibited transaction under Section 406 of ERISA, Section 4975 of the Code or any Similar Law for which an exemption is not applicable and otherwise complies with all applicable requirements of ERISA, Section 4975 of the Code and any applicable Similar Law, (ii) neither the Company nor any director, officer, employee, agent or affiliate thereof has acted as a "fiduciary" (within the meaning of Section 3(21) of ERISA, Section 4975(e)(3) of the Code or any Similar Law) in connection with such Investor's purchase of a Unit and (iii) if such Investor is investing assets of an employee benefit plan subject to any Similar Law, such Investor's purchase and holding of a Unit will not cause any assets of the Company to be treated as assets of such employee benefit plan for purposes of such Similar Law and will not otherwise subject the Company to any requirements under such Similar Law.

- Section 5.13 No Intent to Distribute. Each Investor is acquiring the Units for such Investor's own account, as principal, for investment and not with a view to the resale or distribution of all or any part of such Units in violation of any federal or state securities Laws.
- Section 5.14 <u>Sanctions</u>. None of the Investors or any of its officers, directors, employees, agents, stockholders or partners is a Sanctioned Party.
- Section 5.15 <u>Residence</u>. Each Investor maintains such Investor's domicile or principal place of business at the address shown in <u>Section 8.11</u> and such Investor is not merely transient or temporarily resident there.
- Section 5.16 <u>Certain Regulatory Matters</u>. Each Investor hereby agrees and acknowledges that (i) no Person, including none of BridgeBio, the Company or any of their respective Affiliates, is providing under the terms hereof or otherwise as contemplated hereby any investment advisory, brokerage or similar services, (ii) no investment advisory, brokerage or other investment advisory or service relationship is provided for or contemplated hereby, whether discretionary or non-discretionary, for compensation or otherwise, (iii) none of BridgeBio, the Company or any of their respective Affiliates intends as of the date hereof to register or file under any federal, state or other laws pertaining to investment advisory activity, and (iv) no Investor is relying on, may rely on, or intends to rely on any prior or future investment information, advice, or other communication from any of BridgeBio, the Company or any of their respective Affiliates, including with respect to any decision to acquire a Unit.
- Section 5.17 <u>Brokers</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the transactions contemplated by this Agreement based on arrangements made by such Investor and its Affiliates.

#### Section 5.18 No Other Representations or Warranties.

(a) Except for the representations and warranties contained in this Article V and in the Transaction Documents, none of the Investors or any other Person on behalf of any Investor has made, any representation or warranty, whether express or implied, at law or in equity, with respect to any Investor or their respective businesses or assets or with respect to the accuracy or completeness of any statements or information (written or oral) provided or made available to BridgeBio or the Company by or on behalf of any Investor (including in any data room or presentation), and each Investor expressly disclaims any such other representations and warranties (including with respect to any budgets, forecasts, predictions, projections, estimates or other prospective information).

(b) Notwithstanding anything contained in this Agreement to the contrary, except for the representations and warranties contained in Article IV and in the Transaction Documents, each Investor acknowledges and agrees that none of BridgeBio, the Company or any other Person on their behalf has made, with respect to BridgeBio or its businesses, the Company, the Acquired Subsidiaries, the Owned Shares, the Acquired Assets, the probability of success of any In-Scope Program, the Assumed Liabilities, the Excluded Assets or the Excluded Liabilities or with respect to the accuracy or completeness of any statements or information (written or oral) provided or made available to any Investor by or on behalf of BridgeBio or the Company (including in any data room or presentation), and each Investor expressly disclaims reliance upon any such other representations and warranties. Without limiting the generality of the foregoing, each Investor acknowledges and agrees that (i) no representations or warranties are made with respect to any budgets, forecasts, predictions, projections, estimates or other prospective information that may have been provided or made available any Investor (including in any data room or presentation) and (ii) there are no express or implied conditions as to merchantability, suitability or fitness for a particular purpose, or quality, with respect to any of the Acquired Subsidiaries or the Acquired Assets, or as to the condition or workmanship thereof or the absence of any defects therein, whether latent or patent, all of which are being transferred "as is," except for those representations and warranties expressly made by BridgeBio in Article IV and in the Transaction Documents.

# ARTICLE VI ADDITIONAL AGREEMENTS

#### Section 6.1 Further Assurances; Cooperation.

- (a) Subject to the terms and conditions of this Agreement, following the Closing, each Party shall, and shall cause its Affiliates or Subsidiaries to, as promptly as practicable, execute and deliver, or shall cause to be executed and delivered, such documents and other papers and shall take, or shall cause to be taken, such further actions as may be reasonably necessary or required, or as the requesting Party may reasonably deem reasonably necessary or desirable, to carry out the provisions of this Agreement and give effect to the transactions contemplated by the Transaction Documents.
- (b) Each Party shall, and shall cause its Subsidiaries to, use commercially reasonable efforts to cooperate to make and/or obtain any Consents, Governmental Filings and Approvals or Governmental Orders that may be required in connection with the transactions contemplated by the Transaction Documents.

#### Section 6.2 <u>Insurance</u>.

- (a) Each Investor acknowledges and agrees that, upon the Closing, all insurance coverage provided under BridgeBio's insurance policies or otherwise in relation to the Acquired Subsidiaries or the Acquired Assets pursuant to policies, risk funding programs or arrangements maintained by BridgeBio or by any Affiliate of BridgeBio (other than the Acquired Subsidiaries) (whether such policies are maintained in whole or in part with Third Party insurers or with BridgeBio or its Affiliates and including any captive policies or fronting arrangements) shall cease, and no further coverage shall be available under any such policies, programs or arrangements.
- Section 6.3 <u>Tax Treatment</u>. The Parties agree to treat the Closing Date Cash Contributions, the Additional Cash Contributions and the contribution of the Acquired Subsidiaries (including the Acquired Assets) to the Company pursuant to this Agreement as transfers of property described in Section 721(a) of the Code and shall not take any position inconsistent with such treatment on any Tax Return or any audit, examination or other proceeding with respect to Taxes unless otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code.

Section 6.4 <u>Transfer Taxes</u>. All transfer, documentary, sales, use, registration and real property transfer or gains tax, stamp tax, stamp duty, excise tax, stock transfer tax and other similar Taxes with respect to the transactions effected pursuant to this Agreement (collectively, "<u>Transfer Taxes</u>") shall be borne (1) by BridgeBio in the case of any such Transfer Taxes incurred solely as a result of the Pre-Closing Restructuring, and (2) in the case of any other such Transfer Taxes, 50% by BridgeBio and 50% by the Company. All Tax Returns with respect to Transfer Taxes shall be filed by the party required to file the Tax Return under applicable Law, and, if applicable, BridgeBio and/or the Investors shall reimburse such party for the amount of any Transfer Taxes that are paid by it. BridgeBio, the Investors and the Company shall cooperate in timely making all filings, returns, reports and forms as necessary or appropriate to comply with provisions of all applicable Laws in connection with the payment of such Transfer Taxes and shall cooperate in good faith to minimize the amount of any such Transfer Taxes payable in connection herewith.

Section 6.5 <u>Tax Returns</u>. BridgeBio shall include the income of the Acquired Subsidiaries and the Acquired Assets (including any deferred intercompany items described in Treasury Regulations section 1.1502-13 and any excess loss accounts taken into income under Treasury Regulations section 1.1502-19) for all Pre-Closing Tax Periods in the consolidated Tax Returns of BridgeBio for U.S. federal and applicable state income tax for all periods through the close of the Closing Date and shall pay any Taxes attributable to such income. The income of the Acquired Subsidiaries and Acquired Assets for the final Pre-Closing Tax Period shall be determined based on a closing of the books as of the close of the Closing Date.

Section 6.6 Intercompany Accounts and Arrangements. Except as otherwise contemplated by this Agreement and subject to Section 2.2, on or before the Closing Date, BridgeBio shall, and shall cause its Affiliates to, take such actions as are necessary to terminate, eliminate and release, as applicable, with no continuing Liability to BridgeBio, the Company, the Acquired Subsidiaries or any of their respective Affiliates, each and every arrangement, commitment, receivable, payable, claim, demand, right, loan and Contract solely between BridgeBio or one of its Subsidiaries (other than the Company or any of the Acquired Subsidiaries), on the one hand, and the Company or any Acquired Subsidiary, on the other hand, in each case, pursuant to the Intercompany Termination Agreement, and except for the Assumed Intercompany Receivables and Assumed Intercompany Payables which shall remain outstanding and shall be payable in accordance with their terms promptly after the Closing.

## Section 6.7 <u>Director and Officer Indemnification</u>.

(a) For a period of six (6) years from and after the Closing, the Company shall and shall cause each Acquired Subsidiary to, to the fullest extent permitted by applicable Law, indemnify, defend and hold harmless, and provide advancement of reasonably documented out-of-pocket expenses to, each Person who is now, or has been at any time prior to the date hereof or who becomes prior to the Closing, an officer, manager or director of the Company or any Acquired Subsidiary (in such capacity, each, a "D&O Indemnified Party"), against or with respect to all Losses (including any costs of defense or investigation) that may be incurred by any such D&O Indemnified Party that is based in whole or in part on or arising in whole or in part out of the fact that such Person is or was an officer, manager or director of the Company or any Acquired Subsidiary and pertaining to any matter existing or occurring, or any acts or omissions occurring, at or prior to the Closing, whether asserted or claimed prior to, or at or after, the Closing (including matters, acts or omissions occurring in connection with the approval of this Agreement and the consummation of the transactions contemplated hereby), in each case, solely to the extent required by the Organizational Documents of the Company or the Acquired Subsidiaries as of the date hereof. Without limiting the foregoing, the Company shall and shall cause each Acquired Subsidiary, as applicable, to pay all reasonable and documented out-of-pocket expenses, including attorneys' fees, that may be incurred by any D&O Indemnified Party in enforcing the indemnity and other obligations of the Company or any Acquired Subsidiary set forth in or referred to in this Section 6.7, in each case, solely to the extent required by the Organizational Documents of the Company or the Acquired Subsidiaries as of the date hereof.

- (b) Without limiting the foregoing, for a period of six (6) years after the Closing, the Company shall not (and shall not cause or permit any Acquired Subsidiary) amend or modify in any way adverse to the D&O Indemnified Parties, or to the beneficiaries thereof, the exculpation, indemnification, contribution and expense advancement or reimbursement provisions set forth in the Organizational Documents of the Company or the Acquired Subsidiaries with respect to acts or omissions occurring on or prior to the Closing.
- (c) If the Company or any of the Acquired Subsidiaries or any of their respective successors or assigns proposes to (i) consolidate with or merge into any other Person and the Company or any such Acquired Subsidiary will not be the continuing or surviving entity in such consolidation or merger or (ii) transfer all or substantially all of its properties and assets to any Person, then, in each case, proper provision will be made prior to or concurrently with the consummation of such transaction so that the successors and assigns the Company or any such Acquired Subsidiary, as the case may be, will be able to satisfy the indemnification and other obligations set forth in this Section 6.7.
- (d) Notwithstanding <u>Section 8.4</u>, the provisions of this <u>Section 6.7</u> are intended to be for the benefit of, and shall be enforceable by, each D&O Indemnified Party, his or her heirs and his or her representatives and are in addition to, and not in substitution for, any other right to indemnification or contribution that any such Person may have by Contract or otherwise.

#### Section 6.8 Shared Contracts.

BridgeBio and its Subsidiaries are party to certain Contracts as set forth in Section 6.8 of the BridgeBio Disclosure Letter that relate both to the Acquired Assets and the Excluded Assets (each, a "Shared Contract"). Prior to and after the Closing, until earlier of (i) the expiration or termination date of the applicable Shared Contract and (ii) twenty-four (24) months following the Closing Date, each of BridgeBio and the Company shall, and shall cause their respective Subsidiaries to, use their respective commercially reasonable efforts to obtain from, and to cooperate in obtaining from, and shall, and shall cause their respective Subsidiaries to enter into with, or issue notice to (as applicable), each Third Party to a Shared Contract (whether or not listed on Section 4.19 of the BridgeBio Disclosure Letter), either (A) a separate contract or agreement (a "New Contract") or (B) a partial assignment (a "Partial Assignment"), in each case, in a form reasonably acceptable to BridgeBio and the Company that allocates the rights and obligations of BridgeBio and its Subsidiaries and the Company and the Acquired Subsidiaries, as applicable, under each such Shared Contract as between the Acquired Assets, on the one hand, and the Excluded Assets, on the other hand.

(b) Prior to such entry into such New Contract or Partial Assignment in accordance with Section 6.8(a), until earlier of (i) the expiration or termination date of the applicable Shared Contract and (ii) twenty-four (24) months following the Closing Date, each of BridgeBio and the Company shall, and shall cause their respective Subsidiaries to (A) cooperate in good faith and use their respective commercially reasonable efforts (with each party being responsible for its own out-of-pocket expenses) to enter into such New Contract or Partial Assignment in accordance with Section 6.8(a) as promptly as practicable after the Closing and (B) until the entry into such New Contract or Partial Assignment, enter into an arrangement under which the Company (and/or one of its Subsidiaries) or BridgeBio (and/or one of its Subsidiaries) will, in compliance with Law, obtain the economic benefits of, and assume the obligations and bear the economic burdens associated with, the Acquired Assets or the Excluded Assets, respectively, under any such Shared Contract (including by separating, dividing, modifying, replicating or novating such rights and obligations under and in respect of such Shared Contract).

# Section 6.9 <u>Misallocated Assets; Third-Party Claims</u>.

- (a) If at any time within twenty four (24) months after the Closing, if (i) any Excluded Asset or Excluded Liability is found to have been transferred to the Company or any of its Affiliates in error or otherwise than is provided under this Agreement, either directly or indirectly, the Company shall, or shall cause its Affiliates to, promptly notify BridgeBio and transfer or convey to BridgeBio or its designated Affiliate, and BridgeBio or its designated Affiliate shall accept, such Excluded Asset or Excluded Liability, as applicable, as soon as practicable, or (ii) any Acquired Asset or Assumed Liability is found to have been retained or incurred by BridgeBio or any of its Affiliates in error or otherwise than is provided under this Agreement, either directly or indirectly, BridgeBio shall, or shall cause its Affiliates to, promptly notify the Company and transfer and convey to the Company or its designated Affiliate, and the Company or its designated Affiliate shall accept, such Acquired Asset or Assumed Liability, as applicable, as soon as practicable, in each case, for no consideration therefor. Reasonable and documented out-of-pocket costs and expenses of the foregoing transfers and reimbursements shall be borne by the Company in the case of Excluded Assets and (ii) by BridgeBio in the case of Acquired Assets.
- (b) If at any time within twenty-four (24) months after the Closing, if (i) the Company or any of its Affiliates receives a payment, invoice or material document from a Third Party that, pursuant to the terms of this Agreement or any other Transaction Document, should have been paid or transmitted to BridgeBio or any of its Affiliates, the Company agrees to, or to cause its applicable Affiliate to, promptly notify BridgeBio and hold in trust and promptly deliver such payment, invoice or material document to BridgeBio or its designated Affiliate; or (ii) BridgeBio or any of its Affiliates receives a payment, invoice or material document from a Third Party that, pursuant to the terms of this Agreement or any other Transaction Document, should have been paid or transmitted to the Company or any of its Affiliates, BridgeBio agrees to, or to cause its applicable Affiliate to, promptly notify the Company and hold in trust and promptly deliver such payment, invoice or material document to the Company.

- (c) If at any time after the Closing, the Company or any of its Subsidiaries receives a claim, invoice or demand from a Third Party with respect to any amount that constitutes an Excluded Accounts Payable, the Company shall promptly notify BridgeBio, and BridgeBio shall promptly pay, or shall cause to be paid, such amount to such Third Party (and promptly confirm payment of the same to the Company), or if such amount has already been paid by the Company, BridgeBio shall promptly reimburse the Company for any such amount.
- (d) If at any time after the Closing, BridgeBio or any of its Subsidiaries receives a claim, invoice or demand from a Third Party with respect to any amount that constitutes an Assumed Accounts Payable, BridgeBio shall promptly notify the Company, and the Company shall promptly pay, or shall cause to be paid, such amount to such Third Party (and promptly confirm payment of the same to BridgeBio), or if such amount has already been paid by the BridgeBio or any of its Subsidiaries, the Company shall promptly reimburse BridgeBio or the applicable Subsidiary for any such amount.

## Section 6.10 <u>Employee Matters</u>.

- Prior to the end of the period contemplated by the Transition Services Agreement (such period, the "<u>Transition Period</u>"), the Company shall, or shall cause one of its Affiliates, to make an offer of employment to each In-Scope Program Employee, which offer of employment shall be effective as of the first day following the end of the Transition Period (the "<u>Transfer Date</u>"). Each In-Scope Program Employee who accepts such offer of employment and actually commences employment with the Company or any of its Affiliates as of the Transfer Date shall be a "<u>Transferred Employee</u>." Notwithstanding anything to the contrary in this <u>Section 6.10</u>, if any In-Scope Program Employee is not actively at work and is receiving, or in a waiting or elimination period for, short-term or long-term disability benefits as of immediately prior to the Transfer Date pursuant to an Employee Plan (each, an "<u>Inactive Employee</u>") such Inactive Employee will remain employed with BridgeBio or any of its Affiliates as of the Transfer Date. If such Inactive Employee, within six (6) months following the Transfer Date, is able to return to work, then immediately upon such return, BridgeBio shall so notify the Company, and the Company shall, or shall cause one of its Affiliates to, offer employment to such employee. With respect to each Inactive Employee who commences employment with the Company, references to the "<u>Transfer Date</u>" in this Agreement shall be deemed to refer to the date that such Inactive Employee commences employment with the Company.
- (b) From and after the Closing Date, the Company shall honor, fulfill and fully satisfy any and all obligations for the year in which the Closing Date occurs for any cash bonus, commission or other incentive compensation Employee Plans in which any Transferred Employee participates immediately prior to such Transferred Employee's Transfer Date (collectively, the "Bonus Plans"), whether the obligation with respect thereto arises before, after or as a result of the Closing Date, in accordance with the terms of such Bonus Plans (the aggregate amount of such payments, the "Closing Year Bonus Payments"). As soon as practicable following the payment of the Closing Year Bonus Payments to Transferred Employees (which shall be paid no later than March 15, 2025), the Company shall provide BridgeBio with a full statement of such payments made to Transferred Employees. Within sixty (60) days following receipt of such statement, BridgeBio will reimburse the Company an amount equal to seven-twelfths (7/12th) of (i) the Closing Year Bonus Payments.

- (c) Effective as of the Transfer Date, any unvested equity or equity-based incentive awards of BridgeBio held by Transferred Employees will be forfeited as a result of the transfer of such Transferred Employees' employment pursuant to this Agreement, and BridgeBio shall retain all Liabilities with respect to any equity or equity-based incentive awards of BridgeBio held by any In-Scope Program Employee (including any Transferred Employee). Following the Transfer Date, the Company shall be solely responsible for the payment of any incentive compensation to Transferred Employees.
- (d) Except for Excluded Liabilities or as otherwise specifically set forth in this <u>Section 6.10</u>, the Company shall assume (i) all Liabilities with respect to the Company's or its Affiliate's employment or termination of the Transferred Employees on or after the Transfer Date, and (ii) all Liabilities otherwise set forth in the Transition Services Agreement (the "<u>Assumed Employee Liabilities</u>").
- (e) Prior to the Transfer Date, BridgeBio shall, or shall cause its applicable Affiliate to, fully vest each In-Scope Program Employee in his or her account balances under each Employee Plan intended to be qualified under Section 401(a) of the Code, effective as of the Transfer Date, and make or cause to be made all employee and employer contributions to such Employee Plan on behalf of In-Scope Program Employees had the transactions contemplated by this Agreement not occurred (regardless of any service or end-of-year employment requirements) but prorated for the portion of the plan year prior to the Transfer Date.
- BridgeBio shall, or shall cause its Affiliates to, use their commercially reasonable efforts to ensure that any foreign national In-Scope Program Employee who requires a visa in order to work for BridgeBio or its Affiliates in his or her current position may continue to work in such position as a Transferred Employee on the Transfer Date. In the event such In-Scope Program Employee is not authorized to work for the Company as of the Transfer Date, such In-Scope Program Employee shall become a "Delayed Transfer Employee." From and after the Transfer Date and until the earlier of (i) the date each Delayed Transfer Employee becomes employed by the Company or (ii) a date to be mutually agreed by the Parties (such period, the "Delayed Transfer Employee Term"), BridgeBio and its Affiliates shall make the services of such Delayed Transfer Employee available to the Company until the applicable visa is obtained. During such Delayed Transfer Employee Term, Delayed Transfer Employees shall remain employees of BridgeBio or its applicable Affiliate, subject to any Delayed Transfer Employee's termination of employment due to termination for cause, resignation, death or disability. For and in consideration for such services from BridgeBio and its Affiliates to the Company, the Company shall pay and deliver to BridgeBio and its Affiliates a monthly payment equal to the total amount of compensation, employee benefit expenses, employment taxes, liability insurance costs and overhead expenses (in each case, to the extent reasonable and documented) actually incurred by BridgeBio and its Affiliates for each Delayed Transfer Employee providing services to the Company for the immediately preceding month; provided, however, such amounts may not exceed the total amount of compensation, employee benefit expenses employment taxes, liability insurance costs and overhead expenses actually incurred by BridgeBio and its Affiliates for each such Delayed Transfer Employee immediately prior to the Transfer Date.

- (g) During the period prior to the Closing Date, BridgeBio shall, and shall cause its Affiliates to, use commercially reasonable efforts to make independent contractors who provide services to the In-Scope Programs and are directly engaged by BridgeBio or its Affiliates available to the Company or its Affiliates for the purpose of allowing the Company to interview each such contractor and determine the nature and extent of each such person's continuation with the Company and its applicable Affiliates, if any.
- (h) Nothing contained in this <u>Section 6.10</u>, express or implied, (i) is intended to confer upon any In-Scope Program Employee or any other Person any right to continued employment or any particular term or condition of employment for any period, (ii) will prohibit or limit the ability of the Company or any of its respective Affiliates from amending, modifying or terminating any benefit or compensation plan, program, policy, Contract, agreement or arrangement at any time assumed, established, sponsored or maintained by any of them, or (iii) will constitute an amendment to or any other modification of any Acquired Subsidiary Plan or Employee Plan or other benefit or compensation plan, program, policy, Contract, agreement or arrangement. Further, this <u>Section 6.10</u> will be binding upon and inure solely to the benefit of each of the Parties to this Agreement, and nothing in this <u>Section 6.10</u>, express or implied, is intended to confer upon any other Person any rights or remedies of any nature whatsoever (including any third-party beneficiary rights) under or by reason of this <u>Section 6.10</u>.
- Section 6.11 Transitional Trademark License. Following the Closing, the Company and its Affiliates shall, as soon as practicable, but in no event later December 31, 2024, cease to make any use of or to permit any third party to make any use of any BridgeBio Marks. Any use by the Company of any of the BridgeBio Marks as permitted in this Section 6.11 is subject to their use of the BridgeBio Marks in a form and manner, and with standards of quality, of that in effect for the BridgeBio Marks as of the Closing Date; provided, that neither the Company nor any of its Affiliates shall use the BridgeBio Marks for sales or marketing purposes in any event. The Company and its Affiliates shall not use the BridgeBio Marks in a manner that may reflect negatively on such name and marks or on BridgeBio or its Affiliates. BridgeBio shall have the right to terminate the foregoing, effective immediately, if the Company and its Affiliates fail to comply with the foregoing terms and conditions or otherwise fail to comply with any reasonable direction of BridgeBio in relation to the use of the BridgeBio Marks. Notwithstanding the foregoing or anything to the contrary, nothing in this Agreement will prohibit the Company or any of its Subsidiaries from (x) retaining and using, solely for its internal business purposes, records and other historical or archived documents containing or referencing the BridgeBio Marks and (y) using the BridgeBio Marks in a descriptive or factually accurate manner where such use constitutes permitted non-trademark uses under applicable Law, including uses not in commerce, uses that would not cause confusion as to the origin or sponsorship of a good or service and uses in historical, tax and similar records.

#### Section 6.12 <u>Intellectual Property Cross-Licenses</u>.

- Effective as of the Closing, BridgeBio hereby grants, and shall cause its Subsidiaries to grant, to the Company and its Subsidiaries (a) a perpetual, irrevocable, non-exclusive, non-transferable (except pursuant to Section 6.12(c)), worldwide, sublicensable (in multiple tiers for use in connection with the conduct of any In-Scope Program or any other business of the Company or its Subsidiaries permitted by the Transaction Agreements but not for the independent use by Third Parties), fully paid up, royalty-free license under any Know-How owned by BridgeBio or its Subsidiaries as of the Closing to the extent both Controlled by BridgeBio or its Subsidiaries and used in any In-Scope Program as of the Closing, solely to the extent necessary or reasonably useful for the Company and its Subsidiaries to conduct any In-Scope Program or any other business of the Company or its Subsidiaries permitted by the Transaction Documents, including to make, have made, import, export, use, have used, sell, have sold, offer for sale and otherwise exploit any compound or product under any such program or in any such business and (b) a perpetual, irrevocable, non-exclusive, non-transferable (except pursuant to Section 6.12(c)), worldwide, sublicensable (in multiple tiers for use in connection with the conduct of any In-Scope Program but not for the independent use by Third Parties), fully paid up, royalty-free license under any Patents Controlled by BridgeBio or its Subsidiaries as of the Closing to the extent practiced in any In-Scope Program, including to make, have made, import, export, use, have used, sell, have sold, offer for sale and otherwise exploit any compound or product under any such In-Scope Program.
- (b) Effective as of the Closing, the Company hereby grants, and shall cause its Subsidiaries to grant, to BridgeBio and its Subsidiaries a perpetual, irrevocable, non-exclusive, non-transferable (except pursuant to Section 6.12(c)), worldwide, sublicensable (in multiple tiers for use in connection with the conduct of the business of BridgeBio or any of its Subsidiaries but not for the independent use by Third Parties), fully paid up, royalty-free license under any Know-How owned by the Company or its Subsidiaries as of the Closing to the extent both Controlled by the Company or its Subsidiaries and used in the business of BridgeBio or its Subsidiaries as of the Closing, solely to the extent necessary or reasonably useful for BridgeBio and its Subsidiaries to conduct any business of BridgeBio or its Subsidiaries permitted under the Transaction Documents, including to make, have made, import, export, use, have used, sell, have sold, offer for sale and otherwise exploit any compound or product in any such business.
- (c) The foregoing licenses are assignable, in whole or in part, in connection with the sale or transfer of the business or assets of BridgeBio, the Company or their respective Subsidiaries to which such license relates; provided that such licenses shall not be deemed to extend to any other businesses of any such buyer or transferee.
- Section 6.13 Non-Solicitation. BridgeBio agrees that, for the period commencing on the Closing Date and expiring on the fifth (5<sup>th</sup>) anniversary of the Closing Date, neither it nor any of its controlled Affiliates shall directly or indirectly solicit for employment or any similar arrangement, hire, employ or engage (in any capacity) any In-Scope Program Employee (including any Transferred Employee) of the Acquired Subsidiaries; <u>provided</u>, <u>however</u>, that this <u>Section 6.13</u> shall not prohibit general solicitations for employment through advertisements or other means not specifically directed toward employees of the Company or its Subsidiaries.

## Section 6.14 <u>Non-Competition</u>.

- (a) BridgeBio agrees that for the period commencing on the Closing Date and expiring on the fifth (5th) anniversary of the Closing Date, neither it nor any of its controlled Affiliates shall directly or indirectly (including by way of licensing any Person) own any interest in, engage in, manage, operate, develop or otherwise pursue the following programs or indications: Erythropoietic Protoporphyria (EPP), Alpha-1 antitrypsin deficiency (A1AT) and small molecule method for Tuberous Sclerosis Complex (TSC), and in each case, next generation applications of the foregoing (together, a "Company Competing Program"); provided that nothing in this Section 6.14(a) shall preclude BridgeBio and its controlled Affiliates from (i) owning up to a 5.00% passive interest in any Person engaged (in any manner) in a Company Competing Program and to the extent such investment does not give BridgeBio or such Affiliates the right to designate a member of the board of directors (or similar governing body) of such Person and (ii) BridgeBio's ownership and involvement in the joint venture with the Company as contemplated by the Transaction Documents.
- (b) The Company agrees that for the period commencing on the Closing Date and expiring on the fifth (5th) anniversary of the Closing Date, neither it nor any of its controlled Affiliates shall directly or indirectly own any interest in, engage in, manage, operate, develop or otherwise pursue the following programs or indications: Transthyretin Amyloidosis (ATTR), Autosomal Dominant Hypocalcemia Type 1 (ADH1), Achondroplasia and other skeletal dysplasias, Limb-Girdle Muscular Dystrophy Type 2I (LGMD2I), Congenital Adrenal Hyperplasia (CAH) and gene therapy method for Tuberous Sclerosis Complex (TSC), and in each case, next generation applications of the foregoing (together, a "BridgeBio Competing Program"); provided that nothing in this Section 6.14(b) shall preclude the Company and its controlled Affiliates from owning up to a 5.00% passive interest any Person engaged (in any manner) in a BridgeBio Competing Program and to the extent such investment does not give the Company or such Affiliates the right to designate a member of the board of directors (or similar governing body) of such Person.
- (c) The Parties acknowledge that the restrictions set forth in Section 6.13 and this Section 6.14 are reasonable in scope and duration and in all other respects and are intended to comply with all applicable Laws. The Parties further acknowledge that the restrictions set forth in Section 6.13 and this Section 6.14 are necessary to protect the Investors' significant investment in the Acquired Assets, including its goodwill, and that the Investors would not enter into this Agreement without the restrictions contained in Section 6.13 and this Section 6.14. It is the desire and intent of the Parties that the provisions of Section 6.13 and this Section 6.14 be enforced to the fullest extent permissible under applicable Law. Therefore, the Parties agree that money damages would not be a sufficient remedy for any threatened or actual breach of Section 6.13 and this Section 6.14, and that, in addition to all other remedies it may be entitled to, the non-breaching Party shall be entitled to seek specific performance, or injunctive or other equitable relief, as a remedy for any such breach. To the extent that any part of Section 6.13 and this Section 6.14, as applicable, may be invalid, illegal or unenforceable for any reason, it is intended that such part shall be deemed amended to the extent (and only to such extent) necessary in order that such provision be valid and enforceable, and the remainder of such sections shall not thereby be affected and shall be given full force and effect without regard to invalid portions and such amendment shall apply only with respect to the operation of the covenant in the particular jurisdiction in which such adjudication is made.

#### Section 6.15 Release.

- (a) Effective as of the Closing, BridgeBio, on behalf of itself and its Affiliates, and its and their respective individual, joint or mutual, past, present and future successors, assigns, representatives, administrators, executors, beneficiaries, agents, Affiliates, equityholders, Subsidiaries, officers, directors, employees, partners, members, managers, general or limited partners and advisors (collectively, the "BridgeBio Releasing Parties"), hereby unconditionally and irrevocably waives releases, remises and forever discharges any rights, claims and Losses of any type that it or any of the BridgeBio Releasing Parties has had, now has or might now or hereafter have against the Company and any Acquired Asset, and each of the Company's individual, joint or mutual, past, present and future representatives, equityholders, Affiliates, Subsidiaries, officers, directors, employees, partners, members, managers, agents, general or limited partners, advisors, successors and assigns in respect of, relating to or arising in connection with any actions or failures to take actions on or prior to the Closing Date (including in connection with the ownership and operation of the Company and the Acquired Assets), except for (i) rights, claims and Losses arising from and after the date hereof under the terms of this Agreement or any Transaction Document or (ii) for Fraud. Each Party, for itself and the BridgeBio Releasing Parties, acknowledges that it is aware that it or such BridgeBio Releasing Party may hereafter discover facts different from or in addition to the facts which it or such BridgeBio Releasing Party now knows or believes to be true with respect to the subject matter of this Agreement, but that it or such BridgeBio Releasing Party intends that the general releases herein given shall be and remain in full force and effect, notwithstanding the discovery of any such different or additional facts, other than in the case of Fraud. The Parties acknowledge that this Section 6.15(a) is not an admission of liability or of the accuracy of any alleged fact or claim. The Parties expressly agree that this Section 6.15(a) shall not be construed as an admission in any proceeding as evidence of or an admission by any Party of any violation or wrongdoing.
- (b) Effective as of the Closing, the Company, on behalf of itself and its Affiliates, and its and their respective individual, joint or mutual, past, present and future successors, assigns, representatives, administrators, executors, beneficiaries, agents, Affiliates, equityholders, Subsidiaries, officers, directors, employees, partners, members, managers, general or limited partners and advisors (collectively, the "Company Releasing Parties"), hereby unconditionally and irrevocably waives releases, remises and forever discharges any rights, claims and Losses of any type that it or any of the Company Releasing Parties has had, now has or might now or hereafter have against BridgeBio or any of its Subsidiaries, and each of their respective individual, joint or mutual, past, present and future representatives, equityholders, Affiliates, Subsidiaries, officers, directors, employees, partners, members, managers, agents, general or limited partners, advisors, successors and assigns in respect of, relating to or arising in connection with any actions or failures to take actions on or prior to the Closing Date (including in connection with the ownership and operation of the Company and the Acquired Assets), except for rights, claims and Losses (i) arising from and after the date hereof under the terms of this Agreement or any Transaction Document or (ii) for Fraud. Each Party, for itself and the Company Releasing Parties, acknowledges that it is aware that it or such Company Releasing Party may hereafter discover facts different from or in addition to the facts which it or such Company Releasing Party now knows or believes to be true with respect to the subject matter of this Agreement, but that it or such Company Releasing Party intends that the general releases herein given shall be and remain in full force and effect, notwithstanding the discovery of any such different or additional facts, other than in the case of Fraud. The Parties acknowledge that this Section 6.15(b) is not an admission of liability or of the accuracy of any alleged fact or claim. The Parties expressly agree that this Section 6.15(b) shall not be construed as an admission in any proceeding as evidence of or an admission by any Party of any violation or wrongdoing.

Section 6.16 <u>Prepaid Expenses</u>. The Company shall reimburse the prepaid expenses, up to a cap of \$1,200,000, comprising the "Total Assets" line item in the Financial Statements or that arise after the date of the Financial Statements but prior to the Closing and are of a type that would be reflected on a balance sheet of Portal or Sub21 in the Financial Statements had they existed at June 30, 2024 to BridgeBio at Closing.

# ARTICLE VII SURVIVAL; INDEMNIFICATION

- Section 7.1 <u>Survival</u>. None of the representations and warranties of the Parties in this Agreement or in any instrument delivered by the Parties pursuant to this Agreement shall survive the Closing, except in respect of claims for Fraud. The covenants, agreements and other obligations of the Parties set forth in this Agreement that contemplate performance following the Closing shall survive the Closing in full force and effect until fully performed in accordance with their terms. Each covenant in this Agreement that by its terms applies or is required to be performed in its entirety at or before the Closing shall survive until the occurrence of the Closing and shall thereupon terminate.
- Section 7.2 <u>Indemnification by BridgeBio</u>. Subject to <u>Section 7.4</u>, from and after the Closing Date, BridgeBio shall indemnify, defend, hold harmless and reimburse the Company and its Affiliates and their respective directors, officers, employees, partners, members, managers, general or limited partners, representatives and advisors (the "<u>Company Indemnified Parties</u>") against, and hold them harmless from, and pay and reimburse such parties for, any Losses to the extent such Losses arise from or in connection with any and all Excluded Liabilities.
- Section 7.3 <u>Indemnification by the Company</u>. Subject to <u>Section 7.4</u>, from and after the Closing Date, the Company shall indemnify, defend, hold harmless and reimburse BridgeBio and its Affiliates and their respective directors, officers, employees, representatives and advisors (the "<u>BridgeBio Indemnified Parties</u>") against, and hold them harmless from, and pay and reimburse such parties for, any Losses to the extent such Losses arise from or in connection with any and all Assumed Liabilities.
- Limitations on Indemnification. Each Person entitled to indemnification under this Article VII (an "Indemnified Section 7.4 Party") shall use commercially reasonable efforts to mitigate all Losses that are indemnifiable or recoverable hereunder or in connection herewith after becoming aware of any event that could reasonably be expected to give rise to such Losses (including by taking all reasonable actions to recover reasonably available proceeds under any Contract or insurance). The amount of any Losses for which either BridgeBio or the Company, as the case may be, is liable under this Article VII shall be reduced by (i) any amount actually received by the Indemnified Party pursuant to any indemnification by or indemnification agreement with any Third Party, (ii) the amount of any insurance proceeds actually paid to the Indemnified Party in respect of such Losses as an offset against such Losses (net of out-of-pocket expenses reasonably incurred in obtaining such recovery and the amount of any retrospective or other current increase in insurance premiums to the extent attributable to the payment of such cash recovery or the existence of such Losses), and (iii) any net Tax benefits actually realized by any applicable Acquired Subsidiary as a result of the events or circumstances giving rise to the incurrence, accrual or payment of the Losses that such Indemnified Party is to recover from the indemnifying Party, calculated on a "with and without" basis. If the amount to be netted hereunder from any payment required under Section 7.2 or Section 7.3 is determined after payment by the indemnifying Party ("Indemnifying Party") of any amount otherwise required to be paid to an Indemnified Party pursuant to this Article VII, the Indemnified Party shall repay to the Indemnifying Party, promptly after such determination, any amount that the Indemnifying Party would not have had to pay pursuant to this Article VII had such determination been made at the time of such payment.

#### Section 7.5 Procedure.

- (a) Any Person seeking indemnification provided for under this Article VII in respect of, arising out of or involving a claim made by any Person (other than a Party hereto) against an Indemnified Party (a "Third-Party Claim"), shall promptly notify the Indemnifying Party in writing of the Third-Party Claim; provided that failure to give such notice shall not affect the right to indemnification provided hereunder except to the extent the Indemnifying Party shall have been actually and materially prejudiced as a result of such failure. Thereafter, the Indemnified Party shall deliver to the Indemnifying Party, as promptly as reasonably practicable following such Indemnified Party's receipt thereof, copies of all written notices and documents (including any court papers) received by such Indemnified Party relating to the Third-Party Claim.
- (b) If a Third-Party Claim is made against an Indemnified Party, the Indemnifying Party shall be entitled at its election and its cost to assume the defense of such Third-Party Claim with counsel selected by the Indemnifying Party; provided that such counsel is reasonably acceptable to the Indemnified Party. If the Indemnifying Party assumes such defense, the Indemnified Party shall nonetheless have the right to employ counsel separate from the counsel employed by the Indemnifying Party; provided that the Indemnifying Party shall not be liable to such Indemnified Party for any fees of such separate counsel with respect to the defense of such Third-Party Claim, unless the employment and reimbursement of such separate counsel is authorized by the Indemnifying Party in writing. If the Indemnifying Party does not assume such defense, and for any period during which the Indemnifying Party has not assumed such defense, the Indemnifying Party shall be liable for the reasonable fees and expenses of one single counsel (in addition to reasonable fees and expenses of local counsel required in jurisdictions not central to the Third-Party Claim) employed (and reasonably acceptable to the Indemnifying Party) by such Indemnified Party (which reasonable fees and expenses shall be considered Losses for purposes of this Agreement). If the Indemnifying Party chooses to defend a Third-Party Claim or prosecute a claim in connection therewith, each Indemnified Party shall provide all cooperation as is reasonably requested by the Indemnifying Party in such defense or prosecution.
- (c) Notwithstanding anything to the contrary in this <u>Section 7.5</u>, no Party may settle, compromise or discharge (and in doing so, make any reasonable admission of Liability with respect to) such Third-Party Claim without the prior written consent of the other Party, which consent shall not be unreasonably withheld, conditioned or delayed.
- (d) An Indemnifying Party shall not be entitled to assume or continue control of the defense of any Third-Party Claim if (i) the Third-Party Claim (A) relates to or arises in connection with any criminal Legal Proceeding, or (B) seeks an injunction or other equitable relief against any Indemnified Party, or (ii) the outcome of any judgment or settlement of such Third-Party Claim could reasonably be expected to materially and adversely affect the ability of the Indemnified Party to conduct its or its Affiliates' businesses.

- (e) In the event an Indemnified Party has a claim against an Indemnifying Party under Section 7.2 or Section 7.3 that does not involve a Third-Party Claim, such Indemnified Party shall deliver notice of such claim to the Indemnifying Party (i) stating the amount of the Loss, if known, and method of computation thereof, (ii) giving reasonable supporting documentation relating to such Loss and (iii) containing a reference to the provisions of this Agreement in respect of which such right of indemnification is claimed to arise, promptly after becoming aware of the facts or circumstances giving rise to such claim; provided that failure to give such notice shall not affect the indemnification provided hereunder except to the extent the Indemnifying Party shall have been actually and materially prejudiced as a result of such failure. The Indemnified Party and the Indemnifying Party shall, for a period of not less than sixty (60) days following receipt by the Indemnifying Party of the notice of such claim, negotiate, in good faith, to resolve the claim, and such Indemnified Party shall not commence Proceedings with respect to such claim prior to the end of such period.
- (f) All amounts owed by an Indemnifying Party under this <u>Article VII</u> shall be paid promptly (but in any event within five Business Days from the date of determination of such amounts owed) by the Indemnifying Person through wire transfer of immediately available funds to the account designated in writing by the Indemnified Party entitled to such payment.
- Section 7.6 <u>Exclusive Remedies</u>. Other than (a) with respect to any right a Party may have to injunctive relief or specific performance, or (b) in the case of Fraud, the Parties acknowledge and agree that following the Closing the indemnification provisions of this <u>Article VII</u> shall be the sole and exclusive remedies of the Indemnified Parties pursuant to this <u>Article VII</u> for any claims for monetary relief by any Party; <u>provided</u> that, for the avoidance of doubt, this <u>Section 7.6</u> shall not limit any rights or remedies of any Member or the Company under the LLC Agreement.

# ARTICLE VIII MISCELLANEOUS

Section 8.1 Expenses. Except as otherwise expressly provided herein (including the reimbursement obligation in the following sentence) and in the Transaction Documents, all costs and expenses, including fees and disbursements of legal counsel, financial advisors, accountants, consultants and other representatives and costs and expenses relating to the preparation, negotiation and execution of this Agreement and the other Transaction Documents and the transfer of the Acquired Subsidiaries or the Acquired Assets, incurred in connection with this Agreement and the other Transaction Documents and the transactions contemplated hereby and thereby (the "Transaction Expenses") that is incurred by BridgeBio, the Company or the Acquired Subsidiaries shall be borne and paid by BridgeBio. Notwithstanding anything contained in this Agreement to the contrary, the Company shall pay, or cause to be paid, on the Closing Date (or as soon as practicable thereafter) to or on behalf of each Investor (or any of their respective Affiliates) and BridgeBio (or any of its Affiliates) their incurred (whether or not paid) reasonable and documented out-of-pocket Transaction Expenses to the respective account designated by each Investor and BridgeBio in writing (in each case, which account shall be designated at least one Business Day prior to the Closing Date); provided that such Transaction Expenses of BridgeBio shall exclude fees and expenses incurred by the Special Committee in connection with its engagement of Goldman Sachs & Co. LLC, Goodwin and RTF.

- Section 8.2 Amendment; Waiver. No amendment, supplement or modification of this Agreement shall be effective unless each of BridgeBio and the Company has signed a written instrument expressly referencing this Agreement and the specific provisions hereof that are intended to be so amended, supplemented or modified, and consenting to such amendment, supplement or modification. Any purported amendment, supplement or modification that fails to comply with the foregoing shall be null and void ab initio. Any failure by any such Party to comply with any obligation, covenant, agreement or condition herein may be waived by the other such Parties only by a written instrument signed by such Parties granting such waiver, and such waiver shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure to comply with this Agreement. No failure or delay on the part of any Party in the exercise of any right, power or privilege hereunder shall impair such right, power or privilege or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty or covenant herein, nor shall any single or partial exercise of such right, power or privilege preclude other or further exercise thereof or any other right, power or privilege.
- Section 8.3 <u>Assignment; Successors and Assigns</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any Party (whether by merger, operation of law or otherwise) without the prior written consent of the other Parties and any purported assignment in violation of this <u>Section 8.3</u> shall be void; <u>provided</u> that without the prior written consent of any Party hereto, each Investor and BridgeBio may assign any of its rights or obligations (in whole or in part) under this Agreement to one or more of its respective Affiliates without prior written consent of any other Party; provided that any assignment of obligations shall only release the assigning party from such obligations to the extent actually performed. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by the Parties and their respective successors and permitted assignees.
- Section 8.4 No Third-Party Beneficiaries. Except with respect to (a) D&O Indemnified Parties solely with respect to Section 6.7 and (b) BridgeBio Indemnified Parties and the Company Indemnified Parties solely with respect to Article VII, this Agreement is for the sole benefit of BridgeBio, the Company, the Investors and the Investors' Representatives and their successors and permitted assigns, and nothing in this Agreement or any other Transaction Document is intended to or shall confer upon any other Person any legal or equitable right, claim, benefit or remedy of any nature whatsoever whether under Law or otherwise.
  - Section 8.5 Governing Law; Venue; Waiver of Jury Trial.
- (a) This Agreement, and all claims or causes of action (whether in contract or tort) that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement, shall be governed by and construed in accordance with the Laws of the State of Delaware, without giving effect to the principles of conflicts of law thereof.

- (b) The Parties irrevocably submit to the jurisdiction of the Court of Chancery of the State of Delaware and any state appellate court therefrom within the State of Delaware (or, if the Court of Chancery of the State of Delaware declines to accept jurisdiction over a particular matter, any federal court within the State of Delaware (and any appellate court therefrom)) (collectively, the "Chosen Courts") in any action or Legal Proceeding that arises in respect of the interpretation and enforcement of the provisions of this Agreement and the documents referred to in this Agreement or in respect of the transactions contemplated herein, and hereby waive, and agree not to assert, as a defense in any action, suit or Legal Proceeding for interpretation or enforcement hereof or any such document that it is not subject thereto or that such action, suit or Legal Proceeding may not be brought or is not maintainable in the Chosen Courts or that venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such Chosen Courts, and the Parties irrevocably agree that all claims with respect to such action, suit or Legal Proceeding shall be heard and determined exclusively by such Chosen Courts.
- (c) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY BE BASED UPON, ARISE OUT OF OR RELATE TO THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY BASED UPON, ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREIN. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT: (I) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER; (II) SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THE FOREGOING WAIVER; (III) SUCH PARTY MAKES THE FOREGOING WAIVER VOLUNTARILY; AND (IV) SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVER AND CERTIFICATIONS IN THIS SECTION 8.5.
- Section 8.6 Specific Performance. Each Party acknowledges and agrees that the other Parties will be irreparably damaged if this Agreement is not performed in accordance with its terms and that any breach of this Agreement and the non-consummation of the transactions contemplated by this Agreement would not be adequately compensated in all cases by monetary damages alone. Accordingly, in addition to any other right or remedy to which a Party may be entitled, at Law or in equity, the Parties expressly consent to the granting of an injunction, specific performance or other equitable relief to prevent breaches or threatened breaches of this Agreement and to enforce specifically this Agreement, without the need for proof of actual damages or any requirement under any Law to post a bond or other security as a prerequisite to obtaining equitable relief, in accordance with its terms and to require any other Party to consummate the Closing and perform its obligations hereunder.
- Section 8.7 Severability. If any provision of this Agreement or the application of any provision hereof to any circumstances is held invalid, unenforceable, or otherwise illegal by a court of competent jurisdiction, the remainder of this Agreement and the application of such provision to other circumstances shall not be affected. Upon any such determination that any term or other provision is invalid, unenforceable, or otherwise illegal, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the greatest extent possible.

Section 8.8 <u>Counterparts</u>; <u>Electronic Delivery</u>. This Agreement may be executed and delivered in two or more separate counterparts, each of which when executed and delivered shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. This Agreement, the agreements referred to herein, and each other agreement, consent or instrument entered into in connection herewith or therewith or contemplated hereby or thereby, and any amendments hereto or thereto, to the extent signed and delivered by means of a photographic, portable document format (.pdf), or similar reproduction of such signed writing using electronic mail shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person.

Section 8.9 Descriptive Headings; Interpretation. The descriptive headings of this Agreement are inserted for convenience only and do not constitute a substantive part of this Agreement. Whenever required by the context, any pronoun used in this Agreement shall include the corresponding masculine, feminine or neuter forms. Any singular term in this Agreement shall be deemed to include the plural, and any plural term the singular. Where a word or phrase is defined herein, each of its other grammatical forms shall have a corresponding meaning. The use of the word "include," "includes" and "including" in this Agreement shall be by way of example rather than by limitation. Reference to any agreement, document or instrument means such agreement, document or instrument as amended or otherwise modified from time to time in accordance with the terms thereof, and if applicable, hereof. The use of the words "or," "either" and "any" shall not be exclusive. The word "extent" in the phrase "to the extent" shall convey the concept of degree, and such phrase shall not mean simply "if." Except as otherwise indicated, all references in this Agreement to "Schedules," "Sections" and "Exhibits" are intended to refer to Schedules, Sections and Exhibits to this Agreement. The terms "hereof," "hereunder," "herein" and words of similar import will refer to this Agreement as a whole and not to any particular provision of this Agreement. The use of the word "threatened" in this Agreement shall be deemed followed by "in writing." All references to "dollars" or "\$" shall be to U.S. dollars. References to any provisions of Law shall be construed as references to such provisions as amended, expanded, consolidated or reissued, or as their applicability may be altered from time to time by other rules, and shall include any provision from which they originate (with or without modifications), regulations, instruments or other legal rules subordinate thereto. References to any period of days shall be deemed to be the relevant number of calendar days (unless Business Days are specified); provided that all references to terms or periods in this Agreement shall be counted excluding the date of the event that causes such term or period to begin and including the last day of the relevant term or period. All periods provided for in this Agreement ending on a day that is not a Business Day shall be automatically extended to the first subsequent Business Day. The phrases "provided," "delivered," "made available," or "furnished" (or any phrase of similar import) means that the information referred to has been physically or electronically delivered to the Investors or its representatives (including the Investors' Representatives and their representatives) at least one (1) Business Days prior to the date hereof (and available continuously until the execution of this Agreement), including information or documents that have been uploaded to the data room hosted by Donnelley Financial Solutions Venue in connection with the transactions contemplated hereby. This Agreement shall be construed as if drafted jointly by the Parties. The phrase "primarily related to" in respect of any program or business shall mean "primarily related to, primarily used or primarily held for use in" the conduct of such program or business. The phrase "ordinary course of business" means an action taken by or on behalf of a Person that is consistent (including with respect to frequency and magnitude) with the past practices of such Person and is taken in the ordinary course of the operations of such Person.

Schedules. The BridgeBio Disclosure Letter and the exhibits, schedules and annexes to this Agreement shall be Section 8.10 construed with, and as an integral part of, this Agreement. Each capitalized term used in any exhibit, schedule or annex or the BridgeBio Disclosure Letter but not otherwise defined therein shall be defined as set forth in this Agreement. The BridgeBio Disclosure Letter had been arranged in numbered and lettered sections and subsections corresponding to the applicable numbered and lettered sections and subsections contained in this Agreement. Each item disclosed in the BridgeBio Disclosure Letter shall constitute an exception to, or as applicable, disclosure for the purposes of, the representations and warranties (or covenants, as applicable) to which it makes reference and shall also be deemed to be constructively disclosed or set forth in any other section in the BridgeBio Disclosure Letter relating to other sections of this Agreement to the extent a cross-reference is expressly made to such other section in the BridgeBio Disclosure Letter or to the extent that the relevance of such item as an exception to, or as applicable, disclosure for the purposes of, another section of this Agreement is reasonably apparent from the face of such disclosure that such disclosure also qualifies or applies to, or is disclosed for the purposes of, such other section of this Agreement. The fact that any item of information is disclosed in the BridgeBio Disclosure Letter shall not be construed to mean that such information is required to be disclosed by this Agreement. Such information and the dollar thresholds set forth herein shall not be used as a basis for interpreting the terms "material" or other similar terms in this Agreement. The inclusion of any item in the BridgeBio Disclosure Letter shall not constitute an admission by BridgeBio that such item is or is not material. No disclosure in the BridgeBio Disclosure Letter relating to any possible breach or violation of any Contract, Law or Governmental Order shall be construed as an admission or indication that any such breach or violation exists or has actually occurred. The BridgeBio Disclosure Letter and the information contained therein are intended only to qualify or provide disclosure for the purposes of the applicable representations, warranties and covenants contained in this Agreement and shall not be deemed to expand in any way the scope or effect of any of such representations, warranties or covenants.

Section 8.11 Notices. All notices hereunder shall be in writing and shall be deemed to have been duly given or made (i) as of the date delivered if delivered personally, mailed by registered or certified mail (postage prepaid, return receipt requested) or sent by overnight courier (providing proof of delivery) or (ii) as of the date transmitted if sent by electronic transmission (with confirmation of transmission by the transmitting equipment) to the following electronic mail addresses, in each case, to the addresses below (or at such other contact information for a party hereto as shall be specified by like notice):

## If to BridgeBio:

BridgeBio Pharma, Inc. 3160 Porter Drive, Suite 250

Palo Alto, CA 94304

Attention: Legal Department Email: legal@bridgebio.com

with a copy (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP

One Manhattan West New York, NY 10001

Attention: Stephen F. Arcano

Thomas W. Greenberg

Email: stephen.arcano@skadden.com

thomas.greenberg@skadden.com

#### If to the Company:

Notices shall be sent to the address and attention of the Company (as set forth in the notice provisions of the LLC Agreement).

<u>If to Viking Global Opportunities Illiquid Investments Sub-Master LP or Viking Global Opportunities Drawdown (Aggregator)</u> <u>LP</u>:

c/o Viking Global Investors LP 600 Washington Blvd

Stamford, CT 06901

Attention: General Counsel

Email: legalnotices@vikingglobal.com

with a copy (which shall not constitute notice) to:

Gunderson Dettmer Stough Villeneuve Franklin & Hachigian LLP

1250 Broadway, 23rd Floor

New York, NY 10001

Attention: Greg Volkmar

Jeffrey L. Stricker

Email: gvolkmar@gunder.com

jstricker@gunder.com

# If to Patient Square Bravo Aggregator, LP:

c/o Patient Square Equity Advisors, LP 2884 Sand Hill Road, Suite 100

Menlo Park, CA 94025

Attention: Adam Fliss; Neel Varshney; Trit Garg

Email: adam@patientsquarecapital.com; neel@patientsquarecapital.com;

tgarg@patientsquarecapital.com

with a copy (which shall not constitute notice) to:

Ropes & Gray LLP Three Embarcadero Center San Francisco, CA 94111

Attention: Jason Freedman; Megan Baca

Email: Jason.Freedman@ropesgray.com; Megan.Baca@ropesgray.com

## If to SC US/E GROWTH FUND X MANAGEMENT, L.P. or SC US/E Venture Fund XVIII Management, L.P.:

2800 Sand Hill Road, Suite 101

Menlo Park CA 94025 Attention: Compliance

Email: compliance@sequoiacap.com

with a copy (which shall not constitute notice) to:

Email: sequoiacapital@sequoiacap.com

#### If to Frazier Life Sciences XI, L.P.:

1001 Page Mill Rd, Building 4, Suite B Palo Alto, CA 94304 Attention: Patrick Heron Email: Patrick@frazierls.com

with a copy (which shall not constitute notice) to:

Frazier Life Sciences XI, L.P. Attn: Chief Financial Officer Two Union Square 601 Union St., Suite 3200 Seattle, WA 98101

#### If to Frazier Life Sciences Public Fund, L.P.:

1001 Page Mill Rd, Building 4, Suite B Palo Alto, CA 94304 Attention: Albert Cha Email: albert@frazierls.com

with a copy (which shall not constitute notice) to:

Frazier Life Sciences Public Fund, L.P. Attn: Chief Financial Officer Two Union Square 601 Union St., Suite 3200 Seattle, WA 98101

# If to Frazier Life Sciences Public Overage Fund, L.P.:

1001 Page Mill Rd, Building 4, Suite B Palo Alto, CA 94304 Attention: Albert Cha Email: albert@frazierls.com

with a copy (which shall not constitute notice) to:

Frazier Life Sciences Public Overage Fund, L.P. Attn: Chief Financial Officer Two Union Square 601 Union St., Suite 3200 Seattle, WA 98101

# If to Cormorant Private Healthcare Fund IV, LP, Cormorant Private Healthcare Fund V, LP or Cormorant Global Healthcare Master Fund, LP:

200 Clarendon Street 52nd Floor Boston, MA 02116

Attention: Neb Obradovic

Email: neb@cormorant-asset.com

#### If to Aisling V Bridge Splitter LP:

c/o Aisling Capital V, L.P. 489 Fifth Avenue, 10th Floor New York, NY 10017

Attn: Drew Schiff & Eric Aguiar

Email: dschiff@aislingcapital.com; eaguiar@aislingcapital.com

and

c/o Aisling Capital V, L.P.
489 Fifth Avenue, 10th Floor
New York, NY 10017
Attn: Chief Financial Officer
Email: rwenzel@aislingcapital.com

with a copy (which shall not constitute notice) to:

McDermott Will & Emery LLP One Vanderbilt Avenue New York, NY 10017-3852

Attn: Todd Finger

Email: tfinger@mwe.com

#### If to Kumar Haldea Revocable Trust:

Notices shall be sent to the address and attention of Kumar Haldea Revocable Trust (as set forth in the notice provisions of the LLC Agreement).

# If to the Investors' Representatives:

Notices shall be sent to the address and attention of Patient Square Bravo Aggregator, LP, Viking Global Opportunities Illiquid Investments Sub-Master LP and Viking Global Opportunities Drawdown (Aggregator) LP (as set forth above).

Section 8.12 <u>Entire Agreement</u>. The Transaction Documents (including the BridgeBio Disclosure Letter) collectively constitute the complete agreement among BridgeBio, the Company the Investors and the Investors' Representatives, and supersede all prior agreements and understandings, both written and oral, between the parties hereto with respect to the transactions contemplated by this Agreement, unless otherwise specifically provided in a written instrument executed by the Parties expressly referencing this Agreement and this <u>Section 8.12</u>.

Section 8.13 <u>Public Announcements</u>. Except as otherwise provided herein, (a) the Parties shall consult in advance with each other before issuing any press release or otherwise making any public disclosure or public statements with respect to this Agreement or the transactions contemplated hereunder; and (b) no such press release, public disclosure or public statement shall be made unless mutually agreed upon by the Parties or required by Law or applicable stock exchange regulation; <u>provided</u> that each Investor (to the extent such Investor is a financial sponsor) is permitted to report and disclose, on a confidential basis, the status of this Agreement and the transactions contemplated hereby to its direct or indirect limited partners or current or prospective investors in connection with ordinary course fundraising, reporting or informational activities.

## Section 8.14 <u>Investors' Representatives</u>.

- (a) Each Investor irrevocably authorizes and appoints the Investors' Representatives as such Investor's representatives and attorneys-in-fact to act on behalf of such Investor with respect to this Agreement and any other Transaction Document (other than the Amended and Restated LLC Agreement) and to take any and all actions and make any decisions required or permitted to be taken by the Investors' Representatives pursuant to this Agreement or other Transaction Document (other than the Amended and Restated LLC Agreement), including the exercise of the power to:
  - (i) give and receive notices and communications;
  - (ii) execute and deliver all documents necessary or desirable to carry out the intent of this Agreement and the other Transaction Documents (other than the Amended and Restated LLC Agreement);
  - (iii) make all elections or decisions contemplated by this Agreement and the other Transaction Documents (other than the Amended and Restated LLC Agreement);
  - (iv) engage, employ, or appoint any agents or representatives (including attorneys, accountants, and consultants) to assist the Investors' Representatives in complying with their duties and obligations; and
  - (v) take all actions necessary or appropriate in the good faith judgment of the Investors' Representatives for the accomplishment of the foregoing.
- (b) BridgeBio and the Company shall be entitled to deal exclusively with Investors' Representatives on all matters relating to this Agreement and the other Transaction Documents (other than the Amended and Restated LLC Agreement) and shall be entitled to rely conclusively (without further evidence of any kind whatsoever) on any document executed or purported to be executed on behalf of any Investor by each of the Investors' Representatives, and on any other action taken or purported to be taken on behalf of any Investor by each of the Investors' Representatives, as being fully binding upon such Person. Notices or communications to or from the Investors' Representatives shall constitute notice to or from each of the Investors. Any decision or action by each of the Investors' Representatives hereunder shall constitute a decision or action of all the Investors and shall be final, binding and conclusive upon each such Person. No Investor shall have the right to object to, dissent from, protest or otherwise contest the same. The provisions of this Section 8.14, including the power of attorney granted hereby, are independent and severable, are irrevocable and coupled with an interest and shall not be terminated by any act of any one or more of the Investors, or by operation of Law, whether by death or other event. For the avoidance of doubt, any action taken by the Investors' Representatives in accordance with Section 8.14(a) hereof shall be taken jointly by each Investors' Representative and no Investors' Representative shall be entitled to act independent of the other Investors' Representative.

- (c) Each Investors' Representative may resign at any time; <u>provided</u>, <u>however</u>, in no event shall any Investors' Representative resign without the Investors (by action of the Investors representing a majority in interest of the Closing Date Cash Contribution (the "<u>Required Majority</u>")) having first appointed a new Investors' Representative who shall assume such duties immediately upon the resignation of such Investors' Representative. If, for any reason, either Investors' Representative is unable to serve as an Investors' Representative, a new Investors' Representative shall be appointed by the Investors (by action of the Required Majority). Notice of such appointment shall be sent to BridgeBio.
- (d) Each Investor (pro rata based on the portion of the Closing Date Cash Contribution) hereby severally, for itself only and not jointly and severally, agrees to indemnify and hold harmless each Investors' Representative against all fees, costs, expenses (including reasonable attorneys' fees), damages, liabilities, judgments, fines and amounts paid in settlement actually and reasonably incurred by such Investors' Representative by reason of the fact it is or was acting as an Investors' Representative pursuant to the terms of this Agreement.
- (e) Notwithstanding anything herein to the contrary, other than in the case of Fraud, the Investors' Representatives shall not have any obligation or liability to any Investor relating to the performance of their duties under this Agreement or under any Transaction Document. The Investors' Representatives shall not have by reason of this Agreement a fiduciary relationship in respect of any Investor. The Investors' Representatives shall not be liable to any Investor for any action taken or omitted by them or any agent employed by them hereunder or under any other document entered into in connection herewith, except that the Investors' Representatives shall not be relieved of any liability imposed by applicable Law for willful misconduct or actual common law fraud with the intent to deceive under Delaware law as determined by a final, non-appealable order of a court of competent jurisdiction.
- (f) For the avoidance of doubt, this <u>Section 8.14</u> shall not impact any of the rights, privileges, preferences or obligations of any Investor under the Amended and Restated LLC Agreement.

## Section 8.15 <u>Privileges; Conflicts.</u>

(a) Each of the Parties acknowledges that BridgeBio has retained Skadden, Arps, Slate, Meagher & Flom LLP ("Skadden") to act as counsel to BridgeBio in connection with the transactions contemplated hereby, that Skadden has not acted as counsel for any other Person in connection with the transactions contemplated hereby and that no other Person has the status of a client of Skadden for conflict of interest or any other purposes in connection with the transactions contemplated hereby and thereby. Each of the Parties further acknowledges that, after the Closing, Skadden may act as counsel to the Company or its Subsidiaries in connection with matters arising out of or related to this Agreement, the transactions contemplated hereby and the business activities of the Company and its Subsidiaries and that Skadden's prior representation of BridgeBio shall not be deemed to be a disabling conflict with respect to such representation. Each of the Parties hereby waives any conflict of interest resulting from the foregoing. The Parties further agree that, as to all communications, whether written or electronic, among Skadden, BridgeBio, the Special Committee, the Company or any of their respective Affiliates, and all of their files, attorney notes, drafts or other documents, that relate in any way to this Agreement or the transactions contemplated hereby that predate the Closing and that are protected by the attorney-client privilege, the expectation of client confidence or any other rights to any evidentiary privilege, such protections belong to BridgeBio and may be controlled by BridgeBio and shall not pass to or be claimed by the Company or any of its respective Affiliates or any of the Investors or the Investors' Representatives. The Parties shall take, and shall cause their respective Affiliates to take, all steps necessary to implement the intent of this Section 8.15(a). The Parties further agree that Skadden and their respective partners and employees are third-party beneficiaries of this Section 8.15(a).

Each of the Parties acknowledges that the Special Committee has retained Goodwin Procter LLP ("Goodwin") (b) and Richards, Layton & Finger P.A. ("Richards") to act as counsel to the Special Committee in connection with the transactions contemplated hereby, that neither Goodwin nor Richards has acted as counsel for any other Person in connection with the transactions contemplated hereby and that no other Person has the status of a client of Goodwin or Richards for conflict of interest or any other purposes in connection with the transactions contemplated hereby. Each of the Parties further acknowledges that, after the Closing, Goodwin and Richards may act as counsel to the Company or its Subsidiaries in connection with matters arising out of or related to this Agreement, the transactions contemplated hereby and the business activities of the Company and its Subsidiaries and that Goodwin's and Richards' prior representation of the Special Committee shall not be deemed to be a disabling conflict with respect to such representation. Each of the Parties hereby waives any conflict of interest resulting from the foregoing. The Parties further agree that, as to all communications, whether written or electronic, among Goodwin, Richards, the Special Committee, BridgeBio, the Company or any of their respective Affiliates, and all of their files, attorney notes, drafts or other documents, that relate in any way to this Agreement or the transactions contemplated hereby that predate the Closing and that are protected by the attorney-client privilege, the expectation of client confidence or any other rights to any evidentiary privilege, such protections belong to BridgeBio and may be controlled by BridgeBio and shall not pass to or be claimed by the Company or any of its respective Affiliates or any of the Investors. The Parties shall take, and shall cause their respective Affiliates to take, all steps necessary to implement the intent of this Section 8.15(b). The Parties further agree that Goodwin and Richards and their respective partners and employees are third-party beneficiaries of this Section 8.15(b).

Non-Recourse. All claims, obligations, liabilities, or causes of action (whether in contract or in tort, in law or Section 8.16 in equity, or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with, or relate in any manner to this Agreement, or the negotiation, execution, or performance of this Agreement (including any representation or warranty made in, in connection with, or as an inducement to, this Agreement), may be made only against (and such representations and warranties are those solely of) the Parties and then only with respect to the express obligations set forth herein and pursuant to the terms hereof. No Person who is not a Party, including any current, former or future director, officer, employee, incorporator, member, partner, manager, shareholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any Party, or any current, former or future director, officer, employee, incorporator, member, partner, manager, shareholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any of the foregoing (collectively, the "Nonparty Affiliates"), shall have any liability (whether in contract or in tort, in law or in equity, or granted by statute) for any claims, causes of action, obligations, or liabilities arising under, out of, in connection with, or related in any manner to this Agreement or based on, in respect of, or by reason of this Agreement or its negotiation, execution, performance, or breach, and, to the maximum extent permitted by Laws, each Party hereby waives and releases all such liabilities, claims, causes of action, and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by Law, each Party hereby waives and releases any and all rights, claims, demands, or causes of action that may otherwise be available at law or in equity, or granted by statute, to avoid or disregard the entity form of a Party or otherwise impose liability of a Party on any Nonparty Affiliate, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise. The provisions of this Section 8.16 are intended to be for the benefit of, and shall be enforceable by, each Nonparty Affiliate and each such Person's heirs, representatives, successors or assigns, it being expressly agreed that such Persons shall be third party beneficiaries of this Section 8.16. Nothing in this Section 8.16 shall limit claims against (or liability of) any Person in respect of Fraud.

[THE REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

The undersigned have entered into this Transaction Agreement effective as of the date first written above.

# BRIDGEBIO PHARMA, INC.

By: /s/ Neil Kumar

Name: Neil Kumar

Title: Chief Executive Officer

[Signature Page to Transaction Agreement]

## **INVESTORS AND INVESTORS' REPRESENTATIVES:**

# VIKING GLOBAL OPPORTUNITIES ILLIQUID INVESTMENTS SUB-MASTER LP

By: Viking Global Opportunities Portfolio GP LLC, its general partner

By: /s/ Katerina Novak

Name: Katerina Novak Title: Authorized Signatory

# VIKING GLOBAL OPPORTUNITIES DRAWDOWN (AGGREGATOR) LP

By: Viking Global Opportunities Drawdown Portfolio GP LLC, its general partner

By: /s/ Katerina Novak

Name: Katerina Novak Title: Authorized Signatory

[Signature Page to Transaction Agreement]

# PATIENT SQUARE BRAVO AGGREGATOR, LP

By: Patient Square Equity Advisors, LP

Its: General Partner

By: Patient Square Capital Holdings, LLC

Its: General Partner

By: /s/ Adam Fliss

Name: Adam Fliss Its: General Counsel

#### **INVESTORS' REPRESENTATIVE:**

# PATIENT SQUARE BRAVO AGGREGATOR, LP

By: Patient Square Equity Advisors, LP

Its: General Partner

By: Patient Square Capital Holdings, LLC

Its: General Partner

By:/s/ Adam Fliss

Name: Adam Fliss Its: General Counsel

# SC US/E GROWTH FUND X MANAGEMENT, L.P., as nominee

By: SC US (TTGP), LTD., a Cayman Islands exempted company

Title: General Partner

By: /s/ Roelof F. Botha

Name: Roelof F. Botha Title: Authorized Signatory

# SC US/E VENTURE FUND XVIII MANAGEMENT, L.P., as nominee

By: SC US (TTGP), LTD., a Cayman

Islands exempted company

Title: General Partner

By: /s/ Roelof F. Botha

Name: Roelof F. Botha Title: Authorized Signatory

### FRAZIER LIFE SCIENCES XI, L.P.

By: FHMLS XI, L.P. Its general partner

By: FHMLS XI, L.L.C. Its general partner

By: /s/ Patrick Heron
Name: Patrick Heron
Title: Managing Director

#### FRAZIER LIFE SCIENCES PUBLIC FUND, L.P.

By: FHMLSP, L.P. Its general partner

By: FHMLSP, L.L.C. Its general partner

By: /s/ Albert Cha

Name: Albert Cha

Title: Managing Director

#### FRAZIER LIFE SCIENCES PUBLIC OVERAGE FUND, L.P.

By: FHMLSP Overage, L.P.

Its general partner

By: FHMLSP Overage, L.L.C.

Its general partner

By: /s/ Albert Cha

Name: Albert Cha

Title: Managing Director

### **CORMORANT PRIVATE** HEALTHCARE FUND IV, LP

By: Cormorant Private Healthcare GP IV, LLC

By: /s/ Bihua Chen

Name: Bihua Chen Title: Managing Member

### **CORMORANT PRIVATE** HEALTHCARE FUND V, LP

By: Cormorant Private Healthcare GP V, LLC

By: /s/ Bihua Chen

Name: Bihua Chen Title: Managing Member

# CORMORANT GLOBAL HEALTHCARE MASTER FUND,

LP

By: Cormorant Global Healthcare GP,

LLC

By: /s/ Bihua Chen

Name: Bihua Chen Title: Managing Member

### AISLING V BRIDGE SPLITTER LP

By: Aisling Capital Partners V, LP

Its: General Partner

By: Aisling Capital Partners V LLC

Its: General Partner

By: /s/ Robert Wenzel

Name: Robert Wenzel

Title: Chief Financial Officer

# KUMAR HALDEA REVOCABLE TRUST

By:/s/Neil Kumar

Name: Neil Kumar Title: Trustee

# **BRIDGEBIOX PHARMA, LLC**

By: /s/ James Li

Name: James Li Title: President

### Exhibit 10.2

### **EXECUTION VERSION**

### AMENDED AND RESTATED

### LIMITED LIABILITY COMPANY AGREEMENT

OF

# BRIDGEBIOX PHARMA, LLC,

a Delaware Limited Liability Company

Dated as of August 16, 2024

		<u>Page</u>
ARTICLE	I DEFINITIONS	1
1.1	Definitions	1
1.2	Interpretive Provisions	18
ARTICI F	II THE LIMITED LIABILITY COMPANY	19
ARTICLL	II THE ENVITED EIABIEIT I CONTAINT	17
2.1	Formation	19
2.2	Name	19
2.3	Business Purpose	19
2.4	Registered Office and Agent	19
2.5	Term	19
2.6	Company Powers	19
2.7	Business Transactions of a Member or Director with the Company	19
2.8	Principal Place of Business	20
2.9	Title to Company Property	20
ARTICLE	III THE MEMBERS	20
		_ <u>_</u>
3.1	The Members	20
3.2	Member Meetings	20
3.3	Liability of Members	21
3.4	Power to Bind the Company	21
3.5	Pass-Through Rights and Obligations	21
3.6	Corresponding Management Holdco Units	22
3.7	Keyman Trigger Event	22
ARTICI F	IV THE BOARD AND OFFICERS	23
MICHELL	TV THE BOTTLD OFFICERS	23
4.1	Management by the Board of Directors	23
4.2	Meetings of the Board	24
4.3	Certain Board Approval Rights	25
4.4	Power to Bind Company	28
4.5	Delegation of Authority; Officers	28
4.6	Committees	30
4.7	Reliance by Third Parties	30
4.8	No Fiduciary Duties	30
4.9	Corporate Opportunities	31
4.10	Certain Member Approval Rights	33
4.11	Reversion Rights	35
4.12	Insurance	35
ARTICLE	V CAPITAL STRUCTURE AND CONTRIBUTIONS	35
5.1	Capital Structure	35
5.2	Defaulting Investor; Dilution.	38
5.3	Unit Schedule	39
5.4	Preemptive Rights	39
5.5	No Withdrawal of Capital Contributions	41
5.6	No Other Capital Contributions	41

i

		<u>Page</u>
5.7	Capital Accounts	41
ARTICLI	E VI ALLOCATIONS AND DISTRIBUTIONS	42
6.1	Allocations	42
6.2	Tax Allocations	44
6.3	No Right to Distributions	45
6.4	Distributions	45
6.5	Distributions to Holders of Class B Units	46
6.6	Tax Distributions	46
6.7	Withholding	47
6.8	Restrictions on Distributions	48
6.9	Determinations by the Board	48
ARTICLI	E VII ACCOUNTS	48
7.1	Books	48
7.1	Information Rights	48
7.3	Tax Matters	49
7.4	Fiscal Year	52
7.5	Initial Budget	52
ARTICLI	E VIII TRANSFER OF UNITS IN THE COMPANY	52
8.1	Prohibition	52
8.2	Conditions to Permitted Transfers	53
8.3	Effect of Transfers	54
8.4	Admission of Additional Members	54
8.5	Admission of Assignees as Substitute Members	55
8.6	Withdrawal of Members upon Transfer	55
8.7	Right of First Offer	55
8.8	Call Rights	57
8.9	Tag-Along Rights	58
8.10	Drag-Along Rights	60
8.11	Certain Terms and Conditions Applicable to Drag-Along Sales and Tag-Along Sales	62
8.12	Effect of Notices	64
8.13	•	64
8.14	Management Holdco Purchase Rights	64
ARTICLI	E IX QUALIFIED IPO; REGISTRATION RIGHTS	64
9.1	Qualified IPO Mechanics	64
9.2	Treatment of Units in Conversion Transaction	67
9.3	Registration Rights	67
ARTICLI	E X EVENTS OF DISSOLUTION	68
10.1	Dissolution	68
ARTICLI	E XI TERMINATION	69
44.4		
11.1	Liquidation	69

11.2	Final Accounting	69
11.3	Distribution in Kind	69
11.4	Certificate of Cancellation	69
ARTICLE 2	XII EXCULPATION AND INDEMNIFICATION	69
12.1	Exculpation	69
12.2	Indemnification	69
12.3	Effect of Modification	70
12.4	Nonexclusivity of Rights	70
ARTICLE 2	XIII AMENDMENT TO AGREEMENT	7
13.1	Amendments	7
ARTICLE 2	XIV GENERAL PROVISIONS	7
14.1	Expenses	7
14.2	Notices	7
14.3	Publicity; Confidentiality	72
14.4	Entire Agreement and Waiver	73
14.5	Conflict between This Agreement and Class B Unit Agreement	73
14.6	Counterparts	73
14.7	Severability	73
14.8	Governing Law; Venue; Waiver of Jury Trial	73
14.9	Binding Effect	74
14.10	Additional Documents and Acts	74
14.11	No Third-Party Beneficiary	75
14.12	Nonrecourse	75
14.13	Successors and Assigns	75
14.14	Specific Performance	75
14.15	Representations and Warranties	70
14.16	Survival of Representations and Warranties; Termination of Covenants	76

iii

# <u>Schedules</u>

Schedule A Membership Interests; Capital Contr	ibutions

Schedule B Total Funding Commitments

Schedule C Initial Budget

Schedule D Representations and Warranties

# **Exhibits**

Exhibit A Joinder Agreement Exhibit B Spousal Consent

iv

This AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT (this "<u>Agreement</u>") of BridgeBioX Pharma, LLC, a Delaware limited liability company (the "<u>Company</u>"), is made and entered into as of August 16, 2024, by and among each of the Persons set forth on <u>Schedule A</u>.

#### RECITALS

WHEREAS, the Company was formed by BridgeBio, as the sole member, pursuant to and in accordance with the Delaware Limited Liability Company Act (as amended from time to time, the "Act") by the initial filing of the Certificate of Formation on June 5, 2024, and entered into the original limited liability company agreement of the Company as of June 5, 2024 (the "Existing LLC Agreement");

WHEREAS, the Company has entered into a Transaction Agreement, dated as of August 16, 2024, pursuant to which BridgeBio Pharma, Inc., a Delaware corporation ("BBIO"), and the Investors have agreed to make certain contributions to the Company and to operate the Company as a joint venture (the "Transaction Agreement");

WHEREAS, pursuant to the Transaction Agreement, at the closing of the transactions contemplated by the Transaction Agreement (the "Closing") and concurrent with the execution of this Agreement, (a) BBIO will, or will cause its Subsidiaries to, make contributions of certain assets, liabilities and entities to the Company, (b) each Investor will make cash contributions to the Company and commit to make future cash contributions to the Company, and (c) the Company will (i) admit the Investors as Members, (ii) issue Class A Units to BridgeBio and each Investor as set forth opposite of such Person's name on Schedule A attached hereto and (iii) amend and restate the Existing LLC Agreement as provided herein; and

WHEREAS, BridgeBioX Management Aggregator, LLC, a Delaware limited liability company ("Management Holdco"), will hold Class B Units on behalf of certain individuals who are directors, officers, employees or service providers of the Company or the Company's Subsidiaries.

NOW, THEREFORE, in consideration of the mutual covenants and agreements contained herein and for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties, intending to be legally bound, hereby agree as follows:

#### ARTICLE I DEFINITIONS

1.1 <u>Definitions.</u> As used herein, the following terms have the meanings set forth below:

"Act" shall have the meaning set forth in the recitals hereto.

"Additional Cash Contributions" shall have the meaning set forth in Section 5.1(e)(i).

"Additional Member" shall mean a Person who has acquired newly issued Units directly from the Company after the date hereof and has been admitted as a Member of the Company pursuant to Section 8.4.

"Additional Securities" shall have the meaning set forth in Section 5.4(a).

"Adjusted Capital Account Deficit" shall mean, with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant Fiscal Year after giving effect to the following adjustments: (a) credit to such Capital Account any amounts that such Member is obligated to restore pursuant to Regulations Section 1.704-1(b)(2)(ii)(c) or the penultimate sentences of Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5); and (b) debit to such Capital Account such Member's share of the items described in Regulations Sections 1.704-1(b)(2)(ii)(d)(4), (5) and (6). This definition of Adjusted Capital Account Deficit is intended to comply with the provisions of Regulations Section 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.

"Affiliate" means, in respect of a Person, (a) any entity or person which directly or indirectly controls, is controlled by or is under common control with such Person, and (b) any corporation, trust, limited liability company, general or limited partnership or other entity advised or managed by, or under common control or management with, such Person; and for these purposes the term "control" shall mean (i) the right to exercise or cause the exercise of fifty percent (50%) or more of the voting rights of such entity or person, or (ii) the right to exercise management control by agreement. Without limiting the foregoing, an "Affiliate" of a Person that is a limited liability company, a limited partnership or a registered investment company shall include any fund or entity managed by the same manager or managing member or general partner or management company, investment adviser or by an entity controlling, controlled by, or under common control with such manager or managing member, general partner or management company or investment adviser. Notwithstanding the foregoing, (a) the Company or any of its Subsidiaries shall not be deemed or treated as an Affiliate of BBIO, BridgeBio or any Investor and (b) Management Holdco shall not be deemed or treated as an Affiliate of BBIO, BridgeBio or any Investor.

"Agreement" shall have the meaning set forth in the preamble hereto.

"Applicable Class B Unit Agreement" shall mean, with respect to any Class B Unit, an agreement between a member of Management Holdco and Management Holdco evidencing an award of the Management Holdco Class B Unit that corresponds to such Class B Unit.

"Applicable Holders" shall have the meaning set forth in Section 5.4(a).

"Applicable Persons" shall have the meaning set forth in Section 4.9(a).

"Article 8 Seller" shall have the meaning set forth in Section 8.11.

"Article 8 Transaction" shall have the meaning set forth in Section 8.11.

"Assignee" shall mean a transferee of Units who has not been admitted as a Substitute Member.

"BBIO" shall have the meaning set forth in the Recitals.

"BBIO Indications" shall mean, collectively, Transthyretin Amyloidosis (ATTR), Autosomal Dominant Hypocalcemia Type 1 (ADH1), Achondroplasia and other skeletal dysplasias, Limb-Girdle Muscular Dystrophy Type 2I (LGMD2I), Congenital Adrenal Hyperplasia (CAH) and gene therapy method for Tuberous Sclerosis Complex (TSC), and in each case, next generation applications of the foregoing.

"Board" shall have the meaning set forth in Section 4.1(a).

"Board Determined Cash Contributions" shall have the meaning set forth in Section 5.1(e)(i).

"Book Value" shall mean, with respect to any Company asset, the adjusted tax basis of the asset for United States federal income tax purposes, except that (a) the initial Book Value of any property contributed by a Member to the Company (other than cash) shall be the gross Fair Market Value of such property, (b) the Book Values of all Company assets shall be adjusted to equal their respective gross Fair Market Values, in accordance with the rules set forth in Section 1.704-1(b)(2)(iv)(f) of the Regulations, except as otherwise provided herein, immediately prior to: (i) the date of the acquisition of any additional Units by any new or existing Member in exchange for more than a de minimis Capital Contribution; (ii) the date of the Distribution of more than a de minimis amount of Company property (other than a pro rata Distribution) to a Member; (iii) the date of the actual liquidation of the Company within the meaning of Regulations Section 1.704-1(b)(2)(ii)(g); (iv) the date of the acquisition of any additional Units (other than a de minimis Unit) by any new or existing Member as consideration for the performance of services to or for the benefit of the Company; or (v) any other time at which revaluations of property are permitted to be made under Regulations Section 1.704-1(b)(2)(iv); provided that adjustments pursuant to clauses (i) through (v) above shall be made only if the Board determines in good faith that such adjustments are necessary or appropriate to reflect the relative economic interests of the Members, (c) the Book Value of property distributed to a Member shall be adjusted to equal the Fair Market Value of such property as of the date of such distribution, (d) the Book Value of all property shall be increased (or decreased) to reflect any adjustments to the adjusted basis of such property pursuant to Code Section 734(b) or Code Section 743(b) (including any such adjustments pursuant to Regulations Section 1.734-2(b)(1)), but only to the extent that such adjustments are taken into account in determining Capital Accounts pursuant to Regulations Section 1.704-1(b)(2)(iv)(m), clause (v) of the definition of Net Profits and Net Losses or Section 6.1(b)(viii); provided, however, that the Book Value of property shall not be adjusted pursuant to this clause (d) to the extent that the Board reasonably determines an adjustment pursuant to clause (b) of this definition is necessary or appropriate in connection with a transaction that would otherwise result in an adjustment pursuant to this clause (d) and (e) if the Book Value of property has been determined or adjusted pursuant to clauses (a), (b) or (d) of this definition, such Book Value shall thereafter be adjusted by the Depreciation taken into account with respect to such property for purposes of computing Net Profits and Net Losses and other items allocated pursuant to Article VI.

"BridgeBio" shall mean BridgeBio Pharma LLC, a Delaware limited liability company (and, as applicable, any of its Permitted Transferees that hold Class A Units).

"Business Day" shall mean any day that is not a Saturday or Sunday or other day on which banks in the States of New York or California are authorized or obligated to be closed.

"Call Notice" shall have the meaning set forth in Section 8.8(b).

"Call Notice Deadline" shall have the meaning set forth in Section 8.8(a).

"Call Price" shall have the meaning set forth in Section 8.8(a).

"Call Right" shall have the meaning set forth in Section 8.8(a).

"Called Units" shall have the meaning set forth in Section 8.8(a).

"Capital Account" shall have the meaning set forth in Section 5.7(a).

"Capital Contribution" shall mean any contribution or deemed contribution of cash or property to the Company made by or on behalf of a Member, as set forth from time to time in the books and records of the Company and as set forth on Schedule A hereto. Any reference in this Agreement to the Capital Contribution of a Member shall include the Capital Contribution of its predecessor(s) in interest.

"Cause" shall mean the definition of "cause" set forth in a Management Holder's employment agreement with the Company or any of the Company's Subsidiaries or other written services agreement with the Company or any of the Company's Subsidiaries, in each case that has been approved by the Board; provided that, if no such agreement which defines Cause is in effect at the time of determination, Cause shall mean: (a) a Person's gross negligence or willful misconduct in the performance of his or her duties to the Company or its Affiliates; (b) a Person's intentional nonperformance (other than due to illness, approved leave or vacation) or material failure to perform his or her duties, or refusal to abide by or comply with the lawful directives of the Board, which action(s) continue for, or have not been cured within, a period of fifteen (15) days after delivery to such Person of written notice of the need to cure or cease; (c) fraud, embezzlement, theft or the misappropriation of funds, money, assets or other property of the Company or any of its Subsidiaries, customers or suppliers by a Person; (d) a Person's conviction of, or pleas of guilty or nolo contendere to, a felony or other crime involving moral turpitude; (e) unauthorized use or disclosure by such Person of the Company's confidential information or trade secrets, which unauthorized use or disclosure causes material harm to the Company; (f) material breach by such Person of any agreement between such Person and the Company, which breach (i) continues beyond, or has not been cured within, the cure period specified in such agreement (or if no cure period is specified in such agreement, within fifteen (15) days after delivery to such Person of written notice of the need to cure or cease) and (ii) causes material harm to the Company; (g) failure by such Person to cooperate in good faith with a governmental or internal investigation of the Company or its directors, officers or employees, if the Company has requested such Person's cooperation, which failure to cooperate continues for, or has not been cured within, a period of fifteen (15) days after delivery to such Person of written notice of the need to cure; or (g) violation by a Person of any material written policy of the Company or its Affiliates (including any policy regarding sexual harassment or racial or other discrimination) that has been provided or made available to such Person; provided that, if such violation is capable of being cured without resulting in financial or reputational harm to the Company or any of its Affiliates, such Person shall have fifteen (15) days after delivery of written notice of such violation to so cure.

"CEO" shall have the meaning set forth in Section 4.5(b)(ii).

"Certificate of Cancellation" shall mean the certificate required to be filed with the Secretary of State of the State of Delaware pursuant to Section 18-203 of the Act in connection with a dissolution of the Company.

"Certificate of Formation" shall have the meaning set forth in Section 2.1.

"CFO" shall have the meaning set forth in Section 4.5(b)(iii).

"Change of Control" shall mean (a) any transaction or series of transactions immediately following which a Person or Persons acting as a group becomes the beneficial owner of more than fifty percent (50%) of the outstanding Voting Units of the Company or (b) a sale, lease, license or other disposition of all or substantially all of the Company's assets (including the capital stock or assets of the Company's Subsidiaries).

"Chosen Courts" shall have the meaning set forth in Section 14.8(b).

"Claims" shall have the meaning set forth in Section 12.2(a).

"Class A Holder" shall mean a Member owning Class A Units, in such Member's capacity as such.

"Class A Units" shall mean the Class A Units of the Company, having the powers, preferences, rights, qualifications, limitations and restrictions set forth in Sections 5.1(a) and 5.1(b).

"Class B Limit" shall have the meaning set forth in Section 5.1(c).

"Class B Percentage" shall mean, in respect of any Distributions, the product of (x) the Class B Limit multiplied by (y) the quotient of (i) the number of Class B Units that are issued and outstanding as of the relevant determination date and (ii) the total number of Class B Units authorized to be issued.

"Class B Units" shall mean the Class B Units of the Company, having the powers, preferences, rights, qualifications, limitations and restrictions set forth in Sections 5.1(a) and 5.1(c).

"Closing" shall have the meaning set forth in the preamble hereto.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Committed Equity Amount" shall have the meaning set forth in Section 5.1(c).

"Company" shall have the meaning set forth in the preamble hereto.

"Company Indications" shall mean, collectively, Erythropoietic Protoporphyria ("EPP"), Alpha-1 antitrypsin deficiency ("A1AT") and Tuberous Sclerosis Complex (with respect to small molecule method only for Tuberous Sclerosis Complex), and in each case, next generation applications of the foregoing.

"Company Industry Segment" shall have the meaning set forth in Section 4.9(b).

"Company Minimum Gain" shall mean "partnership minimum gain" as set forth in Regulations Section 1.704-2(b)(2).

"Company Offeror" shall mean (a) the Company (including any entity formed in a Conversion Transaction pursuant to Section 9.1), (b) any successor to the Company or any surviving entity resulting from a merger, consolidation or other business combination involving the Company or any wholly owned Subsidiary of the Company, (c) any wholly owned Subsidiary of the Company that is a holding company for all or substantially all of the assets of the Company and its Subsidiaries or (d) any other entity the securities of which are exchanged for Units pursuant to Article IX in anticipation of a Qualified IPO.

"Corporate Opportunity" shall have the meaning set forth in Section 4.9(a).

"Corresponding Direct Management Units" shall have the meaning set forth in Section 8.14.

"Corresponding Management Holdco Unit" shall have the meaning set forth in Section 3.6.

"Covered Persons" means, as applicable, as of any time of determination, (a) a then-current or former Member or Director, (b) an Affiliate of a then-current or former Member or Director, or (c) any then-current or former officer, manager, director, stockholder, partner, member, representative, agent or employee of a then-current or former Member or any of their respective Affiliates, and then-current or former officers, managers, directors, stockholders, partners, members, representatives, agents or employees of such Persons; provided that neither the Company nor its Subsidiaries shall be a Covered Person.

"Cure Period" shall have the meaning set forth in Section 5.2(a).

"<u>Defaulted Cash Contribution Amount</u>" shall have the meaning set forth in <u>Section 5.2(a)</u>.

"Defaulting Member" shall have meaning set forth in Section 5.2(a).

"Demand Registration" shall have the meaning set forth in Section 9.3(a).

"Depreciation" means, for each Fiscal Year or other period, an amount equal to the depreciation, amortization or other cost recovery deduction allowable for U.S. federal income tax purposes with respect to property for such Fiscal Year, except that (a) with respect to any such property the Book Value of which differs from its adjusted tax basis for U.S. federal income tax purposes and for which such difference is being eliminated by use of the "remedial method" pursuant to Regulations Section 1.704-3(d), Depreciation for such Fiscal Year shall be the amount of book basis recovered for such Fiscal Year under the rules prescribed by Regulations Section 1.704-3(d)(2), and (b) with respect to any other such property the Book Value of which differs from its adjusted tax basis at the beginning of such Fiscal Year, Depreciation shall be an amount which bears the same ratio to such beginning Book Value as the U.S. federal income tax depreciation, amortization, or other cost recovery deduction for such Fiscal Year bears to such beginning adjusted tax basis; provided that if the adjusted tax basis of any property at the beginning of such Fiscal Year is zero dollars (\$0.00), Depreciation with respect to such property shall be determined with reference to such beginning Book Value using any reasonable method selected by the Board in good faith.

"Director" shall have the meaning set forth in Section 4.1(a).

"<u>Disclosure Law</u>" shall mean, as to any Person, any law, statute, treaty, rule or regulation or binding determination of any Governmental Authority or stock exchange, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject or pertaining to any or all of the transactions contemplated or referred to herein.

"<u>Distribution</u>" shall mean a transfer of cash, securities or other property by the Company to a Member on account of Units as described in <u>Article VI</u> (including Tax Distributions).

"<u>Distribution Hurdle</u>" shall mean, with respect to each Class B Unit, an amount determined by the Board in good faith and equal to the amount specified as such in the Applicable Class B Unit Agreement for the Management Holdco Class B Unit corresponding to such Class B Unit as may be adjusted by the Board in good faith to account for Capital Contributions, distributions in respect of Units, liquidation of the Company, or other similar events; <u>provided</u> that the Distribution Hurdle with respect to each Class B Unit shall be no less than the amount required to constitute a Profits Interest at the time of issuance.

"Drag-Along Notice" shall have the meaning set forth in Section 8.10(b).

"Drag-Along Portion" shall mean, with respect to any Dragged Member, the product of (a) a fraction, the numerator of which is the number of Units proposed to be sold by the Drag-Along Sellers in a Drag-Along Sale and the denominator of which is the total number of Units then owned by the Drag-Along Sellers and (b) the total number of Units (including Class B Units) that are vested or that will vest in connection with the Drag-Along Sale based upon the price per unit to be received in such Drag-Along Sale held by such Dragged Member.

"<u>Drag-Along Purchaser(s)</u>" shall have the meaning set forth in <u>Section 8.10(a)</u>.

"Drag-Along Sale" shall have the meaning set forth in Section 8.10(a).

"Drag-Along Sellers" shall have the meaning set forth in Section 8.10(a).

"Drag-Along Units" shall have the meaning set forth in Section 8.10(b).

"Dragged Member" shall have the meaning set forth in Section 8.10(a).

"ERISA" shall mean the Employee Retirement Income Security Act of 1974.

"Event of Dissolution" shall have the meaning set forth in Section 10.1.

"Exchange Act" shall mean the Securities Exchange Act of 1934.

"Excluded Issuance" shall mean, as applicable, an issuance of equity or debt securities or any securities exercisable for, or convertible or exchangeable into equity or debt securities of the Company or any of its Subsidiaries (a) solely in the case of equity securities, to employees, directors, consultants and/or advisors of the Company or its Subsidiaries who are natural persons directly or pursuant to employee incentive plans, similar benefits programs or other arrangements, including grants to new hires, in connection with promotions or otherwise, in accordance with this Agreement, including the issuance by the Company of any Class B Units or other profits interests or other compensatory grants of Company securities, directly or indirectly through the issuance of Corresponding Management Holdco Units and a related issuance to Management Holdco; (b) issued as acquisition consideration in connection with an acquisition of a business or entity or shares or assets of another Person that has been approved by the Board; (c) upon the exercise, conversion or exchange of securities previously issued in compliance with Section 5.4; (d) in connection with a dividend, split, combination or subdivision of units made on a proportionate basis to all holders of Class A Units; (e) in connection with any recapitalization, reorganization, exchange or reclassification or similar transaction of the Company that has been approved by the Board in which all Class A Units are treated on a proportionate basis; (f) pursuant to, or in connection with, a Drag-Along Sale; (g) pursuant to, or in connection with, a Qualified IPO; (h) that is an issuance by a wholly owned Subsidiary of the Company or another wholly owned Subsidiary of the Company; and (i) in connection with Additional Cash Contributions pursuant to Section 5.1(e)(i).

"Executive Chair" shall have the meaning set forth in Section 4.5(b)(i).

"Existing LLC Agreement" shall have the meaning set forth in the recitals hereto.

"Fair Market Value" means (i) with respect to any asset as of any given date of determination, the purchase price that a willing buyer having all relevant knowledge would pay a willing seller for such asset in an "arm's-length" transaction (without discount for lack of control, minority interests, illiquidity or restrictions on transfer) and (ii) with respect any particular Unit as of any given date of determination, the amount that would be received on account of such Unit if all of the assets of the Company were sold for "Fair Market Value," as determined in accordance with clause (i) of this definition, and all amounts paid or received upon such sale were distributed to or among the Members in accordance Section 6.4 of this Agreement (for the avoidance of doubt, taking into account any Distribution Hurdle with respect to the Class B Units), in each case, as determined in good faith by the Board.

"Fiscal Year" shall have the meaning set forth in Section 7.4.

"Fully-Exercising Applicable Holder" shall have the meaning set forth in Section 5.4(a).

"Fund Investor" shall have the meaning set forth in Section 14.12.

"GAAP" shall mean United States generally accepted accounting principles, consistently applied.

"Governmental Authority" shall mean any federal, state, local, municipal, county or other governmental, quasi-governmental, administrative or regulatory authority, body, agency, court, tribunal, commission or other similar entity and any self-regulatory organization (including in each case any branch, department or official thereof).

"Indemnified Person" shall have the meaning set forth in Section 12.2(a).

"Indirect Management Member" shall have the meaning set forth in Section 8.14.

"Indirect Management Purchase Units" shall have the meaning set forth in Section 8.14.

"Indirect Management Repurchase" shall have the meaning set forth in Section 8.14.

"In-Scope Programs" shall have the meaning set forth in the Transaction Agreement.

"Information Waiver" shall have the meaning set forth in Section 4.9(b)(ii).

"Initial Budget" shall have the meaning set forth in Section 7.5.

"Initial Subscribing Holder" shall have the meaning set forth in Section 5.4(e).

"Investor" shall mean each of (a)(i) Viking Global Opportunities Illiquid Investments Sub-Master LP and (ii) Viking Global Opportunities Drawdown (Aggregator) LP (together with their Affiliates and Permitted Transferees, "Viking"), (b) Patient Square Bravo Aggregator, LP (together with its Affiliates and Permitted Transferees, "Patient Square"), (c)(i) SC US/E GROWTH FUND X MANAGEMENT, L.P. and (ii) SC US/E Venture Fund XVIII Management, L.P. (together with their Affiliates and Permitted Transferees, "Sequoia"), (d)(i) Frazier Life Sciences XI, L.P., (ii) Frazier Life Sciences Public Fund, L.P. and (iii) Frazier Life Sciences Public Overage Fund, L.P. (together with their Affiliates and Permitted Transferees, "Frazier"), (e)(i) Cormorant Private Healthcare Fund IV, LP, (ii) Cormorant Private Healthcare Fund V, LP and (iii) Cormorant Global Healthcare Master Fund, LP (together with their Affiliates and Permitted Transferees, "Cormorant"), (f) Aisling V Bridge Splitter LP (together with its Affiliates and Permitted Transferees, "NK").

"Investor Member Sell-Down Event" shall mean, with respect to the applicable Investor, a Transfer of more than fifty percent (50%) of the Class A Units acquired by such Investor (inclusive of Class A Units acquired after the Closing in exchange for any Additional Cash Contributions), other than to a Permitted Transferee.

"IRS" means the Internal Revenue Service.

- "Joinder Agreement" shall have the meaning set forth in Section 8.1(c).
- "Keyman Trigger Event" shall have the meaning set forth in Section 3.7(a).
- "Law" shall mean any law, statute, ordinance, rule, regulation, code, judgment, decree, order, award, decision, injunction, settlement, process, ruling, subpoena, verdict or governmental authorization (in each applicable case, whether temporary, preliminary or permanent) enacted, issued, promulgated, enforced or entered into by a Governmental Authority.
- "Lien" shall mean any lien, mortgage, lease, easement, servitude, levy, right of way, charge, pledge, security interest, covenant, condition, restriction or other encumbrance of any kind, or any conditional sale contract, title retention contract or other contract or agreement creating any of the foregoing.
- "<u>Liquidity Event</u>" shall mean (a) an Event of Dissolution, (b) a Change of Control or (c) such other extraordinary transaction the Board determines, in its discretion, to be a "Liquidity Event"; it being understood that in no event shall an initial public offering constitute a "Liquidity Event" for any purposes of this Agreement.
  - "Lock-Up Period" shall have the meaning set forth in Section 8.1(a).
  - "Losses" shall have the meaning set forth in Section 12.2(a).
- "Majority Interest" shall mean, as of any given date, an aggregate Voting Unit Percentage equal to more than fifty percent (50%) on such date.
  - "Management Holdco" shall have the meaning set forth in the recitals hereto.
- "Management Holdco Class B Holder" shall mean any member of Management Holdco owning Management Holdco Class B Units, in such member's capacity as such.
- "Management Holdco Class B Unit" shall mean the Class B Units of Management Holdco, having the powers, preferences, rights, qualifications, limitations and restrictions set forth in the Management Holdco Operating Agreement.
- "Management Holdco Operating Agreement" shall mean the limited liability company agreement of Management Holdco, as the same may be amended, restated or otherwise modified from time to time in accordance with the terms thereof.
- "Management Holder" shall mean any (a) director, officer, employee or service provider of the Company or any of its Subsidiaries for such time as such director, officer, employee or service provider is a holder of Management Holdco Class B Units or Corresponding Management Holdco Units or (b) other Person that holds Corresponding Management Holdco Units.
  - "Management Repurchase Closing" shall have the meaning set forth in Section 8.8(c).

"Mandatory Cash Contributions" shall have the meaning set forth in Section 5.1(e)(i).

"Member" shall mean each Person listed as a member on Schedule A hereto as of the date hereof, and each other Person that is an Additional Member or Substitute Member, in each case, and in the case of a Substitute Member, so long as constituting one hereunder.

"Member Minimum Gain" shall mean "partner nonrecourse debt minimum gain," as determined under Regulations Section 1.704-2(i)(3).

"Member Nonrecourse Debt" shall mean "partner nonrecourse debt," as set forth in Regulations Section 1.704-2(b)(4).

"Member Nonrecourse Deductions" shall mean "partner nonrecourse deductions" as set forth in Regulations Section 1.704-2(i). The amount of Member Nonrecourse Deductions shall be determined as set forth in Regulations Section 1.704-2(i).

"Member Related Parties" shall have the meaning set forth in Section 14.12.

"Member's Quarterly Estimated Tax Amount" shall mean for any fiscal quarter, the excess (if any) of (x) the product of (I) ¼ in the case of the first fiscal quarter of the Fiscal Year, ½ in the case of the second fiscal quarter of the Fiscal Year, ¾ in the case of the third fiscal quarter of the Fiscal Year and 1 in the case of the fourth fiscal quarter of the Fiscal Year and (II) the Tax Distribution Amount for such Fiscal Year over (y) all prior distributions paid to such Member pursuant to Section 6.4 or Section 6.6 with respect to all prior fiscal quarters of such Fiscal Year.

"Minimum Return Condition" shall have the meaning set forth in Section 8.10(a).

"Net Profits" and "Net Loss" shall mean, for each Fiscal Year or other period, an amount equal to the Company's taxable income or loss for such year or period, determined in accordance with Code Section 703(a) (for this purpose, all items of income, gain, loss or deduction required to be stated separately pursuant to Code Section 703(a)(1) shall be included in taxable income or loss), with the following adjustments:

- (i) any income of the Company that is exempt from federal income tax not otherwise taken into account in computing Net Profits or Net Loss shall be added to such taxable income or loss;
- (ii) any expenditures of the Company described in Code Section 705(a)(2)(B) or treated as Code Section 705(a)(2)(B) expenditures pursuant to Regulations Section 1.704-1(b)(2)(iv)(i) and not otherwise taken into account in computing Net Profits or Net Loss shall be subtracted from such taxable income or loss;
- (iii) income, gain or loss resulting from any disposition of Company property shall be computed by reference to the Book Value of the property disposed of, notwithstanding that the adjusted tax basis of such property differs from its Book Value:

- (iv) in the event the Book Value of any Company asset is adjusted pursuant to clause (b) or (c) of the definition of Book Value above, the amount of such adjustment shall be taken into account as gain or loss from the disposition of such asset for purposes of computing Net Profits and Net Losses;
- (v) to the extent an adjustment to the adjusted tax basis of any asset pursuant to Code Section 734(b) or Code Section 743(b) is required pursuant to Regulations Section 1.704-1(b)(2)(iv)(m)(4) to be taken into account in determining Capital Accounts as a result of a Distribution (other than in liquidation of a Member's interest in the Company), the amount of such adjustment shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases the basis of the asset) from the disposition of the asset and shall be taken into account for purposes of computing Net Profits or Net Losses;
- (vi) in lieu of the depreciation, amortization, and other cost recovery deductions taken into account in computing such taxable income or loss, there shall be taken into account Depreciation for such Fiscal Year; and
- (vii) any items that are allocated pursuant to <u>Section 6.1(b)</u> shall not be taken into account in computing Net Profits and Net Losses.

"Non-Exercised Securities" shall have the meaning set forth in Section 5.4(a).

"Nonrecourse Debt" shall mean a "nonrecourse liability" as set forth in Regulations Section 1.704-2(b)(3).

"Nonrecourse Deductions" shall mean "nonrecourse deductions," as set forth in Regulations Sections 1.704-2(b) and 1.704-2(c). The amount of Nonrecourse Deductions shall be determined as set forth in Regulations Sections 1.704-2(b) and 1.704-2(c).

"Non-Transferring Members" shall have meaning set forth in Section 8.7(a).

"Observer" shall have the meaning set forth in Section 4.1(g).

"Offer" shall have the meaning set forth in Section 5.4(a).

"Offered Tag Units" shall have the meaning set forth in Section 8.9(a).

"Offered Units" shall have the meaning set forth in Section 8.7(a).

"Other Preemptive Rights Holder" shall have the meaning set forth in Section 5.4(e).

"Participating Class B Unit" shall have the meaning set forth in Section 6.5(a).

"Participation Notice" shall have the meaning set forth in Section 5.4(a).

"Percentage Interest" shall mean, as of any date of determination with respect to each Member, the percentage determined by dividing the number of Units held by such Member as of such date by the aggregate number of Units held by all Members as of such date (with Class A Units and Class B Units treated equally for this purpose); provided, however, that no Class B Unit shall be included in calculating "Percentage Interest" (either in the numerator or the denominator) until such time as it becomes a Participating Class B Unit.

"Permitted Transfer" shall have the meaning set forth in Section 8.1(e).

"<u>Permitted Transferee</u>" shall mean (a) in the case of a Member who is a natural Person, such Member's spouse or issue, including adopted children, or a trust for the exclusive benefit of such Member's spouse or issue and (b) in the case of a Member which is not a natural Person, such Member's Affiliates; <u>provided, however</u>, that such Affiliates shall continue at all times to be Affiliates of the Member that makes such Transfer; <u>provided further</u> that (x) the Permitted Transferee of a Member who became a Member by virtue of being a Permitted Transferee must be a Permitted Transferee of the individual or entity who originally made the transfer of such interests in the first instance and (y) any direct or indirect Transfer of interests (including equity or beneficiary interests) of a Permitted Transferee which became a Member must not result in such Permitted Transferee ceasing to be a Permitted Transferee of the individual or entity who originally made the transfer of such interests in the first instance.

"Person" shall mean any individual, firm, corporation, partnership, limited liability company, trust or other entity.

"<u>Preemptive Portion</u>" shall mean an amount equal to (x) the number of Additional Securities proposed to be issued *multiplied by* (y) a fraction, the numerator of which is the number of Class A Units held by the Applicable Holder and their Permitted Transferees and the denominator of which is the number of then issued and outstanding Class A Units.

"Preemptive Right" shall have the meaning set forth in Section 5.4(a).

"Preemptive Right Period" shall have the meaning set forth in Section 5.4(a).

"Preemptive Securities" shall have the meaning set forth in Section 5.4(e).

"Price Per Unit" shall have the meaning set forth in Section 5.1(e)(i).

"Prime Rate" shall mean a fluctuating interest rate *per annum* equal at all times to the interest rate published from time to time in *The Wall Street Journal* and designated therein as the prime rate.

"Priority Return" shall mean eight percent (8%) per annum on the Unreturned Capital of the Class A Holders; provided that the Priority Return on any particular amount of such Unreturned Capital shall be calculated commencing from the applicable contribution date of such amount of Unreturned Capital.

"Pro Rata Share" shall have the meaning set forth in Section 8.11(a)(i).

"Profits Interest" shall mean an interest in the future profits of the Company satisfying the requirements for a partnership profits interest transferred in connection with the performance of services, as set forth in IRS Revenue Procedures 93-27 and 2001-43, or any future IRS guidance or other authority that supplements or supersedes the foregoing Revenue Procedures.

"Qualified IPO" shall mean an initial bona fide underwritten public offering and sale of Units (or other equity securities of the Company or any Company Offeror) pursuant to an effective registration statement (other than on Form S-4, S-8 or comparable form) filed under the Securities Act of 1933, as amended, pursuant to which securities are to be listed on a national securities exchange or other public listing on an accredited exchange, whether by way of direct listing, merger with a special purpose acquisition vehicle or similar transaction.

"Registration" shall have the meaning set forth in Section 9.3(b).

"Regulations" shall mean the U.S. Treasury Regulations.

"Related Party" shall mean any Member (or an Affiliate of a Member) or any Person in which any Member (or an Affiliate of a Member) holds an interest or any employee, officer, director, general partner, managing director, manager, principal or controlling equityholder of the foregoing.

"Related Party Transaction" shall mean any transaction between the Company or one of its Subsidiaries, on the one hand, and any Related Party, on the other hand (for the avoidance of doubt, excluding any transaction expressly provided for pursuant to the Transaction Agreement).

"Representatives" shall have the meaning set forth in Section 14.3.

"Reversion Rights" shall have the meaning set forth in Section 4.11.

"ROFO Acceptance Notice" shall have the meaning set forth in Section 8.7(c).

"ROFO Acceptance Period" shall have the meaning set forth in Section 8.7(c).

"ROFO End Period" shall have the meaning set forth in Section 8.7(e).

"ROFO Election Period" shall have the meaning set forth in Section 8.7(b).

"ROFO Offer" shall have the meaning set forth in Section 8.7(b).

"ROFO Portion" shall have the meaning set forth in Section 8.7(b).

"ROFO Sale Notice" shall have the meaning set forth in Section 8.7(a).

"ROFO Terms" shall have the meaning set forth in Section 8.7(a).

"Sale Notice" shall have the meaning set forth in Section 8.9(b).

"Sanctioned Party" shall mean any Person: (a) organized under the laws of, ordinarily resident in, or located in a country or territory that is the subject of comprehensive Sanctions (which as of the date of this Agreement include Cuba, Iran, North Korea, Syria, and the Crimea, Donetsk, and Luhansk regions of Ukraine ("Restricted Countries")); (b) 50% or more owned or controlled by the government of a Restricted Country; or (c) (i) designated on a sanctioned parties list administered by the United States, including the U.S. Department of the Treasury's Office of Foreign Assets Control's Specially Designated Nationals and Blocked Persons List, Foreign Sanctions Evaders List, Sectoral Sanctions Identification List (collectively, "Designated Parties"); or (ii) 50% or more owned or, where relevant under applicable Sanctions, controlled, individually or in the aggregate, by one or more Designated Party, in each case only to the extent that dealings with such Person is are prohibited pursuant to applicable Sanctions.

"Sanctions" shall mean economic sanctions administered by the Office of Foreign Assets Control of the United States Department of the Treasury, His Majesty's Treasury, the European Union and the Bureau of Industry Security of the U.S. Department of Commerce, and any sanctions measures under the U.S. International Emergency Economic Powers Act, the U.S. Trading with the Enemy Act, the U.S. Iran Sanctions Act, the U.S. Comprehensive Iran Sanctions, Accountability and Divestment Act of 2010, and the U.S. Iran Threat Reduction and Syria Human Rights Act of 2012, the U.S. National Defense Authorization Act of 2013 and the U.S. National Defense Authorization Act of 2013, all as amended, and any executive order, directive or regulation pursuant to the authority of any of the foregoing.

"SEC" shall mean the U.S. Securities and Exchange Commission.

"Securities Act" shall mean the Securities Act of 1933.

"Spousal Consent" shall have the meaning set forth in Section 8.1(c).

"Subsidiary" shall mean, with respect to any Person, any corporation, limited liability company, partnership, joint venture, association or other entity of which (a) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof or (b) if a limited liability company, partnership, joint venture, association or other entity (other than a corporation), a majority of limited liability company, partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons shall be deemed to have a majority ownership interest in a limited liability company, partnership, joint venture, association or other entity (other than a corporation) if such Person or Persons shall be allocated a majority of limited liability company, partnership, joint venture, association or other entity gains or losses or shall be or control any managing member, general partner or analogous controlling Person of such limited liability company, partnership, joint venture, association or other entity. For purposes hereof, references to a "Subsidiary" of any Person shall be given effect only at such times that such Person has one or more Subsidiaries and, unless otherwise indicated, the term "Subsidiary" refers to a Subsidiary of the Company.

"Substitute Member" shall mean an Assignee who has been admitted to all of the rights of membership pursuant to Section 8.3(b) or Section 8.5.

- "Suspension Period" shall have the meaning set forth in Section 3.7(b).
- "Tag-Along Election Notice" shall have the meaning set forth in Section 8.9(b).
- "Tag-Along Period" shall have the meaning set forth in Section 8.9(b).
- "<u>Tag-Along Portion</u>" shall have the meaning set forth in <u>Section 8.9(a)</u>.
- "Tag-Along Purchaser(s)" shall have the meaning set forth in Section 8.9(a).
- "Tag-Along Right" shall have the meaning set forth in Section 8.9(a).
- "<u>Tag-Along Rightholder</u>" shall have the meaning set forth in <u>Section 8.9(a)</u>.
- "Tag-Along Sale" shall have the meaning set forth in Section 8.9(a).
- "Tag-Along Seller" shall have the meaning set forth in Section 8.9(a).
- "Tagging Member" shall have the meaning set forth in Section 8.9(b).
- "Tax Audit" shall have the meaning set forth in Section 7.3(e).
- "Tax Distribution Amount" shall mean, for a given Member for a given fiscal quarter, the product of (a) the excess of (1) the net taxable income or gain of the Company that is expected to be allocated to a Member (taking into account any items of taxable income or gain, and associated allocations under Section 704(c) of the Code, and any tax basis adjustments applicable to such Member pursuant to Section 754 of the Code) during the entire Fiscal Year that includes the applicable fiscal quarter, as estimated by the Board in its sole discretion, over (2) the cumulative net taxable losses (taking into account any items of taxable loss or deduction, and associated allocations under Section 704(c) of the Code, and any tax basis adjustments applicable to such Member pursuant to Section 754 of the Code) allocated to such Member for prior Fiscal Years and not previously applied to reduce such Member's Tax Distribution Amount and available to offset such taxable income (taking into account any applicable limitations on the deductibility of capital losses and state and local taxes), *multiplied by* (b) the Tax Rate applicable to the Fiscal Year that includes such fiscal quarter.
  - "<u>Tax Distributions</u>" shall have the meaning set forth in <u>Section 6.6</u>.
- "<u>Tax Rate</u>" shall mean the sum of the highest applicable U.S. federal, state and local income tax rate applicable to the taxable income of an individual or corporation (whichever is higher) resident in Palo Alto, California, taking into account the character of such income, gain or loss (*e.g.*, capital gains or losses, dividends, ordinary income, etc.) and, if applicable, the deductibility of state and local income taxes for U.S. federal income tax purposes during each applicable Fiscal Year.
- "<u>Tax Return</u>" shall mean any return, election, declaration, report, schedule, document or statement, including any amendments or attachments thereof, which are submitted to, or required to be submitted to, any Governmental Authority with respect to taxes.

"<u>Termination</u>" shall mean, with respect to any Management Holder, such Management Holder's termination of employment or service (including as a director or officer) with the Company or any of its Subsidiaries for any reason or no reason.

"Total Funding Commitments" shall have the meaning set forth in Section 5.1(e)(i).

"TSA" means the Transition Services Agreement by and between BridgeBio Services, Inc., a Delaware corporation, on the one hand, and the Company, on the other hand, dated as of the date hereof.

"Transaction Agreement" shall have the meaning set forth in the recitals hereto.

"Transfer" shall mean any direct or indirect (including by way of transfers of equity interests in a Member (if such Member is an entity)), synthetic, voluntary or involuntary sale, donation, contribution, assignment, transfer, grant of participation in or reference under a derivatives contract or any other arrangement, pledge, mortgage, encumbrance or other disposition, whether in whole or in part, by operation of law or otherwise, except, with respect to any Investor, a sale, transfer, pledge, assignment, creation of an encumbrance or other disposition in a secondary transaction by a limited partner of an investment fund or other vehicle that is an Affiliate of the Investor shall not be deemed a "Transfer" hereunder.

"Transferring Member" shall have the meaning set forth in Section 8.7(a).

"Undersubscription Notice" shall have the meaning set forth in Section 5.4(a).

"Unit Consideration" shall have the meaning set forth in Section 5.1(e).

"<u>Units</u>" shall have the meaning set forth in <u>Section 5.1(a)</u>.

"Unrelated Party Transfer" shall have the meaning set forth in Section 8.7(a).

"<u>Unreturned Capital</u>" shall mean, as to any holder of Class A Units, the Capital Contributions made in respect of such Class A Units, *decreased* by the applicable aggregate amounts distributed to such holder in respect of such Class A Units pursuant to Section 6.4(a), as applicable.

"<u>Unvested Class B Unit</u>" shall mean, on any date of determination, any Class B Unit held by a Member that is not a Vested Class B Unit.

"<u>Vested Class B Unit</u>" shall mean, on any date of determination, any Class B Unit that corresponds to a Management Holdco Class B Unit that is "vested" in accordance with the Applicable Class B Unit Agreements applicable to such Management Holdco Class B Unit.

"<u>Voting Unit Percentage</u>" shall mean, with respect to any Member holding Voting Units as of a specified date, the percentage determined by dividing (a) the aggregate number of Voting Units held by such Member as of such date, by (b) the aggregate number of issued and outstanding Voting Units as of such date.

"Voting Units" shall mean the Class A Units, together with any other class of Units issued after the date hereof that the Board designates as Voting Units; provided that Class B Units shall not be entitled to any voting rights.

"Warrantor" shall have the meaning set forth in Section 14.15.

#### 1.2 <u>Interpretive Provisions. Unless the express context otherwise requires:</u>

- (a) the words "hereof," "herein" and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement;
- (b) terms defined in the singular shall have a comparable meaning when used in the plural, and vice versa;
  - (c) the terms "dollars" and "\$" mean U.S. dollars;
- (d) references herein to a specific Section, Subsection, Recital, Schedule or Exhibit shall refer, respectively, to Sections, Subsections, Recitals, Schedules or Exhibits of this Agreement;
- (e) wherever the word "include," "includes" or "including" is used in this Agreement, it shall be deemed to be followed by the words "without limitation";
  - (f) references herein to any gender shall include each other gender;
- (g) references herein to any Person shall include such Person's heirs, executors, personal representatives, administrators, successors and assigns; <u>provided</u>, <u>however</u>, that nothing contained in this clause (g) is intended to authorize any assignment or transfer not otherwise permitted by this Agreement;
- (h) references herein to a Person in a particular capacity or capacities shall exclude such Person in any other capacity;
- (i) with respect to the determination of any period of time, the word "from" means "from and including" and the words "to" and "until" each means "to but excluding";
- (j) references herein to any Act, Law or Disclosure Law shall be deemed to refer to such Act, Law or Disclosure Law, as the case may be, as amended, modified, codified, reenacted, supplemented or superseded in whole or in part and in effect from time to time, and also to all rules and regulations promulgated thereunder;
- (k) the headings contained in this Agreement are intended solely for convenience and shall not affect the rights of the parties to this Agreement; and
- (l) if the last day for the giving of any notice or the performance of any act required or permitted under this Agreement is a day that is not a Business Day, then the time for the giving of such notice or the performance of such action shall be extended to the next succeeding Business Day.

#### ARTICLE II

#### THE LIMITED LIABILITY COMPANY

- 2.1 <u>Formation</u>. The Company has been formed as a limited liability company pursuant to the provisions of the Act. A Certificate of Formation for the Company (the "Certificate of Formation") has been filed in the Office of the Secretary of State of the State of Delaware in conformity with the Act. The Company and, if required, each of the Members shall execute or cause to be executed from time to time all other instruments, certificates, notices and documents and shall do or cause to be done all such acts and things (including keeping books and records and making publications or periodic filings) as may now or hereafter be required for the formation, valid existence and, when appropriate, termination of the Company as a limited liability company under the Laws of the State of Delaware and as may be necessary in order to protect the liability of the Members as members under the Laws of the State of Delaware.
- 2.2 Name. The name of the Company shall be "BridgeBioX Pharma, LLC", and its business shall be carried on in such name with such variations and changes as the Board shall determine or deem necessary to comply with requirements of the jurisdictions in which the Company's operations are conducted.
- 2.3 <u>Business Purpose.</u> The Company is formed for the purpose of holding capital stock of any Subsidiary and any activities incidental to such enumerated purposes and to engage in such other businesses or activities as the Board may determine.
- 2.4 <u>Registered Office and Agent.</u> The registered office required to be maintained by the Company in the State of Delaware pursuant to the Act will initially be the office and the agent so designated on the Certificate of Formation. The Company may, upon compliance with the applicable provisions of the Act, change its registered office or registered agent from time to time in the determination of the Board.
- 2.5 Term. The term of the Company commenced on the date of filing of the Certificate of Formation in the Office of the Secretary of State of the State of Delaware and shall continue until the Company is dissolved pursuant to Article X.
- 2.6 <u>Company Powers.</u> The Company shall possess and may exercise all the powers and privileges granted by the Act or by any other law or by this Agreement, together with any powers incidental thereto, insofar as such powers and privileges are necessary, advisable, incidental or convenient to the conduct, promotion or attainment of the business purposes or activities of the Company.
- 2.7 <u>Business Transactions of a Member or Director with the Company.</u> Subject to Section 4.3 and Section 4.10 (as applicable), a Member or Director may transact business with the Company and, subject to applicable Law, shall have the same rights and obligations with respect to any such matter as a Person who is not a Member or a Director.

- 2.8 <u>Principal Place of Business.</u> The principal place of business of the Company shall be at such location as the Board may from time to time select.
- 2.9 <u>Title to Company Property.</u> Legal title to all property of the Company shall be held and vested and conveyed in the name of the Company, and no real or other property of the Company shall be deemed to be owned by any Member individually. The Units of the Members in the Company shall constitute personal property.

# ARTICLE III THE MEMBERS

3.1 <u>The Members.</u> The name, address, number and type of Units and Voting Unit Percentage and the Capital Contribution of each Member is set forth on Schedule A hereto, as such Schedule may be amended from time to time to reflect the admission of any Additional Members or Substitute Members, the acquisition of additional Units by any Member (including with respect to Additional Cash Contributions), and the transfer, repurchase, forfeiture or exchange of Units, each as permitted or required by the terms of this Agreement.

#### 3.2 <u>Member Meetings.</u>

- (a) Actions by the Members; Meetings. Except as expressly contemplated by this Agreement or required by applicable Law, the approval or consent of the Members shall not be required in order to authorize the taking of any action by the Company and/or the Board. The Members may vote, approve a matter or take any action by the vote of the Members holding Voting Units entitled to vote at a meeting, in person or by proxy, or without a meeting by the written consent of the Members pursuant to Section 3.2(b). Meetings of the Members may be called by the Board or the Members holding a Majority Interest and shall be held upon not less than two (2) Business Days nor more than sixty (60) days' prior written notice of the time and place of such meeting delivered to each holder of Voting Units in the manner provided in Section 14.2. Notice of any meeting may be waived by any Member before or after any meeting. Meetings of the Members may be conducted in person or by conference telephone, videoconference or webcast facilities. Members holding Class B Units shall not, in respect of any such Class B Units, be entitled to vote on any matter, nor shall they be entitled to notice of, or to attend, any meeting of the Members.
- (b) Action by Written Consent. Any action required or permitted to be taken at any meeting of Members may be taken by the Members without a meeting if authorized by the written consent of the Members holding Voting Units sufficient to approve such action pursuant to the terms of this Agreement. In no instance where action is authorized by written consent will a meeting of the Members be required to be called or notice be required to be given; provided that the Members entitled to vote but not executing the written consent shall be provided with copies of any executed written consent with respect to any action taken by written consent promptly following the execution thereof. Electronic mail or similar transmission by a Member, or a photographic, facsimile or similar reproduction of a writing signed by a Member, shall be regarded as signed by the Member for purposes of this Section 3.2(b).

- (c) Quorum; Voting. For any meeting of the Members, the presence in person or by proxy of the Members owning Voting Units representing at least a Majority Interest shall constitute a quorum for the transaction of any business. With respect to any matter, other than a matter for which the affirmative vote of the holders of a specified portion of Members entitled to vote is required by the nonwaivable provisions of the Act (if any) or by this Agreement (including in Section 4.10), the affirmative vote of the Members holding a Majority Interest at a meeting of Members at which a quorum is present shall be the act of the Members.
- 3.3 <u>Liability of Members.</u> All debts, obligations, and liabilities of the Company, whether arising in contract, tort, or otherwise, shall be solely the debts, obligations and liabilities of the Company, and no Member shall be obligated personally for any such debt, obligation or liability of the Company solely by reason of being a Member.
- 3.4 <u>Power to Bind the Company.</u> No Member (acting in its capacity as such) shall have any authority to bind the Company to any third party with respect to any matter except pursuant to a resolution expressly authorizing such action which resolution is duly adopted by the Board by the affirmative vote required for such matter pursuant to this Agreement or the Act.

#### 3.5 Pass-Through Rights and Obligations.

- Management Holdco is a special purpose investment vehicle through which its members indirectly (a) own Class B Units. In furtherance of the foregoing, the Members hereby acknowledge and agree that it is their intent that the Management Holdco Class B Holders have the same rights and obligations as they would if they held the Class B Units owned by Management Holdco directly in the same number of Management Holdco Class B Units they own directly. Accordingly, for purposes of this Agreement: (1) the Class B Units held by Management Holdco shall, at any time, together with the rights and obligations set forth in the Management Holdco Operating Agreement, give each holder of Management Holdco Class B Units the same rights and obligations that they would have if they were admitted directly to the Company holding a number of Class B Units in the Company equal to the Management Holdco Class B Units, that such holder owns at such time; and (2) each such Management Holdco Class B Holder will be treated as a Management Holder for purposes of this Agreement. In implementing the foregoing stated intent in this Section 3.5 and in order to determine equitably the rights and obligations of the Company and the Members, the Company shall treat each member of Management Holdco for all applicable purposes of this Agreement as if such member of Management Holdco were a Member holding Class B Units as provided in the immediately preceding sentence of this Section 3.5 and, without limiting the foregoing, the provisions of, with respect to Management Holders, Section 8.1, Section 8.8, Section 8.10, Section 8.11 (with respect to Dragged Member provisions only), Section 14.3, and Article IX shall apply mutatis mutandis to the Management Holdco Operating Agreement as if they were incorporated therein, with any reasonable modifications as the Board deems necessary to give effect to this Section 3.5.
- (b) This <u>Section 3.5</u> is not intended to result in any Person being treated as a "partner" of the Company for U.S. federal income tax purposes if such Person would not otherwise be so treated, including any Management Holdco Class B Holder, and this <u>Section 3.5</u> shall be interpreted and applied consistent with this intent.

3.6 <u>Corresponding Management Holdco Units.</u> It is the intent of the Company that, upon issuance of each Class B Unit or other class of Unit owned by Management Holdco, a corresponding Management Holdco Class B Unit or such other class of Unit (the "<u>Corresponding Management Holdco Unit</u>") shall be issued by Management Holdco. Furthermore, it is the intent of the Company that, upon the forfeiture, repurchase or other termination of any Corresponding Management Holdco Unit, the Company will cancel the corresponding Class B Unit issued to the Management Holdco in respect of such Corresponding Management Holdco Unit. Without limiting <u>Section 3.5</u>, if the Company shall make any subdivision (by any Unit split, Unit dividend or distribution, reclassification, reorganization, recapitalization, exchange or otherwise) or combination (by reverse Unit split, reclassification, recapitalization, exchange or otherwise) of any Class B Units or other class of Units owned by Management Holdco, Management Holdco shall make corresponding equitable adjustments to the Corresponding Management Holdco Units, as applicable. Management Holdco hereby agrees to take, or cause to be taken, all actions necessary or advisable in order to give effect to this <u>Section 3.6</u>.

### 3.7 <u>Keyman Trigger Event.</u>

- (a) As used herein, the term "<u>Keyman Trigger Event</u>" shall mean, the occurrence prior to eighteen (18) months following the Closing of Neil Kumar voluntarily resigning from his Executive Chair position with the Company such that he no longer serves in any officer capacity with the Company or any of its Subsidiaries, except as a result of actions taken by the Board or the Investors.
- If a Keyman Trigger Event occurs, then the Member or Members (excluding BBIO and any of its (b) Affiliates or Permitted Transferees for purposes of this Section 3.7(b)) representing at least a majority of the issued and outstanding Voting Units (which majority must include each of (i) Viking, for so long as the Voting Unit Percentage of Viking is at least twenty percent (20%) and (ii) Patient Square, for so long as the Voting Unit Percentage of Patient Square is at least twenty percent (20%)) shall have the right, subject to the terms hereof, by delivery of a written notice to the Company to require that the Company immediately suspend any unfunded Additional Cash Contributions by the Investors for a period of at least ninety (90) days (the "Suspension Period"). The Company shall promptly notify each Investor of such suspension. The Suspension Period shall terminate if, at any time during the Suspension Period, a replacement is appointed for Neil Kumar; provided that any such replacement must be approved by the Member or Members representing at least a majority of the issued and outstanding Voting Units (which majority must include each of (i) Viking, for so long as the Voting Unit Percentage of Viking is at least twenty percent (20%) and (ii) Patient Square, for so long as the Voting Unit Percentage of Patient Square is at least twenty percent (20%)); provided, further, that Viking, Patient Square and BridgeBio shall reasonably cooperate with the Company during the Suspension Period to consider in good faith a replacement for Neil Kumar. Any successor to Neil Kumar approved by the Board and the requisite Member or Members necessary to declare a Suspension Period at such time pursuant to this Section 3.7(b) shall be deemed to be the subject to the "Keyman Trigger Event" definition and Neil Kumar shall no longer be considered the subject of the "Keyman Trigger Event" definition.

(c) During the thirty (30) day period after the Suspension Period, the Member or Members representing at least a majority of the issued and outstanding Voting Units (which majority must include each of (i) Viking, for so long as the Voting Unit Percentage of Viking is at least twenty percent (20%) and (ii) Patient Square, for so long as the Voting Unit Percentage of Patient Square is at least twenty percent (20%)) shall have the right by delivery of a written notice to the Company during such period to require that the Company initiate a liquidation of the Company and its assets (subject to compliance with the Reversion Rights).

# ARTICLE IV THE BOARD AND OFFICERS

#### 4.1 <u>Management by the Board of Directors.</u>

- (a) General. Subject to Section 4.10 and other terms of this Agreement, the business and affairs of the Company shall be managed, operated, and controlled by and under the direction of the board of managers (the "Board") (and not the Members), and the Board (and not the Members) shall be responsible for policy setting, approving the overall direction of the Company, and making all decisions affecting the business and affairs of the Company. The Board shall have and is hereby granted, the full and complete power, authority and discretion for, on behalf of and in the name of the Company, to take such actions as it may in its sole discretion deem necessary or advisable to carry out any and all of the objectives and purposes of the Company, subject only to the terms of this Agreement and the Act. It is the intent of the parties hereto that each manager of the Company (each a "Director") shall be deemed to be a "manager" of the Company (as defined in Section 18-101(12) of the Act) for all purposes under the Act. The Board shall consist of the number of Directors from time to time as may then be designated in accordance with Section 4.1(b) below. Each Director shall hold office until his or her successor shall have been appointed and qualified or until his or her earlier resignation, removal, death or disability.
- (b) Composition of the Board. Subject to the terms of this Section 4.1(b), the Board shall initially consist of four (4) Directors, designated as follows:
- (i) so long as the Investor Member Sell-Down Event in respect of Viking has not occurred, Viking shall be entitled to separately designate, in its sole discretion, one (1) Director, who shall initially be Eric Aguiar;
- (ii) so long as the Investor Member Sell-Down Event in respect of Patient Square has not occurred, Patient Square shall be entitled to separately designate, in its sole discretion, one (1) Director, who shall initially be James Momtazee;
- (iii) so long as the Investor Member Sell-Down Event in respect of NK has not occurred, NK shall be entitled to separately designate, in its sole discretion, one (1) Director, who shall initially be Neil Kumar; and
  - (iv) Richard Scheller.
- (c) *Indemnification*. The Company shall enter into an indemnification agreement with each Director, in customary form and substance acceptable to the Company and reasonably acceptable to the Board.

- (d) Removal. A Director may only be removed from the Board by (i) the applicable Person entitled to appoint such Director pursuant to Section 4.1(b), with or without cause, or (ii) if there is no appointment right or if such applicable Person is no longer entitled to appoint such Director pursuant to Section 4.1(b) as a result of an Investor Member Sell-Down Event, then by approval of all of the other Directors on the Board.
- (e) *Vacancies*. If at any time a vacancy is created on the Board by reason of the death, removal or resignation of any Director, a designee shall be appointed to fill such vacancy or vacancies by (i) the Person entitled to appoint such Director pursuant to Section 4.1(b) or (ii) if there is no appointment right or if such Person is no longer entitled to appoint such Director pursuant to Section 4.1(b) as a result of an Investor Member Sell-Down Event, then by approval of all of the Directors on the Board.
- (f) Compensation. Except for the payment of salaries, benefits, and other compensation to those Directors who are employees of the Company or one of its Subsidiaries and except for the reimbursement of expenses as provided in Section 4.2(g), no compensation shall be paid to any Director for serving as a Director of the Company or any of its Subsidiaries or any committee thereof, unless otherwise approved by the Board.
- Observers. With respect to (i) each of BBIO, Viking, Patient Square and Frazier, for as long as the Voting Unit Percentage of BBIO, Viking, Patient Square or Frazier, as applicable, is at least five percent (5%), each of BBIO, Viking, Patient Square and Frazier, as applicable, shall be entitled to designate one (1) non-voting Board observer and (ii) Cormorant, for so long as the Investor Member Sell-Down Event in respect of Cormorant has not occurred, Cormorant shall be entitled to designate one (1) non-voting Board observer (each, an "Observer"). The Observers shall be provided copies of all documents as and when given to the Directors and shall be invited to attend any meeting of the Directors (and the Company shall provide the Observers with written notice of any such meeting as and when notice is given to the Directors), except that any such Observer may be excluded from the portion of any meeting of the Board and shall not be entitled to receive any materials or minutes of such meeting or relating to any written consent of the Board, in each case, to the extent necessary to maintain the attorney-client privileged status of such discussion or materials, as reasonably determined by the other members of the Board in good faith. No Observer shall have the right to vote on any matter that comes before the Board. Any Observer may be removed or replaced by the Member that has the right to appoint such observer, for any reason or no reason.
- (h) *Confidentiality Agreement*. Each Director and Observer shall be required to enter into a customary confidentiality agreement with the Company.

## 4.2 <u>Meetings of the Board.</u>

(a) Frequency. Subject to Section 4.2(c), the Board shall meet at such times and at such places as may be necessary for the Company's business, as determined by the Board.

(b) Quorum. The presence of all of the Directors then in office and entitled to vote shall constitute a
quorum at any meeting of the Board. If a quorum is not present at any meeting of the Board, the Directors present at such meeting
may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present. If
a Director is not present at two (2) consecutive meetings for which notice was duly given in accordance with the terms hereof, then the
presence of such Director shall no longer be required to constitute quorum in the subsequent meeting following such two (2) consecutive
adjourned meetings.

- (c) *Notice; Waiver of Notice.* The Board shall meet at such times as may be necessary for the Company's business on at least forty-eight (48) hours' prior written notice from any Director to each Director including the time and place of such meeting, unless the person calling the meeting determines that shorter notice is necessary or appropriate to discuss matters requiring prompt action. Notice of any Board meeting may be waived by any Director either before or after such meeting.
- (d) Required Vote. Each Director shall be entitled to one vote. All actions of the Board shall require the affirmative vote of a majority of the Directors at a duly called meeting at which a quorum is present. Any reference in this Agreement to the affirmative vote of a majority of the Directors shall be deemed to mean a majority of the votes cast by all Directors.
- (e) Meetings. Meetings of the Board (including, any committees thereof) may be conducted in person or by conference telephone, videoconference facilities, or other communications equipment by means of which all the persons participating in the meeting can hear and communicate with each other at the same time. Each Director shall be entitled to participate in any meeting of the Board (whether or not conducted in person) by telephone.
- (f) Action by Written Consent. Subject to Section 4.10, any action required or permitted to be taken at any meeting of the Board may be taken without a meeting if all Directors consent thereto in writing (including by use of electronic portable document format (.pdf), DocuSign or other equivalent alternatives).
- (g) Reimbursement. The Company or one of its Subsidiaries shall reimburse each Director and Observer for the reasonable and documented out-of-pocket travel and accommodation costs incurred by such Director or Observer to attend meetings of the Board or a committee thereof or the board of directors (or similar governing body) of a Subsidiary of the Company.

## 4.3 <u>Certain Board Approval Rights.</u>

- (a) Prior to a Qualified IPO, the Company shall not take, and shall cause each of its Subsidiaries not to take, any of the following actions without (in addition to any approval required pursuant to <u>Section 4.10</u> and <u>Article XIII</u> (to the extent applicable)) the prior written consent or approval of a majority of the entire Board:
- (i) approve any Related Party Transaction, except for any Member's exercise of Preemptive Rights described in Section 5.4;

- (iii) amend, modify, terminate or repeal the Certificate of Formation, this Agreement or any of the other organizational documents of the Company or any of its Subsidiaries;
- (iv) approve any merger or other business combination or consolidation with any other Person or the sale, disposal, transfer or licensing of any material assets or businesses of the Company or its Subsidiaries, in each case, except (A) in accordance with the sale rights described in <u>Section 8.10</u> or (B) in connection with a liquidation following a Suspension Period pursuant to <u>Section 3.7(c)</u>;
- (v) appoint, dismiss or replace the CEO, CFO or any other senior management executive with an annual base salary in excess of \$500,000;
- (vi) dissolve, liquidate, wind up or terminate the legal existence of the Company or any of its Subsidiaries (or, in each case, take any steps thereto) or commence a voluntary proceeding seeking reorganization or other similar relief, including distributing assets upon any such event, making any arrangement with creditors generally, appointing a receiver or administrator or filing for bankruptcy or similar protection under applicable Law, except in connection with a liquidation following a Suspension Period pursuant to Section 3.7(c);
- (vii) change the composition or size of the Board; <u>provided</u>, that any change to the composition or size of the Board that would adversely affect the Board designation rights of any Member shall also require approval of such affected Member; <u>provided</u>, <u>further</u>, that any increase to the size of the Board shall be deemed to adversely affect each Member with Director designation rights;
  - (viii) create or make changes to the composition or size of any committees of the Board;
- (ix) acquire (including by merging or consolidating with, purchasing any equity interest in or a substantial portion of the assets of, licensing, or by any other manner), any assets, securities or any business, corporation, partnership, association or other business organization or division thereof, in any such case with a value in excess of \$10,000,000;
- (x) other than pursuant to Article IX, approve any initial bona fide underwritten public offering and sale of equity interests of any Subsidiary of the Company pursuant to an effective registration statement filed under the Securities Act of 1933, as amended, pursuant to which securities are to be listed on a national securities exchange or other public listing on an accredited exchange, whether by way of direct listing, merger with a special purpose acquisition vehicle or similar transaction;

- (xi) subject to <u>Section 4.3(b)(i)</u>, issue any Units or other equity securities of the Company or any of its Subsidiaries;
  - (xii) make any capital calls, including pursuant to <u>Section 5.1(e)(i)</u>, subject to the Mandatory

Cash Contributions;

- (xiii) commence or settle any litigation where the amount in controversy is reasonably expected to be in excess of \$3,000,000 or where such litigation or settlement could result in a material reputational impact or result in materially impairing the operations of the Company or any of its Subsidiaries or other material equitable remedy against the Company or any of its Subsidiaries;
  - (xiv) incur any indebtedness or issue any debt securities;
- (xv) establish or amend any incentive equity plan, or issue any incentive equity (other than (A) the issuance of Class B Units up to the Class B Limit or (B) the issuance of any incentive equity of any Subsidiary of the Company representing up to fifteen percent (15%) of the fully diluted equity interests of such Subsidiary (including any equity incentive interests outstanding as of the Closing)), pension plan or other material benefit plan;
- (xvi) approve the annual budget and business plan of the Company (and any material deviations therefrom);
- (xvii) approve any material change in the Company's or any of its Subsidiaries' respective businesses (including introducing, discontinuing or materially modifying any line of business);
  - (xviii) appoint or remove the Company's auditors;
- (xix) enter into, amend or terminate any contracts or series of related contracts with a contract value in excess of \$3,000,000 or that would impose restrictions on the ability of the Company or any of its Subsidiaries to freely conduct business;
- (xx) except in accordance with <u>Section 8.10</u>, approve any split, combination, recapitalization or reclassification of Units or other equity securities of the Company or any of its Subsidiaries;
  - (xxi) enter into, amend or terminate any material insurance policies;
  - (xxii) enter into, amend or terminate any joint venture or partnership; and
- (xxiii) approve the issuance of any incentive equity of any Subsidiary of the Company to any holder of a Corresponding Management Holdco Unit, other than pursuant to any anti-dilution rights granted to Charles Homcy with respect to Sub21, Inc. in existence on the date hereof.

,	<b>11</b> \	3.7 . 1.1 . 11	4 *	. 4	
(	b)	Notwithstanding	anything to	the contrary	z herein:

(i) the Company shall have the right, in its sole discretion and without approval of the Board and at the direction of the senior management, to make issuances of (x) Class B Units to Management Holdco for further issuance of Management Holdco Class B Units to directors, officers, employees or scientific or medical consultants of the Company or any of its Subsidiaries or members of the Investment Committee or individuals providing services pursuant to the TSA, in each case, in accordance with <u>Article V</u> of this Agreement and (y) subject to <u>Section 4.3(a)(xxiii)</u>, equity securities of any Subsidiary of the Company to directors, officers, employees or scientific or medical consultants of such Subsidiary or members of the Investment Committee or individuals providing services pursuant to the TSA representing up to fifteen percent (15%) of the fully diluted equity interests of such Subsidiary (including any equity incentive interests outstanding as of the Closing);

(ii) with respect to any Related Party Transaction, the applicable Director designated by a potentially conflicted Member shall recuse themselves from the Board approval of such Related Party Transaction, except for (A) reimbursement of expenses as provided in Section 4.2(g) or (B) issuance of Units or other equity interests or debt securities in accordance with the terms of this Agreement; and

(iii) issuance of any Units or other equity securities of the Company or any of its Subsidiaries in connection with an equity capital raise (excluding equity consideration issued in connection with an acquisition of a business or entity or shares or assets of another Person) from any Person shall require approval of each Investor that has outstanding Additional Cash Contributions commitments (other than an Investor that fails or has failed to make an Additional Cash Contribution when due), except with respect to issuances of any Units or other equity securities pursuant to Section 5.1(e) or Section 5.4.

- 4.4 <u>Power to Bind Company.</u> No Director (acting in his or her capacity as such) shall have any authority to bind the Company to any third party with respect to any matter except pursuant to a resolution expressly authorizing such action which resolution is duly adopted by the Board by the affirmative vote required for such matter pursuant to this Agreement.
- 4.5 <u>Delegation of Authority; Officers.</u> Subject to <u>Section 4.10</u>, the Board shall have the authority to appoint and terminate officers of the Company and the Board shall take all necessary actions to cause such appointment or termination of such officers. The Board shall have the authority to retain and terminate employees, agents and consultants of the Company and to delegate such duties to any such officers, employees, agents and consultants as the Board deems appropriate, including the power, acting individually or jointly, to represent and bind the Company in all matters, in accordance with the scope of their respective duties.

#### (a) General.

(i) The officers of the Company shall initially include an Executive Chair and President and may include a CEO and CFO and other officers the Board may from time to time appoint. Any number of offices may be held by the same person, unless otherwise prohibited by Law. The officers of the Company need not be Members of the Company nor, except in the case of the Executive Chair, need such officers be Directors of the Company.

(ii)	Subject to Section 4.10,	the officers of the Com	pany shall hold their off	ices for such terms
and shall exercise such powers and perfor	rm such duties as shall be d	letermined from time to	o time by the Board, and	d all officers of the
Company shall hold office until their succ	cessors are chosen, or until	I their earlier death, dis	sability, resignation or r	removal. Subject to
Section 4.10, any officer elected by the Bo	oard may be removed at any	time, with or without (	Cause, by the affirmative	vote of the Board.
Subject to Section 4.10, any vacancy occur	rring in any office of the Co	ompany shall be filled b	by the Board. The Board	may delegate such
duties to any such officers or other emplo	yees, agents and consultant	ts of the Company as t	he Board deems approp	riate, including the
power, acting individually or jointly, to rep	present and bind the Compa	any in all matters, in ac	cordance with the scope	of their respective
duties	_	-	_	_

(iii) The Board shall delegate the day-to-day operation of the Company to officers, including to elect or appoint any other officer.

#### (b) Officers.

- (i) <u>Executive Chair</u>. So long as Neil Kumar continues to serve on the Board, Neil Kumar shall act as an Executive Chair of the Board ("<u>Executive Chair</u>") and preside at all meetings of the Board. The Executive Chair may serve as institutional representative of the Company before private and public third parties, but shall not have any rights or powers different from any other Directors other than with respect to procedural matters to the extent delegated by the Board; <u>provided</u>, <u>however</u>, that any such procedural rights or powers granted to the Executive Chair shall not be in derogation of any rights or powers granted by this Agreement to any Member or Director.
- (ii) <u>Chief Executive Officer</u>. If appointed by the Board, the chief executive officer (the "<u>CEO</u>") of the Company, subject to the powers of the Board, shall have general charge of the business, affairs and property of the Company, and control over its officers, agents and employees, and shall see that all orders and resolutions of the Board are carried into effect. The CEO shall have such other powers and perform such other duties as may be assigned by the Board.
- (iii) <u>Chief Financial Officer</u>. If appointed by the Board, the chief financial officer (the "<u>CFO</u>") of the Company, subject to the powers of the Board, shall be responsible for all administrative, financial, external financing, treasury, tax and insurance matters of the Company and its Subsidiaries. The CFO shall have such other powers and perform such other duties as may be assigned by the Board.
- (iv) <u>President</u>. The president (the "<u>President</u>") of the Company shall have such powers and perform such duties as may be assigned by the Board. The initial President shall be James Li.
- (c) <u>Resignation and Removal</u>. Any officer may resign as such at any time. Such resignation shall be made in a written notice to the Board and shall take effect at the time specified therein, or if no time is specified, at the time of its receipt by the Board. The acceptance of a resignation shall not be necessary to make it effective, unless expressly so provided in the resignation.

#### 4.6 Committees.

- (a) The Board may designate one or more committees and delegate thereto such power and authority as is permitted by Law and as may be set forth in the resolutions creating any such committee. Each committee shall keep regular minutes and report to the Board when required. Subject to Section 4.10, the members of any such committees shall be appointed or removed by the majority of the entire the Board. Any committee may adopt a charter, which may establish requirements or procedures relating to the governance and operation of such committee.
- (b) Without limiting the foregoing, there shall be an Investment Committee composed of three or more Directors, officers or other service providers of the Company or other individuals selected by the Board. Initial members of the Investment Committee shall be Charles Homey, Richard Scheller, Frank McCormick, Len Post, Phil Reilly and Andrew Lo. The Investment Committee shall have such power and authority as is permitted by Law and as delegated to the Investment Committee by the Board.
- 4.7 <u>Reliance by Third Parties.</u> Persons dealing with the Company are entitled to rely conclusively upon the power and authority of the Board.

## 4.8 <u>No Fiduciary Duties.</u>

- (a) Except as otherwise expressly provided herein or in any written agreement entered into by such Person and the Company and to the maximum extent permitted by the Act, no present or former Director nor any such Director's Affiliates, employees, agents or representatives shall be liable to the Company or to any Member for any act or omission performed or omitted by such Person in its capacity as a Director; <u>provided</u> that, except as otherwise provided herein, such limitation of liability shall not apply to the extent the act or omission was attributable to such Person's fraud or bad faith as determined by a final judgment, order or decree of an arbitrator or a court of competent jurisdiction (which is not appealable or with respect to which the time for appeal therefrom has expired and no appeal has been perfected). Each Director shall be entitled to rely upon the advice of legal counsel, independent public accountants and other experts, including financial advisors, and any act of or failure to act by such Director in good faith reliance on such advice shall in no event subject such Person or any of such Person's Affiliates, employees, agents or representatives to liability to the Company or any Member.
- (b) To the fullest extent permitted by the Act, and notwithstanding any other provision of this Agreement or in any agreement contemplated herein or applicable provisions of Law or equity, no Covered Person shall have any duties at law or in equity (including any fiduciary duties) to the Company or any Member or Director, or any other Person arising out of, relating to, or in connection with the Company, the conduct of the Company's business and affairs, or any action or omission taken or omitted to be taken, or consent or approval given or withheld, in each case, in such Person's capacity as a Covered Person, whether pursuant to this Agreement or otherwise; provided that the foregoing shall not eliminate the implied contractual covenant of good faith and fair dealing, to the extent applicable. To the fullest extent permitted by Law, each Member hereby acknowledges, agrees and consents to the foregoing waiver of duties (including fiduciary duties) and agrees not to make or threaten to make, and shall cause its controlled Affiliates not to make or threaten to make any claim or bring any action, suit or other proceeding or take any similar action, in each case alleging that any Covered Person shall have breached any such duty (including any fiduciary duty). Notwithstanding any provision to the contrary in this Section 4.8, nothing herein will limit any duties of any officer or employee of the Company or any of its Subsidiaries.

(c) Notwithstanding any other provision of this Agreement or any applicable provision of Law or in equity, whenever the Board or any Covered Person takes any action (or omits to take any action), or is permitted or required to make any decision or determination with respect to the Company or its business and affairs, whether pursuant to the terms of this Agreement or otherwise, then, to the fullest extent permitted by Law, the Board and each Covered Person shall be entitled to take such action (or omit to take such action), or to make such decision or determination, in its sole and absolute discretion, and shall be entitled to consider only such interests and factors as it desires, including its own interests, and shall have no duty or obligation (fiduciary or otherwise) to give any consideration to any interest of or factors affecting the Company, any of the Members, any of their respective Affiliates or any other Person, and shall not be subject to any other or different standards that may otherwise apply under applicable Law or in equity. For all purposes of this Agreement, each Covered Person (acting in its capacity as such) and the Board, acting on behalf of the Company or in connection with the Company's business and affairs, shall be conclusively presumed to be acting in good faith if such Person (or, in the case of the Board, a majority of the Directors participating in the decision) subjectively believe(s) that the action taken (or omitted to be taken), the consent or approval given or withheld, or the decision or determination made or not made, is in or is not opposed to the best interests of the Company.

## 4.9 <u>Corporate Opportunities.</u>

(a) None of the Members or any of their respective Permitted Transferees or any of the Directors (unless also an officer of the Company or any of its Subsidiaries (including members of the Investment Committee), then only in such capacity as an officer), nor any of their Affiliates, other equityholders or Representatives (the "Applicable Persons") shall have any duty to communicate or present an investment or business opportunity or prospective economic advantage to the Company in which the Company may otherwise have an interest or expectancy (a "Corporate Opportunity"). No Applicable Person shall be deemed to have breached any fiduciary or other duty or obligation to the Company by reason of the fact that any such Applicable Person pursues or acquires a Corporate Opportunity for itself, its Affiliates or its Permitted Transferees, directs, sells, assigns or transfers such Corporate Opportunity to another Person or does not communicate information regarding such Corporate Opportunity to the Company.

(b) The Company and each Member acknowledges that the Applicable Persons may be engaged in the business of investing in private and public companies in a wide range of industries, including the industry segment in which the Company and its Subsidiaries operate (the "Company Industry Segment"). Accordingly, the Company, its Subsidiaries, and the Members each acknowledge and agree that, without limiting Section 14.3 or any confidentiality agreement entered into between the Company or its Subsidiaries and any other Person, the Applicable Persons shall:

- (i) have no duty to the Company or any other Member to refrain from participating as a director, investor or otherwise with respect to any company or other Person that is engaged in the Company Industry Segment or is otherwise competitive with the Company or any of its Subsidiaries, unless as otherwise set forth in any restrictive covenant agreement between the Company (or any of its Subsidiaries) and such Applicable Person; and
- (ii) in connection with making investment decisions, to the fullest extent permitted by Law, have no obligation of confidentiality or other duty to the Company or any Members to refrain from using any general information, such as market trend and market data, which comes into the Applicable Persons' possession, whether as a director, investor or otherwise (the "Information Waiver"); provided that the Information Waiver shall not apply, and therefore such Member shall be subject to such obligations and duties of confidentiality as would otherwise apply to such Member under applicable Law, if the information at issue (A) constitutes nonpublic, confidential or proprietary information concerning the Company, any of its Subsidiaries or the business of the Company or its Subsidiaries or (B) to the extent covered by a contractual obligation of confidentiality to which the Company is a party. Notwithstanding anything in this Section 4.9(b)(ii) to the contrary, nothing herein shall be construed as a waiver of any Member's obligation of confidentiality with respect to the disclosure of confidential information of the Company or any of its Subsidiaries.
- (c) Notwithstanding any other provision of this Agreement, any officer of the Company or any of its Subsidiaries (including members of the Investment Committee) who is also a director or officer of BBIO or its Subsidiaries (including members of BBIO's investment committee) (each, a "Specified Person") shall not have any duty to communicate or present a Corporate Opportunity to the Company.
- (d) Notwithstanding Section 4.9(c), until the fifth (5th) anniversary of the Closing, each Specified Person shall (i)(A) present to BBIO and its investment committee any corporate opportunities with respect to the BBIO Indications (to the exclusion of the Company) (the "BBIO Indication Opportunities") and (B) present to the Company and its Investment Committee any business opportunities with respect to the Company Indications (to the exclusion of BBIO) (the "Company Indication Opportunities") and (ii) with respect to business opportunities that are neither BBIO Indication Opportunities nor Company Indication Opportunities (such business opportunities, "Whitespace Opportunities"), present first to the Board of Directors of BBIO, provided that any such Whitespace Opportunity (i) developed by a full-time employee of the Company without relying on BBIO or any of BBIO's intellectual property or confidential information or (ii) related to (A) GNAS Fibrous dysplasia, (B) PMM2 Congenital glycosylation disorder type Ia, (C) ADPKD Autosomal dominant polycystic kidney diseases or (D) PRSS1 Inhibitor Pancreatitis, in each case, shall be excluded from the definition of "Whitespace Opportunities" and shall be presented first to the Company (the foregoing clause (d), the "Specified Corporate Opportunities Allocation"). The Specified Corporate Opportunities Allocation shall expire on the fifth (5th) anniversary of the Closing.

(e) Pursuant to the Transaction Agreement, during the five-year period following the Closing, the Board
of Directors of BBIO shall have thirty (30) days from being presented a Whitespace Opportunity from a Specified Person to determine
whether to pursue such Whitespace Opportunity, in its sole discretion (including to the exclusion of the Company), and if the Board of
Directors of BBIO determines not to pursue such Whitespace Opportunity, or does not affirmatively determine to have BBIO pursue such
Whitespace Opportunity within such thirty (30) day period, then the Board will be presented with and have the ability to pursue such
Whitespace Opportunity.

(f) Any Person purchasing or otherwise acquiring any interest in any Units shall be deemed to have notice of and to have consented to the provisions of this <u>Section 4.9</u>.

#### 4.10 Certain Member Approval Rights.

- (a) Notwithstanding any other provision of this Agreement to the contrary (but without limiting the rights set forth in Section 4.3(b)(i) of this Agreement), without the prior written consent of each of Viking, Patient Square and BridgeBio for so long as such Member, together with such Member's Affiliates and Permitted Transferees, holds a Voting Unit Percentage of at least ten percent (10%), the Company shall not, and shall not permit any of its Subsidiaries to, and the Board shall not approve the Company or a Subsidiary, or any officer thereof, directly or indirectly, by amendment, merger, recapitalization, sale, consolidation or otherwise, to:
- (i) cause the Company to be treated for U.S. federal income tax purposes as an association taxable as corporation (excluding any Subsidiary that is classified as such on the date hereof);
- (ii) approve any Related Party Transaction, except for any Member's exercise of Preemptive Rights described in Section 5.4;
- (iii) dissolve, liquidate, wind up or terminate the legal existence of the Company or any of its Subsidiaries (or, in each case, take any steps thereto) or commence a voluntary proceeding seeking reorganization or other similar relief, including distributing assets upon any such event, making any arrangement with creditors generally, appointing a receiver or administrator or filing for bankruptcy or similar protection under applicable Law; except in connection with a liquidation following a Suspension Period pursuant to Section 3.7(c);
- (iv) amend, modify, terminate or repeal the Certificate of Formation, this Agreement, or any of the other organizational documents of the Company or any of its Subsidiaries, in each case, in a manner that would (A) have a disproportionate and adverse effect on such Member, as compared to the other Members holding the same class of Units, as a collective, or (B) adversely alter any of the rights, privileges, preferences or obligations of such Member under this Agreement; provided, further, subject to compliance with Section 5.1(e) and Section 5.4, the authorization and issuance of additional Units at the direction of the Board and the amendment of this Agreement and Schedule A to reflect the terms of such additional Units, even if the issuance of such additional Units would have a dilutive effect on one or more classes of Units shall not, in and of itself, require approval under this Section 4.10(a)(iv);

(v)	incur any indebtedness or issue any debt securities, in either case in one or series of related
transactions with an aggregate principal amo	unt in excess of \$10,000,000;

- (vi) create or make changes to composition or size of any committees of the Board;
- (vii) effect any fundamental changes to the Company's or any of its Subsidiaries' business;
- (viii) approve any (A) non-pro rata dividends, Distributions (other than Tax Distributions) or repurchases of Units or other equity securities or (B) pro-rata dividends, Distributions (other than Tax Distributions) or repurchases of Units or other equity securities prior to January 1, 2026, in each case, other than repurchases of Units or other equity securities held by directors, officers, employees or other service providers of the Company or any of its Subsidiaries pursuant to Article VIII of this Agreement and/or the Board-approved terms applicable to such Units or other equity securities, including under any Applicable Class B Unit Agreement, in each case, at a purchase price that is no greater than the applicable Call Price as determined in accordance with Section 8.8 of this Agreement;
- (ix) approve any merger or other business combination or consolidation with any other Person or the sale, disposal, transfer or licensing of any material assets or businesses of the Company or its Subsidiaries, in each case, except in accordance with the sale rights described in <u>Section 8.10</u> or in connection with a liquidation following a Suspension Period pursuant to <u>Section 3.7(c)</u>;
- (x) approve any sale, disposition, out-license or other monetization event of any material In-Scope Program, except in accordance with the sale rights described in <u>Section 8.10</u> or in connection with a liquidation following a Suspension Period pursuant to <u>Section 3.7(c)</u>;
  - (xi) appoint, terminate or replace the CEO;
  - (xii) enter into, amend or terminate any joint venture or partnership; or
- (xiii) other than a Qualified IPO pursuant to Article IX, approve any initial bona fide underwritten public offering and sale of equity interests of any Subsidiary of the Company pursuant to an effective registration statement filed under the Securities Act of 1933, as amended, pursuant to which securities are to be listed on a national securities exchange or other public listing on an accredited exchange, whether by way of direct listing, merger with a special purpose acquisition vehicle or similar transaction.

The provisions of this Section 4.10 shall terminate upon the consummation of a Qualified IPO.

- 4.11 Reversion Rights. Subject to Section 4.3 and Section 4.10, in the event of a final decision by the Board to fully shut down and discontinue an In-Scope Program and not transfer it to a third-party for consideration determined in good faith by the Board to be reasonable or save it for a later date, as determined by the Board in good faith, BBIO will have the right to re-acquire such In-Scope Program (the "Reversion Rights"). The terms of such re-acquisition shall be subject to the mutual agreement of the Board and BBIO, and BBIO will be required to invest in such programs upon reversion to the extent determined by the Board of Directors of BBIO (in its sole discretion) at the time of the re-acquisition. In the event that the Board and BBIO, each acting reasonably and in good faith, are unable to reach mutual agreement on the terms of such re-acquisition within thirty (30) days, the Reversion Rights shall cease to apply to such In-Scope Program.
- 4.12 <u>Insurance.</u> As determined by the Board, the Company may maintain one or more key-person insurance policies for certain employees of the Company and/or its Subsidiaries with the Company as beneficiary of such policy. Such policy shall contain terms and conditions (including the amount of coverage and other similar matters) satisfactory to the Board. The Members shall cooperate reasonably with the Company, and shall use reasonable efforts to cause employees proposed to be or otherwise subject to any such key-person insurance policies to cooperate with the Company, in order for the Company to establish any such key-person insurance policies, including by submitting to reasonably required examinations and tests and by preparing, signing and delivering such applications, statements of health, and other documents as may be reasonably required. In addition, the Company shall purchase and at all times maintain customary directors' and officers' liability insurance to cover its and its Subsidiaries' directors and officers in amounts satisfactory to the Board.

# ARTICLE V CAPITAL STRUCTURE AND CONTRIBUTIONS

## 5.1 <u>Capital Structure.</u>

(a) General. Subject to the terms of this Agreement, the Company is authorized to issue equity interests in the Company designated as "Units", which shall constitute limited liability company interests under the Act and shall include initially Class A Units and Class B Units (whether any such Units are issued on or after the date hereof). Subject to Section 5.4, the Board shall have the authority to cause the Company to issue such number of Units of any class, series or tranche as may be determined by the Board from time to time. The Company is authorized to issue options or warrants to purchase Units, restricted Units, Unit appreciation rights, phantom Units, and other securities convertible, exchangeable or exercisable for Units on such terms as may be determined by the Board from time to time. If the Board authorizes any subdivision, split, combination or reclassification of Units or any other similar action, if necessary or appropriate, the Board shall, acting in good faith, appropriately adjust the economic provisions of this Agreement to reflect such action, including by amending the distribution provisions of Section 6.4 and associated definitions. Other than as set forth in this Agreement or in the instruments governing the terms of any Units issued pursuant to this Section 5.1(a), each Unit shall be identical in all respects with each other Unit.

- (b) Class A Units. The Class A Units shall have such rights to allocations and Distributions as may be authorized and set forth under this Agreement. The relative rights, powers, preferences, duties, liabilities and obligations of holders of the Class A Units shall be as set forth herein. Each holder of Class A Units shall be entitled to vote, in person or by proxy, on a pro rata basis in accordance with the Voting Unit Percentage for each Member as of the applicable date and time on all matters upon which Members have the right to vote as set forth in this Agreement and provided under the Act.
- Class B Units. The Company may provide for the issuance and terms of Class B Units as and when corresponding Management Holdco Class B Units are issued by Management Holdco representing, in the aggregate, up to twenty percent (20%), on a fully diluted basis (including after giving effect to the issuance of the total number of Class B Units) (the "Class B Limit"), of the sum of (A) the Class A Units of the Company outstanding as of the Closing and (B) the number of Class A Units issuable as a result of all Additional Capital Contributions being made in accordance with Section 5.1(e)(i) (such sum, the "Committed Equity Amount") to provide equity incentive compensation to selected Directors, officers, employees and other service providers of the Company and its Affiliates (including its Subsidiaries); provided, that ten percent (10%) of such Class B Units shall remain uncommitted and unallocated following the initial allocation of Class B Unit grants so that they may be granted following the Closing to Directors, officers, employees and other service providers of the Company; provided, further, that the maximum amount of Class B Units that can be Vested Class B Units at any given time shall be equal to the product of (1) the Class B Limit and (2) a fraction, the numerator of which is the sum of (x) the number of Class A Units of the Company outstanding as of the Closing and (y) the number of Class A Units issued as a result of Additional Capital Contributions that have been made as of such time, and the denominator of which is the Committed Equity Amount. It is the intention of the parties to this Agreement that each Class B Unit and each corresponding Management Holdco Class B Unit constitutes a Profits Interest. The Company shall have the authority to issue such number of Class B Units and corresponding Management Holdco Class B Units (subject to the Class B Limit) as the Company shall from time to time determine with such terms, conditions, rights and obligations, including forfeiture and repurchase, as may be determined by the Company as set forth herein, in the Management Holdco Operating Agreement and in the related Applicable Class B Unit Agreements pursuant to which any Management Holdco Class B Unit may be awarded; provided that each such Class B Unit and corresponding Management Holdco Class B Unit shall vest in equal annual installments over a four-year period from the date of grant. Members shall not have any voting rights with respect to their Class B Units. Subject to compliance with Section 13.1(a) of this Agreement, the Company shall be permitted to take all actions, including amending this Agreement, to the extent necessary or appropriate, to cause the Class B Units to be treated as Profits Interests for all U.S. federal income tax purposes and valued based on liquidation value or similar principles. Each Member shall respect any allocations of income to be made to any such Member even if such Class B Units are subject to risk of forfeiture. The Company and all Members will (A) treat such Class B Units as outstanding for tax purposes, (B) treat any holder of Class B Units as a Member of the Company for tax purposes with respect to such Class B Units and (C) file all tax returns and reports consistently with the foregoing.
- (d) <u>Section 83(b)</u> *Election*. Notwithstanding any provision of this Agreement to the contrary, each Member shall execute and deliver a valid and timely election under Section 83(b) of the Code with respect to their Class B Units to both the Internal Revenue Service and the Company within thirty (30) days of the date of receipt of any Class B Units.

## (e) Issuance of Additional Units.

(i) Each Investor shall (A) if the Company has less than \$53,000,000 in cash on its balance sheet as of January 1, 2025, promptly following January 1, 2025 but in no event later than January 15, 2025, make an additional cash contribution of its pro rata portion of the amount equal to (x) \$53,000,000 minus (y) the amount of cash on the Company's balance sheet as of January 1, 2025 (the "Mandatory Cash Contributions") and (B) make additional cash contributions at such times as determined by the Board with at least ten (10) days' notice to such Investor ("Board Determined Cash Contributions" and together with the Mandatory Cash Contributions, the "Additional Cash Contributions"), in each case, up to the aggregate amount of the total committed funding set forth opposite of such Investor's name on Schedule B (inclusive of the cash contribution made at the Closing and the Mandatory Cash Contribution) (the "Total Funding Commitments"), in each case, in exchange for additional newly issued Class A Units based on a per Class A Unit price of \$1.00 (the "Price Per Unit"); provided, that any additional Board Determined Cash Contribution shall be allocated pro rata among the Investors as of any date of determination according to a percentage determined by dividing (A) the number of Class A Units held by such Investor as of such date by (B) the aggregate number of Class A Units held by all of the Investors as of such date (assuming for this purpose that all Investors have timely funded all prior Additional Cash Contributions); provided, further, to the extent any portion of the Total Funding Commitments from each Investor remains uncalled by the Board as of immediately prior to the fifth (5th) anniversary of the Closing, then the funding obligation with respect to such remaining portion of the Total Funding Commitments from each Investor shall expire on the fifth (5th) anniversary of the Closing and no Investor shall have any further obligations in respect thereof.

(ii) Subject to the express provisions of this Agreement, including Section 5.1(e)(i) and Section 5.4, the Company is authorized to issue Units to any Person at such prices per Unit as may be determined by the Board to be the fair market value and in exchange for contributions of cash or property, the provision of services or other consideration (collectively, "Unit Consideration"), as may be determined in good faith by the Board. Subject to the express provisions of this Agreement, including Section 5.2 and Section 5.4, the number of Units held by each Member shall not be affected by any (A) issuance by the Company of Units to other Members or (B) change in the Capital Account of such Member. If necessary or appropriate in the Board's discretion (in accordance with this Agreement), the Board shall, acting in good faith, appropriately adjust the economic provisions of this Agreement to reflect the issuance of additional Units pursuant to this Section 5.1(e) that are of a new class not previously reflected in this Agreement, including by amending the distribution provisions of Section 6.4 and associated definitions.

(f) Certificates; Legend. In the sole discretion of the Board, the issued and outstanding Units may be represented by certificates. In addition to any other legend required with respect to a particular class, group or series of Units or pursuant to any other agreement among Members and the Company, each such certificate, if applicable, shall bear the following legend:

"THE UNITS REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO, AND MAY NOT BE SOLD, ASSIGNED, TRANSFERRED, GRANTED A PARTICIPATION IN OR REFERENCE UNDER A DERIVATIVES CONTRACT OR ANY OTHER ARRANGEMENT, PLEDGED, MORTGAGED, ENCUMBERED OR OTHERWISE DISPOSED OF ("TRANSFERRED") WITHOUT COMPLYING WITH, THE PROVISIONS OF THE AMENDED AND RESTATED LIMITED LIABILITY COMPANY AGREEMENT, BY AND AMONG THE MEMBERS OF BRIDGEBIOX PHARMA, LLC (THE "COMPANY"), AS IT MAY BE AMENDED FROM TIME TO TIME, A COPY OF WHICH IS ON FILE WITH THE COMPANY. IN ADDITION TO THE RESTRICTIONS ON TRANSFER SET FORTH IN SUCH AGREEMENT, NO TRANSFER OF THE UNITS REPRESENTED BY THIS CERTIFICATE MAY BE MADE EXCEPT (A) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND THE RULES AND REGULATIONS IN EFFECT THEREUNDER, AND ALL APPLICABLE STATE SECURITIES LAWS OR (B) IF SUCH TRANSFER IS PURSUANT TO AN EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE ACT. THE HOLDER OF THIS CERTIFICATE, BY ACCEPTANCE OF THIS CERTIFICATE, AGREES TO BE BOUND BY ALL OF THE PROVISIONS OF THE AFORESAID AGREEMENT."

#### 5.2 <u>Defaulting Investor; Dilution.</u>

- (a) In the event that an Investor does not make, in whole or in part, any portion of its Additional Cash Contributions when due (such Investor, the "<u>Defaulting Member</u>", and such amount, the "<u>Defaulted Cash Contribution Amount</u>") on or before the date that such contribution is required to be made, such Defaulting Member shall have thirty (30) days from the day such contribution amount was originally due to cure such Defaulted Cash Contribution Amount (such period, the "<u>Cure Period</u>").
- (b) If the Defaulting Member fails to satisfy the Defaulted Cash Contribution Amount within the Cure Period then:
- (i) interest shall accrue on the Defaulted Cash Contribution Amount at an annual rate equal to the Prime Rate from the due date until the date on which the Defaulted Cash Contribution Amount, together with accrued but unpaid interest thereon, is paid in full by the Defaulting Member to the Company; and
- (ii) the Defaulting Member's Preemptive Rights, voting rights and rights to participate in the management of the business of the Company (including to act as or appoint a Director, participation in any committee of the Board and other decisions relating to the Company or any of its Subsidiaries) shall be suspended and the Defaulting Member's Units shall be deemed not to be outstanding for purposes of any voting or approval rights; <u>provided</u> that the Defaulting Member shall have the right to restore its governance rights by paying the Defaulted Cash Contribution Amount, together with accrued but unpaid interest thereon, to the Company on or prior to the date that is sixty (60) days after the end of the Cure Period.
- (c) If Defaulting Member fails to satisfy the Defaulted Cash Contribution Amount, together with accrued but unpaid interest thereon, on or prior to the date that is sixty (60) days after the end of the Cure Period, then the Company shall reduce the number of Class A Units held by the Defaulting Member by the number equal to the product of (y) quotient of (A) Defaulted Cash Contribution Amount (together with accrued but unpaid interest thereon) *divided by* (B) the Price Per Unit, *multiplied by* (z) two (2); provided that the number Class A Units held by a Defaulting Member shall not be less than zero (0). Without limiting the foregoing, the Members agree that this Section 5.2 shall not be deemed to limit any other rights or remedies of the Company or any of its Affiliates under this Agreement.

5.3 <u>Unit Schedule.</u> The number and type of Units issued to Members shall be listed on Schedule A hereto, which shall be amended from time to time as required to reflect issuances of Units to Additional Members, Transfers of Units to Substitute Members, acquisition of additional Units by Members, repurchase of Units and to reflect the cessation or withdrawal of Members.

## 5.4 <u>Preemptive Rights.</u>

General. If prior to a Qualified IPO, the Company or any of its Subsidiaries issues any (i) equity (a) securities, (ii) debt securities or (iii) any securities exercisable for, or convertible or exchangeable into equity or debt securities of the Company or any of its Subsidiaries (all items referred to in clauses (i), (ii) and (iii) are collectively referred to as "Additional Securities"), each Member holding Class A Units (the "Applicable Holders") shall have the right to purchase, upon the same terms and conditions (the "Preemptive Right"), up to its Preemptive Portion of such Additional Securities. Nothing in this Section 5.4 shall apply to any Excluded Issuance. Such Preemptive Right will be offered to each Applicable Holder by written notice (a "Participation Notice") given by the Company to the Applicable Holders at least fifteen (15) days prior to the proposed issuance date specifying therein the amount of the Additional Securities being offered, the purchase price and other material terms and conditions of such offer (such offer, the "Offer"). Each Applicable Holder shall have a period of ten (10) days from and after the date of the Participation Notice within which to irrevocably accept such Offer in whole or in part (the "Preemptive Right Period"). If an Applicable Holder fails to give written notice to the Company within the Preemptive Right Period, such Applicable Holder shall be deemed to have declined such Offer. If any Applicable Holder or Applicable Holders fail to accept the Offer with respect to any Additional Securities in whole within the Preemptive Right Period (any such Additional Securities that were not accepted by such Applicable Holder or Applicable Holders, "Non-Exercised Securities"), then, promptly following the Preemptive Right Period, the Company will provide written notice (an "Undersubscription Notice") to each Applicable Holder that did accept the Offer in whole with respect to the Additional Securities it was entitled to acquire in the Offer (each a "Fully-Exercising Applicable Holder") and each such Fully-Exercising Applicable Holder will have a period of five (5) days following the date of the delivery of the Undersubscription Notice to irrevocably elect to purchase (in addition to the Additional Securities accepted pursuant to the Offer) up to a number of Non-Exercised Securities equal to the product of (A) the total number of Non-Exercised Securities multiplied by (B) a fraction, the numerator of which is such Fully-Exercising Applicable Holder's Preemptive Portion and the denominator of which is the aggregate Preemptive Portions of all Fully-Exercising Applicable Holders. A Fully-Exercising Applicable Holder will be deemed to have declined such offer to purchase Non-Exercised Securities if it fails to give written notice to the Company of its acceptance of such offer within such five (5) day period. The procedures described in the two immediately preceding sentences shall be repeated, iteratively (with only those Applicable Holders who fully exercise their rights to acquire Non-Exercised Securities pursuant to the most recent Undersubscription Notice having the right to participate in any subsequent offer of remaining Non-Exercised Securities), until either all Additional Securities have been subscribed for, or no such Fully-Exercising Applicable Holders elect to purchase any remaining Non-Exercised Securities. All Non-Exercised Securities that any Fully-Exercising Applicable Holder so elects to acquire shall be deemed to be added to the Offer accepted by such Fully-Exercising Applicable Holder. Notwithstanding the foregoing, if the Company or any of its Subsidiaries offers Additional Securities and the Company is not able to comply with, or it would be commercially impracticable to comply with, the time period above prior to the closing of the issuance, with respect to such Additional Securities, the Company or such Subsidiary shall comply with its obligations to the Applicable Holders under this Section 5.4 as promptly as reasonably practicable following the closing of such issuance (in accordance with Section 5.4(e)).

- (b) Closing. The closing with respect to any Additional Securities to be purchased by an Applicable Holder shall be held within thirty (30) days of the date of such Applicable Holder's election to purchase or on such other date as the Company and such Applicable Holder shall agree; provided, however, that the Company may, in its discretion, hold one closing with respect to the purchase of all Additional Securities on a date determined by the Company to be not less than fifteen (15) and not more than sixty (60) days after the last Applicable Holder or accepts or declines (or the applicable offer period expires with respect to) any Offer in connection with the same offering of Additional Securities.
- (c) Offer and Sale. If any Additional Securities are not subscribed for pursuant to accepted Offers under Section 5.4(a), any of such Additional Securities not purchased pursuant to such Offer may be offered for sale and sold by the Company to any other Person as determined by the Board for a period of one hundred twenty (120) days from the last day of Preemptive Right Period, but only for an amount no lower than the price per Additional Security set forth in the initial Offer (and on other terms that are not materially more favorable to such other Person from those offered to the Applicable Holders) to the Applicable Holders made in accordance with this Section 5.4.
- (d) Securities Law Considerations. In the event that the issuance of Additional Securities to an Applicable Holder would require under applicable Law the registration or qualification of such securities or of any Person as a broker or dealer or agent with respect to such securities where such registration or qualification is not otherwise required for such issuance, such Applicable Holder shall not have the right to participate in the issuance. Without limiting the generality of the foregoing, it is understood and agreed that neither the Applicable Holders nor the Company shall be under any obligation to effect a registration of such securities under the Securities Act or similar state statutes except as otherwise specifically provided herein.
- (e) Other Preemptive Rights Holder. Notwithstanding anything to the contrary contained herein, in the event the Board determines in good faith that exigent circumstances exist with respect to complying with the mechanics of Section 5.4(a), 5.4(b) and 5.4(c), the Company may, in order to expedite the issuance of the Additional Securities subject to this Section 5.4 (the "Preemptive Securities"), issue all or a portion of such Additional Securities to one or more Persons (each an "Initial Subscribing Holder"), without complying with the provisions of this Section 5.4; provided that the Company agrees to offer to sell an additional amount of such Additional Securities to each Applicable Holder who is not an Initial Subscribing Holder (each such holder, an "Other Preemptive Rights Holder") in an amount and manner which provides such Other Preemptive Rights Holder with rights substantially similar to the rights outlined in Section 5.4(a), 5.4(b) and 5.4(c). Such Additional Securities shall be offered to the Other Preemptive Rights Holders in a manner such that such Other Preemptive Rights Holders shall be in substantially the same position as they would have been in if the preemptive rights process in Section 5.4(a) had been effected. The Company shall offer to sell such Additional Securities to each Other Preemptive Rights Holder within ten (10) days after the closing of the purchase or issuance of the Preemptive Securities by or to the Initial Subscribing Holder.

- (f) Further Assurances. Each Applicable Holder shall take or cause to be taken all such reasonable actions as may be necessary or reasonably desirable in order to consummate expeditiously each issuance pursuant to this Section 5.4, including executing, acknowledging and delivering consents, assignments, waivers and other documents or instruments; filing applications, reports, returns, filings and other documents or instruments with Governmental Authorities, and otherwise reasonably cooperating with the Company and any prospective purchaser.
- 5.5 <u>No Withdrawal of Capital Contributions.</u> Except upon a dissolution and liquidation of the Company effected in accordance with <u>Article X</u> and <u>XI</u>, no Member shall have the right to withdraw its Capital Contributions from the Company.
- 5.6 No Other Capital Contributions. Except pursuant to Section 5.1(e)(i), no Member shall be obligated to make any cash or non-cash contribution to the Company's capital. Except with the approval of the Board and subject to compliance with Section 4.10 and Section 5.4, and except as pursuant to Section 5.1(e)(i), no Member shall be permitted to make any cash or non-cash contribution to the Company's capital.

## 5.7 <u>Capital Accounts.</u>

(a) The Company shall establish and maintain a separate capital account (each a "Capital Account") for each Member in accordance with Regulations Section 1.704-1(b)(2)(iv) and this Section 5.7(a) shall be interpreted and applied in a manner consistent with such regulations. No Member shall have any obligation to restore any portion of any deficit balance in such Member's Capital Account, whether upon liquidation of its interest in the Company, liquidation of the Company or otherwise. From and after the date of this Agreement, each Member's Capital Account shall be credited with (i) the amount of any money contributed by such Member to the capital of the Company, (ii) the Fair Market Value of any property contributed (or deemed contributed, whether by the terms of this Agreement or upon a determination by the Board) by such Member, as determined by the Board and the contributing Member at arm's length at the time of contribution (net of liabilities assumed by the Company or subject to which the Company takes such property), (iii) such Member's share of Net Profits (or items of income or gain) and (iv) the amount of any Company liabilities that are considered assumed by such Member or which are secured by any property distributed to such Member. Each Member's Capital Account shall be decreased by (x) the amount of any money actually distributed (or deemed distributed, including pursuant to Section 6.4), (y) the Fair Market Value of any property distributed to such Member at the time of distribution and (z) such Member's share of Net Loss (or items of loss, expense or deduction).

(b) If any Units or other interest in the Company (or portion thereof) is transferred in accordance with the terms of this Agreement, the transferee shall succeed to the Capital Account of the transferor to the extent such Capital Account is attributable to Transferred Units or other interest in the Company (or portion thereof).

## ARTICLE VI ALLOCATIONS AND DISTRIBUTIONS

#### 6.1 Allocations.

(a) Allocations to Capital Accounts. After giving effect to the allocations set forth in Section 6.1(b), the Net Profit and Net Loss of the Company, including each item of income, gain, loss, credit and deduction, for each Fiscal Year or other applicable period, shall be allocated among the Members in a manner such that the Capital Account of each Member, immediately after giving effect to such allocation, is, as nearly as possible, equal (proportionately) to (1) the amount of the distributions that would be made to such Member during such Fiscal Year or other applicable period if: (i) the Company were dissolved and terminated; (ii) its affairs were wound up and each Company asset was sold for cash equal to its Book Value (except that any Company asset that is sold in such Fiscal Year shall be treated as if sold for an amount of cash equal to the sum of (A) the amount of any net cash proceeds actually received by the Company in connection with such disposition and (B) the Book Value of any property actually received by the Company in connection with such disposition); (iii) all Company liabilities were satisfied (limited with respect to each nonrecourse liability to the Book Value of the assets securing such liability); (iv) all Unvested Class B Units became Vested Class B Units; and (v) the net assets of the Company were distributed in accordance with Section 11.1 to the Members immediately after giving effect to such allocation minus (2) such Member's share of Company Minimum Gain and Member Minimum Gain, computed immediately prior to the hypothetical sale of assets, and the amount any such Member is treated as obligated to contribute to the Company, computed immediately after the hypothetical sale of assets.

(b) The following special allocations shall be made in the following order prior to any allocations under Section 6.1(a):

(i) Company Minimum Gain Chargeback. Except as otherwise provided in Regulations Section 1.704-2(f), notwithstanding the provisions of Section 4.1, if there is a net decrease in Company Minimum Gain during any Fiscal Year, each Member shall be specially allocated items of Company income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Member's share of the net decrease in Company Minimum Gain, determined in accordance with Regulations Section 1.704-2(g). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with Regulations Sections 1.704-2(f)(6) and 1.704-2(j)(2). This Section 6.1(b)(i) is intended to comply with the partnership minimum gain chargeback requirement in Regulations Section 1.704-2(f) and shall be interpreted and applied consistently therewith.

- (ii) Member Minimum Gain Chargeback. Except as otherwise provided in Regulations Section 1.704-2(i)(4), notwithstanding any other provision of this Article VI, if there is a net decrease in Member Minimum Gain attributable to a Member Nonrecourse Debt during any Fiscal Year, any Member with a share of that Member Minimum Gain attributable to such a Member Nonrecourse Debt (as determined under Regulations Section 1.704-2(i)(5)) as of the beginning of the year shall be allocated items of Company income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Member's share of the net decrease in Member Minimum Gain, determined in accordance with Regulations Section 1.704-2(i)(4). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with Regulations Sections 1.704-2(f)(6) and 1.704-2(j)(2). This Section 6.1(b)(ii) is intended to comply with the partner minimum gain chargeback requirements in the Regulations and shall be interpreted and applied consistently therewith.
- (iii) Qualified Income Offset. In the event any Member unexpectedly receives any adjustments, allocations, or distributions described in Regulations Sections 1.704-1(b)(2)(ii)(d)(4), (5) or (6), items of Company income and gain shall be specially allocated to such Member in an amount and manner sufficient to eliminate, to the extent required by the Regulations, the Adjusted Capital Account Deficit as quickly as possible; provided, however, that an allocation pursuant to this Section 6.1(b)(iii) shall be made only if and to the extent that such Member would have an Adjusted Capital Account Deficit after all other allocations provided in this Article VI have been tentatively made as if this Section 6.1(b)(iii) were not in this Agreement. This Section 6.1(b)(iii) is intended to comply with the qualified income offset provision in Regulations Section 1.704-1(b)(2)(ii)(d) and shall be interpreted and applied consistently therewith.
- (iv) Gross Income Allocation. In the event any Member has a deficit Capital Account at the end of any Fiscal Year that is in excess of the amount that such Member is deemed to be obligated to restore pursuant to the penultimate sentences of Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5), each such Member shall be specially allocated items of Company income and gain in an amount and manner sufficient to eliminate such deficit as quickly as possible; provided, however, that an allocation pursuant to this Section 6.1(b)(iv) shall be made only if and to the extent that such Member would have a deficit Capital Account in excess of such sum after all other allocations provided in this Article VI have been tentatively made as if Section 6.1(b)(iii) and this Section 6.1(b)(iv) were not in this Agreement.
- (v) Nonrecourse Deductions. Nonrecourse Deductions for any Fiscal Year shall be specially allocated to the Members in proportion to their ownership of Class A Units and each Member's share of excess Nonrecourse Debt shall be allocated in the same manner.
- (vi) Member Nonrecourse Deductions. Member Nonrecourse Deductions for any Fiscal Year shall be specially allocated to the Member who bears the economic risk of loss with respect to the Member Nonrecourse Debt to which such Member Nonrecourse Deductions are attributable in accordance with Regulations Section 1.752-3. If more than one Member bears the economic risk of loss for a Member Nonrecourse Debt, any Member Nonrecourse Deductions attributable to that Member Nonrecourse Debt shall be allocated among the Members according to the ratio in which they bear the economic risk of loss.

(vii) Limitation on Allocation of Net Loss. To the extent that any allocation of Net Loss (or items of loss or deduction) would cause or increase an Adjusted Capital Account Deficit as to any Member, such allocation of Net Loss (or items of loss or deduction) shall be reallocated to the holders of Class A Units in accordance with their respective Percentage Interests as determined by the Company, subject to the limitations of this Section 6.1(b)(vii).

(viii) Code Section 754 Adjustments. To the extent an adjustment to the adjusted tax basis of any Company asset pursuant to Code Sections 734(b) or 743(b) is required following an election pursuant to Section 7.3(a), pursuant to Regulations Sections 1.704-1(b)(2)(iv)(m)(2) or (4), to be taken into account in determining Capital Accounts as a result of a distribution to a Member in complete liquidation of its interest in the Company, the amount of such adjustment to the Capital Accounts shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases such basis), and such gain or loss shall be specially allocated to the Members in a manner consistent with the manner in which their Capital Accounts are required to be adjusted pursuant to such Sections of Regulations.

(c) Construction. If the Board determines that it is necessary or appropriate to modify the manner in which the balances of the Capital Accounts are computed or the items of Net Profits and Net Losses are determined in order to comply with the requirements of Section 704(b) of the Code and the Regulations promulgated thereunder, the Board may make such modification.

#### 6.2 Tax Allocations.

- (a) Items of Company income, gain, loss and deduction shall be allocated, for United States federal, state, and local income tax purposes, among the Members in the same manner as the items of income, gain, loss and deduction were allocated pursuant to the other sections of this Article VI; provided, however, that notwithstanding any other provision of this Agreement, the Company shall cause income, gain, loss and deductions (including any depreciation and/or cost recovery deductions) attributable to property contributed by a Member or revalued by the Company to be allocated among the Members for income tax purposes in accordance with Code Section 704(c) and Regulations promulgated thereunder using the traditional allocation method described in Treasury Regulations Section 1.704-3(b). Subject to the preceding sentence, the Board shall make such allocations for tax purposes, including allocations required under Code Section 704(c) and Regulations promulgated thereunder, as it determines in its reasonable discretion to be appropriate to ensure allocations are made in accordance with a Member's interest in the Company.
- (b) Allocations pursuant to this <u>Section 6.2</u> are solely for purposes of United States federal, state, and local tax purposes and shall not affect, or in any way be taken into account in computing, any Member's Capital Account or share of items of income, gain, loss and deduction.

(c)	The Members are aware of the tax	x consequences of the allocation	ns made by this <u>Section 6.2</u> and agree
to be bound by the provisions of thi	s Section 6.2 in reporting their sha	ares of items of Company incor	ne, gain, loss and deduction.

- (d) In the event there is any change of a Member's interest in the Company during any Fiscal Year, Net Profit and Net Loss of the Company and other items of income, gain, loss and deduction of the Company attributable to such Member's interest for such Fiscal Year shall be determined using the interim closing of the books method.
- 6.3 No Right to Distributions. No Member shall have the right to demand or receive Distributions of any amount, except as expressly provided in this Article VI.
- 6.4 <u>Distributions</u>. Subject to <u>Section 4.10</u> and <u>Section 6.6</u>, the Company shall make distributions to the Members at the discretion of the Board or promptly following the receipt of proceeds from a Liquidity Event in which the Company receives proceeds of such Liquidity Event (and upon the occurrence of a Liquidity Event, all amounts received by the Company from such Liquidity Event reduced by existing liabilities and expenses of the Company and a reasonable reserve established by the Board for future liabilities and expenses), such amount to be distributed or paid in the following order and priority:
- (a) <u>first</u>, to the Class A Holders in proportion to their Unreturned Capital in respect of such Class A Units, until such time as they have received cumulative Distributions under this <u>Section 6.4(a)</u> equal to all of their Capital Contributions, and no Distribution shall be made pursuant to paragraphs (b), (c) or (d) below until the aggregate Unreturned Capital of all Class A Units is equal to zero;
- (b) <u>second</u>, to the Class A Holders in proportion to their unpaid Priority Returns until such time as they have received cumulative Distributions under this <u>Section 6.4(b)</u> equal to their Priority Return, and no Distribution shall be made pursuant to paragraphs (c) or (d) below until the full amount of all Priority Returns have been distributed to all Class A Holders;
- (c) <u>third</u>, except as provided below in <u>Section 6.5</u>, to Management Holdco in respect of its Class B Units until such time as it has received in respect of such Class B Units an amount equal to (i) the Class B Percentage multiplied by (ii) the sum of (1) all Distributions to the Class A Holders pursuant to Section 6.4(b) and (2) all Distributions pursuant to this Section 6.4(c); and
- (d) <u>thereafter</u>, except as provided below in <u>Section 6.5</u>, to the Members *pro rata* in accordance with their respective Percentage Interests.

In connection with any Liquidity Event pursuant to which all or some of the Units are sold or transferred directly to one or more purchasers, any proceeds payable to the Persons that were Members as of immediately prior to the consummation of such Liquidity Event shall be allocated and paid to the Members in accordance with the order and priority set forth in this <u>Section 6.4</u>.

#### 6.5 Distributions to Holders of Class B Units.

- (a) Notwithstanding anything to the contrary set forth in this Agreement, each holder of Class B Units shall be entitled to participate in Distributions pursuant to Section 6.4 only after cumulative Distributions to Class A Units and Class B Units with a lower Distribution Hurdle pursuant to this Agreement exceed the Distribution Hurdle applicable to such Class B Unit (each Class B Unit that has become entitled to participate in Distributions in accordance with this Section 6.5(a), a "Participating Class B Unit").
- (b) Notwithstanding anything to the contrary set forth in this Agreement, a Member holding an Unvested Class B Unit shall only be entitled to receive a Distribution in respect of such Unvested Class B Unit in an amount equal to the Tax Distributions (if any) made with respect to such Unvested Class B Unit, in accordance with Section 6.6. The amount of any Distributions that, but for the preceding sentence, would be made in respect of Unvested Class B Units pursuant to Section 6.4 shall instead be made to the Members holding Class A Units and Vested Class B Units that are Participating Class B Units in accordance with their respective Percentage Interests. Following the vesting of any Unvested Class B Unit (such that it becomes a Vested Class B Unit), the Management Holdco Class B Holder holding such Vested Class B Unit shall thereafter be entitled to receive a "catch up distribution" from any funds being distributed under Section 6.4, at the time such funds are distributed and before any other amounts are distributed pursuant to Section 6.4, of the amount such Management Holdco Class B Holder would have received if such Vested Class B Units were vested on their date of grant.
- (c) The Members acknowledge that Class B Units that are not Participating Class B Units may become Participating Class B Units as a result of distributions to Class A Holders under Section 6.4(b). Such Class B Units shall only be entitled to receive distributions under Section 6.4(c) with respect to the distributions to the Class A Holders under Section 6.4(b) after such time as such Class B Units became Participating Class B Units. In addition, no distributions will be made under Section 6.4(c) with respect to any Unvested Class B Units which became Vested Class B Units as a result of distributions under Section 6.4(c). Any distributions not distributable to Class B Units pursuant to this Section 6.5(c) shall not otherwise be distributed with respect to Class B Units under Section 6.4(c).
- Tax Distributions. Notwithstanding any other provision herein to the contrary, so long as the Company 6.6 is treated as a partnership for federal income tax purposes, the Company shall, subject to having cash available after taking into account reserves as reasonably determined in the good-faith discretion of the Board and subject to Section 6.8, make quarterly advance cash distributions to each Member in an amount equal to the Member's Quarterly Estimated Tax Amount for such quarter ("Tax Distributions"), and the Board shall use reasonable best efforts to cause such Tax Distributions to be made at least five (5) days before estimated U.S. federal income tax payments for individuals are due. If, on the date of a quarterly estimated distribution, the cash available to the Company (as determined in the good-faith discretion of the Board) is not sufficient to distribute to each Member the full amount of such Member's Tax Distribution that would otherwise be required under this Section 6.6, then (A) distributions shall be made by the Company to the Members pursuant to this Section 6.6 to the extent of the cash so available in proportion to the amounts that would have been distributed to each Member pursuant to this Section 6.6 if there had been a sufficient amount of cash available to the Company to make such Tax Distribution in full, (B) the Company shall make future distributions as soon as reasonably practicable following the date on which there exists an amount of cash available to the Company after taking into account reserves as determined in the goodfaith discretion of the Board sufficient to pay the remaining portion of such Member's required Tax Distribution and (C) the Company shall not, until the remaining portion of each such Member's required Tax Distribution is so distributed, make a Distribution other than pursuant to this sentence. The Company shall make commercially reasonable efforts to ensure that any financing documents allow for full Tax Distributions. Each Tax Distribution made to a Member pursuant to this Section 6.6 shall be treated as an advance against, and thus reduce, the amount of the next succeeding Distribution or Distributions which would otherwise have been paid to such Member pursuant to Section 6.4 or Section 11.1, if applicable.

Withholding. The Company at all times shall be entitled to make payments with respect to each Member 6.7 in amounts required to discharge any obligation of the Company to withhold or make payments to any U.S. federal, state, local or foreign taxing authority with respect to any distribution or allocation of income or gain or otherwise with respect to such Member and to withhold (or deduct) the same from distributions to such Member. Any funds withheld from a distribution or paid in respect of a Member (including related tax credits or benefits to which such Member may be entitled as a result of any payment by the Company to any taxing authority and the portion of any imputed underpayment (as determined under Code Section 6225) attributable to such Member) by reason of this Section 6.7 shall be treated as actually distributed to such Member for all purposes of this Agreement. If the Company makes any payment to a taxing authority in respect of a Member (including the portion of any imputed underpayment (as determined under Code Section 6225) attributable to such Member) that is not withheld from actual distributions to the Member, then the Member shall indemnify and reimburse the Company for the amount of such payment (including any interest, penalties and related expenses), on demand. The amount of a Member's indemnification and reimbursement obligations under this Section 6.7, to the extent not paid, shall bear interest at a rate of the Prime Rate as of such due date, and shall be deducted from the distributions to such Member; any amounts so deducted shall constitute a repayment of such Member's obligation hereunder. Each Member's indemnification and reimbursement obligations under this Section 6.7 shall survive a transfer by such Member of its interest in the Company, a withdrawal by such Member and a winding up or dissolution of the Company. Each Member agrees to furnish the Company with any representations, forms, documentation and other information reasonably requested by the Company to assist it in determining the extent of, and in fulfilling, any withholding obligations it may have. Each Member agrees to indemnify and hold harmless the Company and the other Members from and against any liability with respect to taxes, interest, penalties and related expenses that may be asserted by reason of the failure to deduct and withhold tax on amounts distributable or allocable to such Member. Any amount payable as indemnity hereunder by a Member will be paid promptly to the Company or such other Member, as applicable, and if not so paid, the Company will be entitled to retain or otherwise offset any distributions due to such Member for all such amounts. Any income taxes, penalties and interest payable by the Company (including any entity-level tax incurred by the Company and creditable against an individual's personal tax liability, including, but not limited to, state taxes described in Internal Revenue Service Notice 2020-75) shall be treated as specifically attributable to the Members and shall be allocated among the Members such that the burden of (or any diminution in distributable proceeds resulting from) any such taxes, penalties or interest is borne by those Members to whom such amounts are specifically attributable (whether as a result of their status, actions, inactions or otherwise), in each case as reasonably determined by the Board, which shall take into account for this purpose any special tax treatment or status of a Member or its indirect owners.

- 6.8 Restrictions on Distributions. Notwithstanding anything to the contrary provided herein, no Distribution or Tax Distribution shall be made if, and for so long as, such Distribution or Tax Distribution would violate (x) any Law then applicable to the Company or (y) any contractual covenant or agreement between the Company or any of its Subsidiaries and any third-party lender or other *bona fide* third-party source of financing.
- 6.9 <u>Determinations by the Board</u>. Notwithstanding anything to the contrary in this Agreement (other than Section 6.2), in the event the Board determines that it would be prudent to modify the manner in which the Capital Accounts and the allocation of net income or loss (including items thereof, and other tax items) are computed in order to effectuate the intended economic sharing arrangement of the Members or to comply with Section 704 and the Regulations thereunder, the Board may make such modification; provided that, to the extent such a modification has a material disproportionate economic impact on an Investor, the Board shall provide reasonable advance written notice to such impacted Investor of the intended modification (together with any written analysis and information available substantiating such modification) and provide such Investor with a reasonable opportunity to comment on such modification.

## ARTICLE VII ACCOUNTS

7.1 <u>Books.</u> The Board shall cause to be maintained complete and accurate books of account of the Company's affairs at the Company's principal place of business. Such books shall be kept on such method of accounting as the Board shall select. For purposes of the Company's financial statements, the Company's assets and liabilities and statements of operations and cash flows shall be prepared in conformity with GAAP. The Company's accounting period shall be as determined by the Board. The books of account of the Company shall be closed after the close of each Fiscal Year.

## 7.2 <u>Information Rights</u>.

- (a) For so long as a Member owns any Class A Units, the Company shall afford such Member and its Representatives access during normal business hours to (i) the Company's and its Subsidiaries' properties, (ii) the corporate, financial and similar records, reports and documents of the Company and its Subsidiaries, including all books and records and (iii) any officers and senior employees of the Company or its Subsidiaries and provide such Member and its Representatives the opportunity to discuss the affairs, finances and accounts of the Company with such officers and senior employees, in each case, (y) for the proper business purposes related to the Company and its Subsidiaries or in order to comply with applicable Law and (z) in such manner as to not interfere with the normal business operations of the Company and its Subsidiaries.
- (b) For so long as a Member owns any Class A Units, the Company shall deliver or cause to be delivered to such Member:

- (i) within one hundred and twenty (120) days after the end of each Fiscal Year (or such longer period of time as is approved by the Board), an audited consolidated balance sheet of the Company and its Subsidiaries as of the end of such Fiscal Year and the related consolidated income statement, consolidated statement of Members' equity and consolidated statement of cash flows including all footnotes thereto for such Fiscal Year prepared in accordance with GAAP, consistently applied; and
- (ii) within sixty (60) days after the end of each quarterly accounting period, an unaudited consolidated balance sheet of the Company and its Subsidiaries as of the end of such quarterly accounting period and an unaudited related consolidated income statement, consolidated statement of Members' equity and consolidated statement of cash flows for such quarterly accounting period prepared in accordance with GAAP (excluding footnotes and normal year-end adjustments), consistently applied.
- (c) Each applicable Member shall (i) bear all expenses incurred in any examination requested by such Member and (ii) obtain written confidentiality agreements in customary form from any agent, employee or certified public accountant auditing or examining the books of account of the Company or its Subsidiaries pursuant to this <u>Section 7.2</u>.
- (d) Notwithstanding anything to the contrary in this Section 7.2, the Company shall not be obligated to provide any Member with access to any information to the extent the Board reasonably believes that disclosure would violate applicable Law, or impair attorney-client or other legal privilege; provided that, the Company and the requesting Member shall cooperate in good faith to permit disclosure of such information in a manner that does not result in such a violation, breach or impairment.

#### 7.3 Tax Matters.

The Company shall arrange for the preparation of all income and other Tax Returns of the Company and shall cause the same to be filed in a timely manner. For each Fiscal Year for which the Company is treated as a partnership for U.S. federal income tax purposes, the Company shall use reasonable best efforts to furnish to each Member, an IRS Schedule K-1 (or any successor form and any corresponding form and schedule required to be issued under state or local income tax law) with respect to such Member no later than March 15 following each tax year (other than the first tax year of the Company) and shall furnish such IRS Schedule K-1(or any successor form and any corresponding form and schedule required to be issued under state or local income tax law) to such Member no later than March 31 following each tax year; provided, that with respect to any Fiscal Year in which the Company disposes of an interest in any of its Subsidiaries, the Company shall make commercially reasonable efforts to furnish an IRS Schedule K-1 (or any successor form and any corresponding form and schedule required to be issued under state or local income tax law) with respect to each Member no later than March 31 following each tax year and otherwise shall do so as soon thereafter as is reasonably practicable. In addition, the Company shall use reasonable best efforts to provide, to the extent reasonably available, such other information as a Member may reasonably request for purposes of complying with tax reporting requirements applicable to such Member. Except as otherwise provided in Section 4.10 and Section 9.2(c), at the discretion of the Board, the Company shall be entitled to make any tax elections allowed under the Code or the tax laws of any state or other jurisdiction having taxing jurisdiction over the Company and monitor any governmental tax authority in any audit that such authority may conduct of the Company's books and records or other documents. The Company shall, if requested by a Member, elect, in accordance with Code Section 754, to adjust basis in the event any interest of the Company is transferred in accordance with this Agreement or any Company property is distributed to any Member.

- (b) At the request of the Company, each Member shall, to the extent it is reasonably able to, provide the Company with such information, representations, waivers, certificates, duly completed forms and other documentation as is reasonably necessary for the Company or its Affiliates to comply with any withholding, reporting or other tax obligation; provided that, notwithstanding anything to the contrary in this Agreement, no Member shall be required to provide information with respect to any direct or indirect owner of such Member, except to the extent that such information is reasonably available to the applicable Member, is not subject to any non-disclosure restrictions imposed by contractual obligations or policies of the applicable Member, and such information is necessary for the Company and/or its Subsidiaries to comply with any applicable tax filing obligations (including determining whether any Subsidiary of the Company has had an "ownership change" within the meaning of Section 382 of the Code and the Treasury Regulations issued thereunder).
- (c) The Members intend that the Company shall be treated as a partnership for federal, state and local income and franchise tax purposes, and the Company shall not elect, and the Board shall not permit the Company to elect, to be treated as an association taxable as a corporation for federal, state or local income tax purposes under Regulations Section 301.7701-3 or under any corresponding provision of state or local law. Each Member and the Company shall file all Tax Returns consistent with such treatment. This characterization, solely for such tax purposes, does not create or imply a general partnership among the Members for any other purpose.
- (d) BridgeBio shall act as the "partnership representative" as defined in Section 6223 of the Code (or corresponding provisions of state or local law) (the "Partnership Representative") in connection with any tax audit or other tax proceeding, and shall appoint an individual to act as a "designated individual" on behalf, and subject to the direction and control of, the Partnership Representative in accordance with Treasury Regulation 301.6223-1. The Partnership Representative shall have the power and authority to perform in such capacity those duties as may be required to be performed by the "partnership representative" under the Code (and in any similar capacity under state or local Law), and to expend Company funds for reasonable out-of-pocket professional services and costs associated therewith. The Partnership Representative shall advise and consult with the Board from time to time regarding the status of tax-related elections, investigations, proceedings and negotiations with Governmental Authorities.
- (e) If the Company becomes the subject of any audit, assessment or other examination relating to taxes by any tax authority or any judicial or administrative proceedings relating to taxes (a "Tax Audit"), each Member agrees to cooperate with the Board and the Partnership Representative and to do or refrain from doing any or all things reasonably required by them to conduct such Tax Audit (including to implement any modifications to any imputed underpayment or similar amount under Section 6225(c) of the Code, any elections under Sections 6221 or 6226 of the Code and any administrative adjustment request under Section 6227 of the Code)). The Partnership Representative shall provide each Member with any information reasonably requested by such Member with respect to such Tax Audit; provided that the Partnership Representative shall not settle any such Tax Audit without a Member's prior written consent if such Member would be materially, disproportionately and adversely impacted by such settlement of such Tax Audit (as reasonably determined by the Board).

Notwithstanding other provisions of this Agreement to the contrary, if any "partnership adjustment" (f) (as defined in the Code) is determined with respect to the Company and the Partnership Representative has not caused the Company to elect pursuant to Section 6226 of the Code to have such adjustment passed through to the Members for the year to which the adjustment relates, then any "imputed underpayment" (as determined in accordance with Section 6225 of the Code) or "partnership adjustment" that does not give rise to an "imputed underpayment" shall be apportioned among the Members of the Company for the taxable year in which the adjustment is finalized in such manner as may be necessary (as determined by the Partnership Representative in good faith after consulting with the Company's accountants, tax return preparers or other advisers, taking into account any differences in the amount of taxes attributable to each Member because of such Member's status, nationality or other characteristics) so that, to the maximum extent possible, the tax and economic consequences of the partnership adjustment or imputed underpayment are borne by the Members in the same manner as they would have been borne had the adjustment or imputed underpayment been properly reflected on the Company's tax returns for the year(s) to which the adjustment relates. In furtherance of the foregoing, in the event the Company is liable for any "imputed underpayment" which is apportioned to a Member under the foregoing rules for the applicable year, such Member shall promptly reimburse the Company for such amount and such reimbursement shall not be considered a Capital Contribution to the Company by such Member. The Partnership Representative shall use commercially reasonable efforts to provide to such Member information allowing such Member to file an amended federal income tax return or to follow an alternative procedure to filing an amended federal income tax return, as described in Section 6225(c)(2)(B) of the Code.

that would cause a Member (or any direct or indirect beneficial owner thereof, as applicable) that is not a United States person within the meaning of Code Section 7701(a)(30) (A) to recognize, solely as a result of its status as a Member (or any direct or indirect owner thereof, as applicable) of the Company, income that is "effectively connected with the conduct of a trade or business within the United States" within the meaning of Code Section 864(c) of the Code, or (B) to be treated as engaged in a "trade or business within the United States" within the meaning of Section 864(b) of the Code, (ii) not to engage in any activity that would cause a Member (or any direct or indirect beneficial owner thereof, as applicable) to be treated as engaged in commercial activities or recognize income that is treated as derived from the conduct of any commercial activity, in each case within the meaning of Section 892 of the Code and (iii) not to engage in any conduct that would cause a Member (or any direct or indirect beneficial owner thereof, as applicable), shall be deemed to have "unrelated business taxable income" as defined in Section 512 of the Code (including pursuant to the application of Section 514 of the Code). The Company shall provide the Members with timely written notice if the Company reasonably believes the Company may be treated as engaged in any activity described in the immediately preceding sentence. The Members acknowledge and agree that the actions and activities of the Company contemplated as of the date of this Agreement are not inconsistent with the foregoing limitations on the activities of the Company.

- (h) At the reasonable written request of any Member, the Company shall use commercially reasonable efforts to promptly provide such Member with any certificate reasonably necessary to mitigate any tax withholding obligations pursuant to a Transfer, including pursuant to Section 1445 or Section 1446 of the Code.
- 7.4 <u>Fiscal Year.</u> The fiscal year of the Company (the "<u>Fiscal Year</u>") for financial statement and federal income tax purposes shall end on December 31 of each calendar year, except as otherwise may be required under Code Section 706 and the Regulations thereunder.
- 7.5 <u>Initial Budget</u>. The Company's budget from the date hereof through the end of calendar year 2025 is set forth on <u>Schedule C</u> (the "<u>Initial Budget</u>").

## ARTICLE VIII TRANSFER OF UNITS IN THE COMPANY

### 8.1 <u>Prohibition.</u>

- (a) Class A Units. Without the prior written consent or approval of a majority of the entire Board, prior to the earlier of (i) five (5) years following the Closing or (ii) a Qualified IPO (the "Lock-Up Period"), each Member and any Assignee or Permitted Transferee shall not, directly or indirectly, Transfer any Class A Units or any right or interest therein, other than Transfers (A) of such Class A Units to a Tag-Along Purchaser pursuant to Section 8.9, (B) of Drag-Along Units to a Drag-Along Purchaser pursuant to Section 8.10, or (C) of such Class A Units to a Permitted Transferee of such Member (or such Assignee or Permitted Transferee) in accordance with this Section 8.1.
- (b) Any attempted Transfer of Units by any Member, other than in strict accordance with this Article VIII, shall be null and void *ab initio* and the purported transferee shall have no rights as a Member or Assignee hereunder, and the Company shall not record such Transfer on its books or treat any purported transferee of such Units as the owner of such units for any purpose. No Member shall avoid the provisions of this Agreement by making one or more Transfers to one or more Permitted Transferees and then disposing of all or any portion of such party's interest in any such Permitted Transferee, and any Transfer or attempted Transfer in violation of this covenant shall be null and void *ab initio*.
- (c) If any Member wishes to Transfer Units to a Permitted Transferee, such Member shall give written notice to the Company of their intention to make such a Transfer not less than ten (10) days prior to effecting such Transfer, which notice shall state the name and address of each Permitted Transferee to whom such Transfer is proposed, the relationship of such Permitted Transferee to such Member and the number of Units proposed to be Transferred to such Permitted Transferee. In connection with any such Transfer, the Permitted Transferee must, prior to any such Transfer, have agreed in writing to be bound by the provisions of this Agreement and have executed and delivered a joinder agreement substantially in the form set forth on Exhibit A (a "Joinder Agreement") and such other documents that may be reasonably requested by the Company. Except as otherwise determined by the Board, it shall be a condition precedent for admittance (and continued admittance) of any natural Person as a party to this Agreement that the spouse of such natural Person execute and deliver to the Company a Spousal Consent, substantially in the form set forth on Exhibit B (a "Spousal Consent").
- (d) Class B Units. Without the prior written consent or approval of a majority of the entire Board, during the Lock-Up Period, each Member and any Assignee or Permitted Transferee shall not, directly or indirectly, Transfer any Class B Units or any right or interest therein, other than Transfers (i) of Drag-Along Units to a Drag-Along Purchaser pursuant to Section 8.10 or (ii) of such Class B Units to a Permitted Transferee of such Member (or such Assignee or Permitted Transferee) in accordance with this Section 8.1. Management Holdco shall not cause or permit any Transfer of any Class B Units or Corresponding Management Holdco Units, except for Transfers of Class B Units or Corresponding Management Holdco Units to the extent permitted by the terms of this Agreement, of the Management Holdco Operating Agreement (to the extent consistent with the terms of this of this Agreement) or the Applicable Class B Unit Agreement, including pursuant to a forfeiture, cancellation or repurchase provision of the Applicable Class B Unit Agreement.

- (e) Permitted Transfer. Any Transfer permitted under this <u>Section 8.1</u> is referred to herein as a "<u>Permitted Transfer</u>".
- (f) Qualified IPO. This <u>Section 8.1</u> shall terminate upon the consummation of a Qualified IPO; <u>provided</u> that, any subsequent lock-up period, if applicable, will be determined in accordance with <u>Section 9.3(b)</u>.

#### 8.2 Conditions to Permitted Transfers.

- (a) Without limiting the restrictions on Transfers and other terms of <u>Section 8.1</u>, a Member shall be entitled to make a Permitted Transfer only upon satisfaction of each of the following conditions, unless waived by the Board:
- (i) such Transfer does not require the registration or qualification of such Units pursuant to any applicable federal or state securities Laws;
  - (ii) such Transfer does not result in a violation of applicable Laws;
- (iii) such Transfer would not cause the Company to become, with respect to any employee benefit plan subject to Title I of ERISA, a "party-in-interest" (as defined in ERISA Section 3(14)) or a "disqualified person" (as defined in Code Section 4975(c));
- (iv) such Transfer would not, in the opinion of legal counsel to the Company, cause any portion of the assets of the Company to constitute assets of any employee benefit plan pursuant to Department of Labor Regulations Section 2510.2-101;
- (v) such Transfer is in compliance with, and does not cause the Company to lose, its status as a partnership for federal, and, where applicable, state and local, income tax purposes;

traded partnership" that would b	(vii) e subject t	such Transfer does not create a material risk that the Company would become a "publicly o tax as a corporation for U.S. federal income tax purposes;
Exchange Act; and	(viii)	such Transfer does not cause the Company to become a reporting company under the
Act of 1940 or the Investment A	(ix) dvisers Ac	such Transfer does not subject the Company to regulation under the Investment Company et of 1940.
to a Company Offeror that is for IPO, (ii) any Transfer in a Qualifin a Qualified IPO or that other Transaction, (iv) any Transfer of Section 8.9, (vi) any Transfer of	following ormed for the field IPO of the wise acts of Units pure Units to a	thstanding anything in <u>Section 8.2(a)</u> to the contrary, none of the conditions set forth in types of Transfers by Members: (i) any Transfer of Units or other securities of the Company he purpose of engaging in a Qualified IPO or that otherwise acts as the issuer in a Qualified f Units or other securities of any Company Offeror that is formed for the purpose of engaging as the issuer in a Qualified IPO, (iii) any Transfer of Units in connection with a Conversion resuant to <u>Section 8.8</u> , (v) any Transfer of Class A Units to a Tag-Along Purchaser pursuant to <u>Drag-Along Purchaser</u> pursuant to <u>Section 8.10</u> or (vii) any Transfer of Class B Units pursuant provision of the Applicable Class B Unit Agreement.
8.3 <u>Effec</u>	ct of Trans	fers.
loss, deduction, credit or simila	nee of the	any Transfer effected in compliance with this <u>Article VIII</u> (other than Permitted Transfers to a Transferred Units shall be entitled to receive the Distributions and allocations of income, gain, which the transferring Member would be entitled with respect to such Units, but shall not be of a Member with respect to the transferring Member's Units, including the right to vote, unless

and until such Assignee is admitted to the Company as a Substitute Member pursuant to Section 8.5.

such Permitted Transferee's compliance with Section 8.1(c), Section 8.2 and Section 8.5.

this Agreement only if and when each of the following conditions is satisfied:

(b)

(a)

(b)

the Company (if any);

such Transfer is not made to any Person who lacks the legal right, power or capacity to own

Upon any Permitted Transfer to a Permitted Transferee, the Assignee of the Transferred Units shall

the Board determines the nature and amount of the Unit Consideration to be paid by such Person to

Admission of Additional Members. A Person shall become an Additional Member pursuant to the terms of

(vi)

Units;

the Board approves such Person's admission;

be entitled to all rights of the transferor with respect to the Transferred Units and shall become a Substitute Member hereunder subject to

- (c) the Company has received such Person's Unit Consideration payable to the Company (if any); and
- (d) the Company receives written instruments, including a Joinder Agreement and, to the extent applicable, a Spousal Consent, that are in substantially the forms attached as Exhibits hereto.
- 8.5 <u>Admission of Assignees as Substitute Members.</u> Except as otherwise provided in <u>Section 8.3(b)</u>, an Assignee of all or any portion of the Units of a Member shall become a Substitute Member of the Company only if and when each of the following conditions is satisfied:
- (a) such Assignee is (i) a Permitted Transferee, (ii) the acquiror in an Unrelated Party Transfer effected in accordance with Section 8.7 or (iii) the Board approves such Assignee's admission; and
- (b) the Company receives a Joinder Agreement and, to the extent applicable, a Spousal Consent, that are in substantially the forms attached as Exhibits hereto.

#### 8.6 Withdrawal of Members upon Transfer.

- (a) If a Member has Transferred all of its Units in one or more Permitted Transfers or otherwise in compliance with this Agreement, then such Member shall withdraw from the Company on the date upon which each Assignee of such Units has been admitted as a Substitute Member in accordance with Section 8.3(b) or Section 8.5, and such Member shall no longer be entitled to exercise any rights or powers of a Member under this Agreement; provided (i) that no cessation of any Person as a Member shall (A) relieve a Person that is not a natural person from their obligation hereunder to reacquire any Units from a Permitted Transferee that ceases to qualify as such hereunder (and if such Person ceases to hold Units due to transfers to one or more Permitted Transferees such Person shall also continue to be subject to the Transfer restrictions in this Article VIII) or (B) affect the obligation of such Member to fund such Member's Additional Cash Contributions (if applicable); and (ii) all rights of such Person in Article XII shall survive any termination of such Person's membership in the Company.
  - (b) No Member shall have the right to withdraw from the Company other than pursuant to Section 8.6(a).

## 8.7 <u>Right of First Offer.</u>

(a) Prior to a Qualified IPO or Change of Control (and subject to the expiration of the Lock-Up Period, if applicable), if a Member holding Class A Units or any of its Assignees or Permitted Transferees (the "<u>Transferring Member</u>") proposes to Transfer, directly or indirectly, any Class A Units or any right or interest therein (the "<u>Offered Units</u>") to any Person other than the Company or a Permitted Transferee of the Transferring Member (an "<u>Unrelated Party Transfer</u>"), then the Transferring Member shall deliver to the other Members holding Class A Units (each a "<u>Non-Transferring Member</u>") a written notice (a "<u>ROFO Sale Notice</u>") setting forth the number of Offered Units proposed to be Transferred and the other material terms and conditions of the proposed Transfer that would reasonably be expected to have a material impact on the value of the Transfer (collectively, the "<u>ROFO Terms</u>").

- Unrelated Party Transfer by a Transferring Member; provided that each Non-Transferring Member shall have the right to purchase such Non-Transferring Member's ROFO Portion of the Offered Units. A Non-Transferring Member's "ROFO Portion" is an amount obtained by multiplying (i) the number of Offered Units by (ii) a fraction, the number of Class A Units then held by such Non-Transferring Member and the denominator of which is the number of Class A Units then issued and outstanding. Each time the Transferring Member proposes to Transfer any Offered Units in an Unrelated Party Transfer, the Transferring Member shall first make an offering of the Offered Units to the Non-Transferring Members in accordance with this Section 8.7. Each Non-Transferring Members may exercise its right of first offer within sixty (60) days following their receipt of a ROFO Sale Notice (such period, the "ROFO Election Period") by delivering to the Transferring Member a written offer to purchase its ROFO Portion of the Offered Units (the "ROFO Offer"), which ROFO Offer sets forth the proposed purchase price and all other material terms and conditions pursuant to which such Non-Transferring Member is willing to purchase the Offered Units. Any ROFO Offer shall be binding and irrevocable for the duration of the applicable ROFO Acceptance Period.
- (c) If any Non-Transferring Member delivers a ROFO Offer pursuant to and in accordance with Section 8.7(b) within the ROFO Election Period, then the Transferring Member may (but shall have no obligation to) accept or reject, for any or no reason, such ROFO Offer during the thirty (30) day period after receipt by the Transferring Member of a ROFO Offer from such Non-Transferring Member (the "ROFO Acceptance Period"); provided that, if the Transferring Member does not accept, or fails to respond to, any ROFO Offer within the ROFO Acceptance Period, then it shall be deemed to have rejected such ROFO Offer. If the Transferring Member accepts any ROFO Offer, the Transferring Member shall notify the applicable Non-Transferring Members of such acceptance in a written notice (a "ROFO Acceptance Notice"). Following the delivery of a ROFO Acceptance Notice, the Transferring Member and the applicable Non-Transferring Members shall negotiate in good faith to complete the sale, purchase and Transfer of (and payment for) the relevant Offered Units at the same price and on the same other terms and conditions of Transfer as set forth in the applicable ROFO Offer within ninety (90) days after the delivery of the ROFO Acceptance Notice.
- (d) If any Non-Transferring Member declines or does not exercise its right of first offer pursuant to Section 8.7(b), the remaining Non-Transferring Members shall have the right to purchase such Class A Units that could have been purchased by such declining Non-Transferring Member; provided that if the remaining Non-Transferring Members have oversubscribed for such Class A Units, then such Class A Units will be allocated on a pro rata basis as between such remaining Non-Transferring Members based on their respective Class A Units holdings). The notice issuances and corresponding timing requirements contemplated by Sections 8.7(a)-(c) shall apply to this Section 8.7(d) mutatis mutandis, except that the referenced 30-day periods shall be replaced with 10-day periods.
- (e) If, following the occurrence of all procedures in <u>Sections 8.7(a)-(d)</u> (such time, the "<u>ROFO End Period</u>"), any portion of the Offered Units was not offered for purchase by any of the Non-Transferring Members, then the Transferring Member shall be free to effect an Unrelated Party Transfer with respect to such unsubscribed Offered Units so long as such Transfer (i) shall be at a price per Class A Unit that is same as or greater than the price per Class A Unit set forth in the ROFO Offer, (ii) shall otherwise be made upon terms no less favorable in the aggregate to the Transferring Member than those set forth in the ROFO Offer, and (iii) shall comply with the provisions of <u>Section 8.2</u>; <u>provided</u> that, if a definitive written agreement evidencing such Unrelated Party Transfer is not executed within one hundred twenty (120) days following the ROFO End Period, then the Transferring Member may not consummate such sale without again complying with the procedures set forth in this <u>Section 8.7</u>.

## 8.8 <u>Call Rights.</u>

- (a) Call Right. With respect to any Management Holder, during any six (6) month period commencing on the date of effectiveness of Termination of such Management Holder (the last day of such period, the "Call Notice Deadline"), the Company shall have the right, but not the obligation, to purchase from Management Holdco, and Management Holdco shall have the obligation to sell to the Company, all or any portion of the Vested Class B Units or any other Units held by Management Holdco, up to the amount of Corresponding Management Holdco Units held by the applicable Management Holder (all of the Vested Class B Units and any other Units held by Management Holdco which may be purchased by the Company pursuant to this sentence, the "Called Units"). The purchase price (the "Call Price") for any Called Units repurchased in accordance with this Section 8.8(a) shall be equal to the Fair Market Value of such Called Units on the date the Call Notice is delivered; provided, however, that in the event that the Termination of the Management Holder is for Cause, the Call Price for any Called Units repurchased in accordance with this Section 8.8 shall be equal to the lesser of (A) the cost paid by the Management Holder to directly or indirectly acquire such Called Units (which, for purposes of Vested Class B Units, shall be deemed to be \$0) less any Distributions previously made with respect to such Called Units and (B) the Fair Market Value (after giving effect to any Losses arising from any conduct constituting Cause) of such Called Units on the date the Call Notice is delivered (the "Call Right"). Management Holdco shall, concurrently with any purchase of Called Units hereunder, repurchase the equivalent number and type of such Units from the Management Holder pursuant to the terms of the Management Holdeo Operating Agreement. The Company and Management Holdco shall reasonably cooperate to consider alternative structures for implementing the exercise of any Call Right in a manner that would minimize any taxes imposed on the Members with respect to the purchase of Called Units and the funding of such purchase; provided, that no such alternative structure shall be implemented without the consent of BridgeBio if such structure could reasonably be expected to have a disproportionate and adverse effect on BridgeBio or any of its Affiliates. Exercise of the Call Right shall not be deemed to limit any other rights or remedies of the Company or any of its Affiliates relating to any conduct constituting Cause.
- (b) *Call Notice*. The Company may exercise its Call Right by delivery of a written notice to Management Holdco and the applicable Management Holder of the Company's election to exercise its Call Right on or prior to the Call Notice Deadline and setting forth the categories and number of Called Units it will purchase (the "Call Notice").

- days following the delivery of the Call Notice at the principal office of the Company or at such other time and location as the parties to such repurchase may mutually determine (the "Management Repurchase Closing"); provided that, if applicable, the deadline for payment by the Company pursuant to this Section 8.8 may be extended as required from time to time by the Company's and its Subsidiaries' debt financing arrangements or for regulatory reasons. At the Management Repurchase Closing, Management Holdco and the applicable Management Holder shall be required to make customary representations and warranties to the Company, including as to Management Holdco and such Management Holder's authority to sell, the enforceability of agreements against such Person, that the Called Units shall be free and clear of any Liens (other than pursuant to U.S. federal and state (and other jurisdiction) securities laws), that it is the sole beneficial and record owner of such Called Units. At the Management Repurchase Closing, the Company shall deliver payment in full in immediately available funds for the Called Units purchased by it. Purchases of any Called Units shall be settled in cash. At the Management Repurchase Closing, all of the parties to the transaction shall execute such additional documents as are otherwise necessary or appropriate.
- (d) Other. Notwithstanding anything set forth in this Section 8.8 to the contrary, the Company shall not purchase any Called Units pursuant to the Call Right until the date that is six (6) months and one (1) day following the date of the issuance or vesting of such Called Units, whichever is later, and in the event that any Called Units subject to the Call Right were issued or vested within the six (6) months immediately preceding the Call Notice Deadline, then the Call Notice Deadline shall become the date that is thirty (30) Business Days following the expiration of such six (6) month period.

## 8.9 <u>Tag-Along Rights.</u>

- (a) General. Subject to Section 8.10 and the other terms of this Agreement, if at any time prior to the consummation of a Qualified IPO or a Change of Control, any Member proposes to Transfer, directly or indirectly, in one or more related transactions, Class A Units then held by such Member ("Tag-Along Seller") representing in excess of one percent (1%) of the Class A Units then issued and outstanding] (the "Offered Tag Units") (a "Tag-Along Sale"), to a Person other than one of such Member's Permitted Transferees (the "Tag-Along Purchaser(s)"), then each other Member holding Class A Units (each a "Tag-Along Rightholder") shall have the right (the "Tag-Along Rightholder"), upon the terms and conditions set forth in this Section 8.9 and Section 8.11, to sell to such Tag-Along Purchaser such Tag-Along Rightholder's Tag-Along Portion of its Class A Units in such Tag-Along Sale. A Tag-Along Rightholder's "Tag-Along Portion" is an amount obtained by multiplying (i) the number of Offered Tag Units in a Tag-Along Sale by (ii) a fraction, the numerator of which is the number of Class A Units then held by such Tag-Along Rightholder and the denominator of which is the number of Class A Units then issued and outstanding.
- Procedure. The Tag-Along Seller shall give written notice (the "Sale Notice") to each Tag-Along Rightholder of any Tag-Along Sale giving rise to the rights pursuant to this Section 8.9 at least twenty (20) Business Days prior to the proposed consummation of such sale, setting forth the material terms of the proposed Tag-Along Sale, including in reasonable detail (to the extent known or available) (A) the number of Offered Tag Units proposed to be sold by the Tag-Along Seller, (B) the maximum number of Class A Units that may be sold by such Tag-Along Rightholder, (C) the identity of the proposed Tag-Along Purchaser, (D) the proposed amount and form of consideration and other material terms and conditions offered by such Tag-Along Purchaser, (E) the proposed date and time of the closing of the Tag-Along Sale, (F) a copy of the form of Transfer agreement proposed to be executed in connection with the Tag-Along Sale and (G) a representation that such Tag-Along Purchaser has been informed of the rights provided for in this <u>Section 8.9</u>. The rights provided by this <u>Section 8.9</u> must be exercised by any Tag-Along Rightholder within fifteen (15) Business Days following receipt of the Sale Notice (the "Tag-Along Period"), by delivery of an irrevocable written notice (the "Tag-Along Election Notice") to the Tag-Along Seller indicating such Tag-Along Rightholder's (each a "Tagging Member") election to exercise its rights, in whole or in part, pursuant to this Section 8.9 and specifying the number of Class A Units to be sold by such Tagging Member in the Tag-Along Sale or indicating that the Tagging Member wants to sell all of the Tagging Member's Class A Units that it is permitted to sell (calculated in accordance with Section 8.9(a)). Any Tag-Along Rightholder may waive its rights under this Section 8.9 with respect to any Tag-Along Sale prior to the expiration of such fifteen (15) Business Day period by giving written notice to the Tag-Along Seller, with a copy to the Company. Each Tag-Along Rightholder who does not timely deliver a Tag-Along Election Notice shall be deemed to have waived its right to participate in such Tag-Along Sale.

- (c) Reduction of Class A Units Transferred. The Tag-Along Seller shall attempt to obtain the inclusion in the proposed Tag-Along Sale of the entire number of Class A Units which each of the Tagging Members shall have requested to be included in the Tag-Along Sale (as evidenced, in the case of the Tag-Along Seller, by the Sale Notice and, in the case of each Tagging Member, by such Tagging Member's Tag-Along Election Notice). In the event that the Tag-Along Seller shall be unable to obtain the inclusion of such entire number of Class A Units in the proposed Tag-Along Sale, the number of Class A Units to be sold in the proposed Tag-Along Sale shall be allocated among the Tag-Along Seller and the Tagging Members in proportion, as nearly as practicable, to the respective number of shares of Class A Units specified for Transfer in accordance with this Section 8.9 by the Tag-Along Seller and each Tagging Member (as evidenced, in the case of the Tag-Along Seller, by the Sale Notice and, in the case of each Tagging Member, by such Tagging Member's Tag-Along Election Notice).
- Tagging Member shall receive the same per unit consideration (including in respect of the amount and form of consideration) for any of its Class A Units that are actually included in the Tag-Along Sale pursuant to this Section 8.9. Each Tagging Member shall otherwise be entitled to participate in such Tag-Along Sale on terms and conditions no less favorable in any material respect than those that apply to the Tag-Along Seller. The consideration to be paid in respect of the Class A Units owned by a Tagging Member that are actually included in such Tag-Along Sale in accordance with this Section 8.9 shall be an amount equal to the aggregate amount that would be distributed to such Tagging Member pursuant to Section 6.4 in respect of such Class A Units in a hypothetical liquidation of the Company, which aggregate liquidation value of the Company shall be based on the aggregate purchase price to be paid in respect of the Class A Units proposed to be sold by the Tag-Along Rightholders in the Tag-Along Sale.
- (e) Consummation of Tag-Along Sale. The Tag-Along Seller shall have sixty (60) days following the expiration of the Tag-Along Period in which to consummate the Tag-Along Sale, on terms not more favorable to the Tag-Along Seller than those set forth in the Sale Notice (which such 60-day period may be extended for a reasonable time, not to exceed ninety (90) days to the extent reasonably necessary to obtain required approvals or consents from any Governmental Authority). If at the end of such period the Tag-Along Seller has not completed the Tag-Along Sale, the Tag-Along Seller may not then effect a Transfer that is subject to this Section 8.9 without again fully complying with the provisions of this Section 8.9.

(f) Scope. The provisions of this Section 8.9 shall not apply to (i) Transfers to a Permitted Transferee of any Member or any director, officer or employee of the Company or any of its Subsidiaries, (ii) any Transfer of Units upon exchange of such Units pursuant to a Conversion Transaction, (iii) any Transfer of Drag-Along Units to a Drag-Along Purchaser pursuant to Section 8.10 or (iv) any Transfer in connection with a Qualified IPO. The provisions of this Section 8.9 are in addition to, and not in limitation of, any other restrictions on Transfers of Units contained in this Agreement.

### 8.10 <u>Drag-Along Rights.</u>

(a) General. Prior to a Qualified IPO or a Change of Control, (x) a Member or Members representing at least a majority of the issued and outstanding Voting Units (which majority must include each of (i) Viking, so long as the Investor Member Sell-Down Event in respect of Viking has not occurred, (ii) Patient Square, so long as the Investor Member Sell-Down Event in respect of Patient Square has not occurred, (iii) BridgeBio, so long as the Investor Member Sell-Down Event in respect of BridgeBio has not occurred and (iv) NK, so long as the Investor Member Sell-Down Event in respect of NK has not occurred), (y) following the fifth (5th) anniversary of the Closing, subject to the Minimum Return Condition being satisfied, a Member or Members representing at least a majority of the issued and outstanding Voting Units or (z) following the seventh (7th) anniversary of the Closing, each of (A) Viking, so long as the Investor Member Sell-Down Event in respect of Viking has not occurred or (B) Patient Square, so long as the Investor Member Sell-Down Event in respect of Patient Square has not occurred (such Persons in prongs (x), (y) and (z), as applicable, the "Drag-Along Sellers"), shall have the right, by delivery of a written notice to the Company to require that the Company pursue and cause a sale of the Company (or one or more of its assets, divisions or Subsidiaries) (a "Drag-Along Sale"), and the Drag-Along Sellers may require each other equityholder of the Company (each a "Dragged Member") to Transfer its Drag-Along Portion of such equityholder's Units to the prospective purchaser (the "Drag-Along Purchaser"). "Minimum Return Condition" shall mean that each Investor has received or will receive its respective Priority Return upon the consummation of the sale of the Company (or one or more of its assets, divisions or Subsidiaries); provided that such Minimum Return Condition may be waived or reduced with written approval from both Viking and Patient Square.

(b) Drag-Along Notice. Drag-Along Sellers shall send written notice (the "Drag-Along Notice") of the exercise of their rights pursuant to this Section 8.10 to each Dragged Member at least thirty (30) days prior to the consummation of the Drag-Along Sale. The Drag-Along Notice shall set forth the material terms and conditions of the Drag-Along Sale, including (to the extent available) (i) the form and amount of consideration to be paid for the Units, (ii) the identity of the Drag-Along Purchaser, (iii) the proposed date and time of the closing of the Drag-Along Sale and (iv) a copy of the form of definitive agreement proposed to be executed in connection with the Drag-Along Sale. If the Drag-Along Sellers consummate the Drag-Along Sale to which reference is made in the Drag-Along Notice, each Dragged Member shall be bound and obligated to Transfer the applicable Drag-Along Portion of its Units (the "Drag-Along Units") in such Drag-Along Sale subject to the terms and conditions set forth in this Section 8.10 and Section 8.11. Any Drag-Along Notice may be rescinded by the Drag-Along Sellers at any time prior to consummation of the Drag-Along Sale by Drag-Along Sellers delivering written notice thereof to each Dragged Member.

- (c) Cooperation. Without limiting Section 8.11(a), in connection with any Drag-Along Sale that is effected in accordance with the terms and conditions set forth in this Section 8.10 and Section 8.11, each Dragged Member hereby agrees: (i) to vote or cause to be voted (whether at a meeting or by written consent in lieu thereof, at the option of the Drag-Along Sellers) all equity securities that the Dragged Member holds or with respect to which the Dragged Member has the power to direct the voting and which the Dragged Member is entitled to vote on such proposed Drag-Along Sale in favor of such proposed Drag-Along Sale; (ii) if the Drag-Along Sale is structured as a merger, consolidation or similar business combination, to waive (and hereby does waive) all appraisal rights, dissenters rights and similar rights (if any) in connection therewith; and (iii) not to make any claim or take any action reasonably likely to hinder, impede, delay or have an adverse effect on such Drag-Along Sale.
- (d) Facilitation. Notwithstanding anything contained in this Section 8.10 to the contrary, the Board shall control the day-to-day management of any Drag-Along Sale and shall (i) pursue such Drag-Along Sale in good faith and use reasonable best efforts to effect such Drag-Along Sale on reasonably acceptable terms and conditions consistent with the Drag-Along Notice; and (ii) consult in good faith with the Drag-Along Sellers with respect to the following matters in connection with such Drag-Along Sale: (A) the number, nature and identity of any potential bidders approached, (B) the selection of any investment banking firm or any other advisors and the terms of engagement with such third parties, (C) drafts of any information memorandums, management presentations, purchase agreements, contracts or other material documents developed or produced in connection with the sale process and (D) the final terms of any such Drag-Along Sale and corresponding documentation.
- (e) Terms and Consideration. Subject to Section 8.11, in connection with any Drag-Along Sale, (i) the Transfer of each Dragged Member's Units shall be on terms and conditions which are not economically less favorable to the Dragged Members in any non-de minimis respect than those applicable to the Drag-Along Sellers and (ii) the Dragged Members shall receive (A) the same per unit consideration as the Drag-Along Sellers with respect to such Dragged Members' Class A Units and (B) the same form or combination of forms of consideration as the consideration proposed to be paid to the Drag-Along Sellers; provided that each Dragged Member will receive in such Drag-Along Sale in respect of its Units the aggregate amount that would be distributed to such Dragged Member in respect of such Units in a hypothetical liquidation of the Company pursuant to Section 6.4, which aggregate liquidation value of the Company shall be based upon the aggregate purchase price to be paid in the Drag-Along Sale. If the Drag-Along Sellers are given an option as to the form and amount of consideration to be received, to the extent reasonably possible the same option will be given to the Dragged Members. When used in this Section 8.10, the phrase "same per unit consideration" shall not include any employment, consulting, change of control or any other consideration payable to the Dragged Members who are Management Holders pursuant to any other arrangements, whether in connection with such Drag-Along Sale or otherwise, and the phrase "same form of consideration" shall not apply to any Units in connection with a "rollover" in such Drag-Along Sale. Each Member acknowledges and agrees that depending upon the aggregate consideration to be distributed in connection with the Drag-Along Sale, certain classes or series of Units may receive less consideration per Unit than other classes or series of Units, and certain classes or series may receive no consideration in the Drag-Along Sale.

- (f) Class B Units. Notwithstanding anything to the contrary herein, in the event of a Drag-Along Sale, the Board may (i) cancel and redeem any Class B Unit or (ii) cause any Class B Unit to be converted into or exchanged for Class A Units or other securities, in each case, at a value equal to the value that would have been distributed to such Class B Unit in such Drag-Along Sale in respect of such Class B Unit in a hypothetical liquidation of the Company pursuant to Section 6.4, which aggregate liquidation value of the Company shall be based upon the aggregate purchase price to be paid in the Drag-Along Sale.
- 8.11 <u>Certain Terms and Conditions Applicable to Drag-Along Sales and Tag-Along Sales.</u> The following provisions shall apply to any proposed Transfer of Units or other transaction to which <u>Section 8.9</u> or <u>Section 8.10</u> applies (each an "<u>Article 8 Transaction</u>"), except to the extent any provision of this <u>Section 8.11</u> by its terms only applies to a specified subset of such Sections:
- (a) Efforts. Each Dragged Member or Tagging Member (each an "Article 8 Seller"), whether in such Person's capacity as a Member, equityholder, officer, director or employee of the Company, shall, to the maximum extent permitted by Law, take all actions customary, necessary or desirable in order to consummate expeditiously any Article 8 Transaction and any related transactions, including executing, acknowledging and delivering any applicable equity purchase agreement, merger agreement, consents, assignments, waivers, releases, letters of transmittal, and other documents or instruments necessary (including a release of claims, furnishing information and copies of documents, filing applications, reports, returns, filings and other documents or instruments with governmental authorities) and otherwise reasonably cooperate with any Drag-Along Seller, Tag-Along Seller, Tag-Along Purchaser or Drag-Along Purchaser, in each case, subject to the applicable terms of Section 8.9 and Section 8.10. Without limiting the generality of the foregoing:
- (i) each Article 8 Seller shall, on a several and not joint basis (in accordance with its *pro rata* portion (determined based on the respective proceeds payable to each Article 8 Seller in connection with such Article 8 Transaction in accordance with Section 6.4) (the "Pro Rata Share")): (A) make customary individual fundamental representations and warranties as to (1) such Article 8 Seller's own organization, (2) such Article 8 Seller's power, authority, capacity and legal right to enter into and consummate the transactions contemplated by any applicable agreement to which such Article 8 Seller becomes party, (3) the unencumbered title to such Article 8 Seller's Units (other than encumbrances arising under this Agreement or pursuant to U.S. federal and state (and any other applicable jurisdiction) securities laws, (4) the absence of any conflict with respect to such Article 8 Seller's Units or the Transfer thereof to the extent such conflict limits its ability to enter into and consummate the transactions contemplated by any applicable agreement to which such Article 8 Seller becomes a party and (5) the binding effect and enforceability of any applicable agreement to which such Article 8 Seller becomes a party; and (B) provide customary indemnification with respect to a breach thereof (taking into account the limitations set forth in Section 8.11(a)(v) other than with respect to preceding prongs (2) and (3));

(ii) subject to Section 8.11(a)(v), each Article 8 Seller shall, on a several and not joint basis (iii
accordance with its Pro Rata Share), be liable (whether by transaction expenses, purchase price adjustment, escrows, indemnity payment
or other similar items) in respect of representations, warranties, covenants and agreements made in respect of the Company and it
Subsidiaries;

- (iii) none of the Article 8 Sellers shall be required to agree to (A) any non-compete, non-solicitation, non-disparagement or other similar restrictive covenants (other than confidentiality obligations that are no more restrictive than the confidentiality obligations applicable to the Tag-Along Seller or the Drag-Along Sellers (as applicable) or (B) any restrictive covenants that bind any portfolio company of any Fund Investor;
- (iv) none of the Article 8 Sellers shall be liable for any individual representation, warranty, covenant or other agreements made by any other Article 8 Seller or the Tag-Along Seller or the Drag-Along Sellers, as applicable; and
- (v) the aggregate liability of each Article 8 Seller shall in no event exceed the proceeds received (taking into account any amounts in escrow) by such Article 8 Seller in connection with such Article 8 Transaction. Further, no Article 8 Seller will be required to bear more than its Pro Rata Share of any escrows, holdbacks, adjustments or indemnification obligations.
- (b) No Liability. All determinations as to whether to complete any Article 8 Transaction be at the sole discretion of the Tag-Along Seller or the Drag-Along Sellers (as applicable), and the Tag-Along Seller and the Drag-Along Sellers and their respective Affiliates shall not have any liability arising from, relating to or in connection with the pursuit, consummation, postponement, abandonment or terms and conditions of any proposed Article 8 Transaction (except to the extent the Tag-Along Seller and the Drag-Along Sellers (as applicable) fails to comply with the provisions of Section 8.9, Section 8.10 or this Section 8.11) or any other transaction pursuant to Article VIII or Article IX.
- (c) Closing. Subject to compliance with this Agreement and applicable law, the closing of an Article 8 Transaction shall take place at such time and place as the Tag-Along Seller or the Drag-Along Sellers, as applicable, shall specify by written notice to each Article 8 Seller. At the closing of such Article 8 Transaction, each Article 8 Seller shall (if applicable) deliver the certificates evidencing the Units to be Transferred by such Article 8 Seller, duly endorsed, or with unit (or equivalent) powers duly endorsed for transfer with signature guaranteed, free and clear of any Liens (other than pursuant to U.S. federal and state (and other jurisdiction) securities laws).
- (d) Expenses. In connection with the negotiation or consummation of any Article 8 Transaction, the Tag-Along Seller or the Drag-Along Sellers, as applicable, and each Article 8 Seller shall bear (i) all costs and expenses incurred by such Person on an individual basis for such Person's sole benefit and (ii) its respective Pro Rata Share of the costs and expenses incurred by such Person, to the extent such costs and expenses are incurred for the benefit of all participating Members (including the costs and expenses incurred in respect of collective documentation and advice for such Article 8 Transaction), are reasonable and documented and are not otherwise paid by the Company or the acquiring party, in each case in connection with such Article 8 Transaction, whether or not such Article 8 Transaction is consummated.

- (e) Expiration. Section 8.9, Section 8.10, and this Section 8.11 (other than any unsatisfied or ongoing obligations pursuant to this Section 8.11 in connection with any Article 8 Transaction) shall expire upon the consummation of a Qualified IPO.
- 8.12 <u>Effect of Notices.</u> Notwithstanding any provision hereof to the contrary, the giving to the holders of Units of any Drag-Along Notice or any Sale Notice shall not obligate the Drag-Along Sellers or Tag-Along Seller, respectively, to consummate or effect any transaction referred to therein.
- 8.13 After-Acquired Securities. All of the provisions of this Agreement shall apply to all of the Units now owned or which may be issued or Transferred hereafter to a Member or other equity holder of the Company in consequence of any additional issuance, purchase, exchange or reclassification of any of such Units, corporate reorganization, or any other form of recapitalization, consolidation, merger, share split or share dividend, or which are acquired by a Member or other equity holder of the Company in any other manner.
- Management Holdco Purchase Rights. If Management Holdco is required pursuant to this Agreement, the Management Holdco Operating Agreement or an agreement approved by the Board and the manager or other managing body of Management Holdco, or the Board and the manager or other managing body of Management Holdco otherwise elect or agree, to repurchase membership interests of Management Holdco (the "Indirect Management Purchase Units") held by a member of Management Holdco (an "Indirect Management Member" and any such repurchase, an "Indirect Management Repurchase"), then the Company shall repurchase from Management Holdco all of the Units held by Management Holdco that are attributable to the Indirect Management Purchase Units (the "Corresponding Direct Management Units") at a price per Corresponding Direct Management Unit equal to the per unit price paid for each Indirect Management Purchase Unit or such other price provided pursuant to Section 5.4. The repurchase described in this Section 8.14 may be effected by causing the Indirect Management Purchase Units being repurchased to be exchanged for Corresponding Direct Management Units followed by a repurchase of such interests by the Company. For clarity, any Indirect Management Repurchase or other transaction contemplated by this Section 8.14 shall be subject to the prior approval of the Board.

# ARTICLE IX QUALIFIED IPO; REGISTRATION RIGHTS

9.1 Qualified IPO Mechanics.

- (i) reorganize the Company into a corporation incorporated under the Laws of the State of Delaware or such other jurisdiction in such manner as the Board deems appropriate, efficient and in the best interests of the Company (and the Board shall use commercially reasonable efforts to effectuate such reorganization in a manner that is tax efficient for both the Company and its Members and the owners of such Members), subject to the requirements of this Section 9.1, including by way of conversion, merger, recapitalization, exchange or asset and liability transfer, in each case, giving effect to the same economic and voting rights contained in this Agreement after taking into consideration the structure of the Company and the Company's Subsidiaries and their respective classes of securities (the transactions described in this clause (i) being referred to herein as a "Conversion Transaction"); (ii) cause the exchange of Units for shares of common stock or other equity securities of a holding company formed to hold all or substantially all of the assets of the Company's Subsidiaries or any other Company Offeror that will be the issuer in such Qualified IPO; (iii) cause the treatment of the Class B Units in connection with such Qualified IPO as determined by the Company to be in accordance with the terms of the Applicable Class B Unit Agreements; or (iv) take any such other action as the Board may determine in good faith to create an appropriate vehicle to effectuate a Qualified IPO.
- (b) Effect of Conversion Transaction. Prior to or upon consummation of a Qualified IPO involving a Conversion Transaction, this Agreement shall be replaced by a stockholders agreement (or agreements) executed by the Company Offeror and the Members containing terms and conditions substantially similar to those set forth herein, except that all references to the Class A Units and the Class B Units, respectively, in this Agreement shall be deemed to refer to the corresponding class, series or tranche of equity securities, as applicable, into which such Class A Units, and Class B Units were converted or exchanged in such Conversion Transaction, respectively, and all references to the Company herein shall be deemed to refer to the Company's successor. The certificate of incorporation and bylaws of the Company Offeror shall be in such form as is approved by the Board; provided that they do not conflict with or vary the terms and conditions of the replacement agreement (or agreements) described in the preceding sentence, and provided, further, that the certificate of incorporation of the Company Offeror shall contain a corporate opportunity provision substantially similar to the one contained in Section 4.9(a)-(c) and provisions reflecting the Company Offeror's election to not be governed by Section 203 of the Delaware General Corporation Law (or similar provision, if the Company Offeror is incorporated in a jurisdiction other than Delaware). The terms of any such stockholders agreement (or agreements) to be entered into among the Company Offeror and the Members in connection with a Conversion Transaction shall, to the extent practicable and permitted by applicable Law, rule, regulation or historical standard, and except as otherwise provided herein, replicate the provisions of this Agreement with respect to (i) the Board designation rights set forth in Article IV (except to the extent that any of the rights granted therein would be inconsistent with the requirements of any securities exchange on which the Company Offeror's securities will be traded or quoted after the Qualified IPO, in which case the rights granted herein shall be replicated to the greatest extent practicable consistent with the requirements of such exchange), (ii) the exculpation and indemnification provisions set forth in Article XII and (iii) the amendment provisions set forth in Article XIII, in each case, with appropriate modification to reflect the corporate structure of the Company Offeror.

- (c) Cooperation of Members. In connection with and in order to facilitate a Qualified IPO and any related Conversion Transaction, each Member shall take all such action as is reasonably necessary (including making any applicable tax elections or the voting of all equity interests in the Company Offeror owned by such person) in order to effect amendments to the Company Offeror's organizational documents as are customary for a company which is to engage in an initial public offering of its equity securities and which are reasonably requested by the managing underwriters in order to expedite or facilitate the disposition of the equity securities of the Company Offeror in connection with such offering and any Conversion Transaction; provided, however, that no Member holding a certain type or class of Units shall be required to take any actions not required from the other Members holding the same type or class of Units. Each Member hereby (i) consents to any such Conversion Transaction or exchange of Units involving a Company Offeror and (ii) agrees to execute, acknowledge and deliver any further agreements, documents or instruments requested by the Company (or any successor thereto) reasonably necessary or customary in connection with this Section 9.1.
- Qualified IPO of Company Offeror. If the Qualified IPO is structured as an offering of the securities (d) of a Company Offeror other than the Company, then this Agreement shall remain in place, substantially on the terms set forth herein, except that: (i) this Agreement shall be amended to (A) provide for (I) the exchange of Class A Units for shares of common stock of the Company Offeror; provided that such exchange shall be effected in a manner consistent with Section 9.2 and (II) the treatment of Class B Units in connection with such Qualified IPO as determined by the Company in accordance with the terms of the Applicable Class B Unit Agreements, (B) adjust the provisions of this Agreement relating to Distributions and allocations to the extent necessary to permit the exchange of Class A Units and treatment of Class B Units contemplated by clause (A) immediately above, and (C) make such other changes as the Board determines to be reasonable and prudent to facilitate such Qualified IPO of such Company Offeror's securities without adversely affecting the relative rights of any Member hereunder in a manner disproportionate (recognizing the priorities and preferences of the Units held by any Member relative to the Units held by any similarly situated Member as of the date hereof as set forth in Section 6.4) as compared to the relative rights of other Members similarly situated; and (ii) the Company shall enter into arrangements with such Company Offeror, in such form as the Board shall deem appropriate, providing for the exercise by the Company of registration rights no less favorable than the registration rights described in Section 9.3 with respect to the common stock or other equity securities of such Company Offeror to be received by the Members upon exchange of Class A Units of the Company; provided that no Member shall have any registration rights hereunder or pursuant to the registration rights described in Section 9.3 or otherwise with respect to Class B Units. Upon any exchange of Class A Units (and Class B Units, if applicable) for shares of capital stock of such Company Offeror, such Units shall no longer be subject to the terms of this Agreement and shall instead become subject to the terms of the certificate of incorporation and bylaws of such Company Offeror. Notwithstanding the consummation of a Qualified IPO effected by a Company Offeror (other than the Company), the rights to Distributions and allocations set forth in Article VI, in a Qualified IPO structured as an "UP-C", the exculpation and indemnification rights and obligations of the Company and the Members set forth in Article XII and the amendment provisions set forth in Article XIII shall each survive such Qualified IPO and continue to apply to the Units under this Agreement.
- (e) Fractional Units. Fractional shares of common stock of a Company Offeror issuable pursuant to a Conversion Transaction or upon exchange of Units for shares of a Company Offeror shall be rounded, cashed out or aggregated and sold in connection with such Qualified IPO in an equitable manner, as determined by the Board.

(f) Qualified IPO Request. Following the (i) fifth (5th) anniversary of the Closing, a Member or Members representing at least a majority of the issued and outstanding Voting Units and (ii) seventh (7th) anniversary of the Closing, each of (A) Viking, so long as the Investor Member Sell-Down Event in respect of Viking has not occurred and (B) Patient Square, so long as the Investor Member Sell-Down Event in respect of Patient Square has not occurred, shall, in each case, have the right, by delivery of a written notice to the Company to require that the Company pursue in a diligent and commercially reasonable manner and consummate a Oualified IPO.

### 9.2 <u>Treatment of Units in Conversion Transaction.</u>

- (a) Conversion Rate. For purposes of the conversion of Units pursuant to a Conversion Transaction or upon the exchange of Units for shares or equity securities of a Company Offeror, each Unit shall be converted at a conversion rate based upon the percentage of the aggregate pre-tax value to be received by each holder of Units being converted that would otherwise have been received if the aggregate equity value of the Company Offeror at the price implied by the Qualified IPO had been distributed in accordance with <a href="https://example.com/Article VI">Article VI</a> as a Distribution following a Liquidity Event.
- (b) Acknowledgment. If a Qualified IPO is effected, in whole or in part, by means of a public offering of the securities of a Company Offeror that is a Subsidiary of the Company that is a holding company for all or substantially all of the assets of the Company and its other Subsidiaries, each of the Members and the Company hereby acknowledges and agrees that such Qualified IPO shall be effected to the greatest extent practicable in a manner that gives effect to this Section 9.2.
- (c) Notwithstanding the foregoing, other than in connection with a Qualified IPO, without the prior written consent of the Fund Investors, the Company (i) shall not effectuate a Conversion Transaction and (ii) shall take any actions, including refraining from making an election to be treated as a corporation, as may be required to ensure that at all times the Company is classified as partnership for United States federal income tax purposes.

### 9.3 Registration Rights.

Registration Rights Agreement. Following a Qualified IPO, the Members (and, if applicable, (a) equityholders of such Members) and the Company Offeror shall enter into a registration rights agreement that shall contain customary registration rights in favor of each Member, including the provisions set forth in Sections 9.3(b) and 9.3(c). All Members (or, as applicable, the equityholders of such Members) shall have customary "piggyback" registration rights, including in any secondary sale effected as part of such Qualified IPO or thereafter, with pro rata cutbacks. The registration rights agreement shall provide that each of (i) Viking, so long as the Investor Member Sell-Down Event in respect of Viking has not occurred and (ii) Patient Square, so long as the Investor Member Sell-Down Event in respect of Patient Square has not occurred), shall, in each case, have the right to demand that the Company or applicable Company Offeror register the resale of all or a part of their securities (a "Demand Registration"), subject to customary limitations. From and after a Qualified IPO, and subject to any applicable lock-up period with respect to such Qualified IPO, as described in Section 9.3(b) below, the Company or Company Offeror will provide reasonable and customary assistance to any Member (or, as applicable, equityholder of such Member) seeking to offer and sell its securities in the public market so as to enable such Member (or, as applicable, equityholders of such Member) to sell its securities pursuant to Rule 144 under the Securities Act or any similar rules or regulations hereinafter adopted by the SEC, including reasonably cooperating with such Member (or, as applicable, equityholders of such Member) to facilitate the timely preparation and delivery of certificates representing the securities to be sold and not bearing any Securities Act legend in connection with any such sale.

- (b) Lock-Up. Each Member (or, as applicable, the equityholders of such Members) shall not Transfer any securities of the Company or a Company Offeror for a period commencing on the date marketing commences with respect to a Qualified IPO and ending not more than one hundred eighty (180) days following the closing of such Qualified IPO. Each Member (or, as applicable, the equityholders of such Members) agrees to execute and deliver such other agreements as may be reasonably requested by the Company or its underwriter which are consistent with the foregoing or which are necessary to give further effect thereto. In addition, if requested by the Company or any representative of the underwriter, each Member shall provide, within ten (10) Business Days of such request, such information as may be required by the Company or such representative in connection with the completion of any public offering of the Company's securities pursuant to a Qualified IPO. The Company may impose stop-transfer instructions with respect to the equity securities subject to the foregoing restriction until the end of the applicable period.
- (c) *Underwriter Selection*. The underwriter for any registration statement filed by the Company of Company Offeror in connection with a sale of units or other equity securities thereof shall be selected by the Board.

# ARTICLE X EVENTS OF DISSOLUTION

- 10.1 <u>Dissolution.</u> The Company shall be dissolved upon the occurrence of any of the following events (each an "Event of Dissolution"):
  - (a) Subject to Section 4.10, a determination by the Board to dissolve the Company; or
  - (b) A judicial dissolution of the Company under Section 18-802 of the Act.

The Members hereby agree that the Company shall not dissolve prior to the occurrence of an Event of Dissolution and that no Member shall seek a dissolution of the Company under Section 18-802 of the Act. No other event, including the retirement, withdrawal, insolvency, liquidation, dissolution, insanity, resignation, expulsion, bankruptcy, death, incapacity, or adjudication of incompetency of a Member, shall cause a dissolution of the Company or cause the existence of the Company to terminate.

### ARTICLE XI TERMINATION

- 11.1 <u>Liquidation.</u> If an Event of Dissolution shall occur, then the Company shall be liquidated and its affairs shall be wound up. All proceeds from such liquidation shall be distributed as set forth below, in accordance with the provisions of Section 18-804 of the Act:
- (a) to creditors, including Members who are creditors to the extent permitted by Law, in satisfaction of the Company's liabilities; and
  - (b) then, to the Members in accordance with Article VI.
- 11.2 <u>Final Accounting.</u> In the event of the dissolution of the Company, prior to any liquidation, a proper accounting shall be made to the Members from the date of the last previous accounting to the date of dissolution.
- 11.3 <u>Distribution in Kind.</u> If the Board determines in connection with the liquidation of the Company following an Event of Dissolution that a portion of the Company's assets must be distributed in kind to the Members, then such assets shall be so distributed in kind to the Members in undivided shares therein as tenants in common in the manner specified in Section 6.1.
- 11.4 <u>Certificate of Cancellation.</u> Upon the completion of the winding up of the Company's affairs and distribution of the Company's assets, the Company shall be terminated and the Members shall cause the Company to execute and file a Certificate of Cancellation in accordance with Section 18-203 of the Act.

# ARTICLE XII EXCULPATION AND INDEMNIFICATION

- 12.1 <u>Exculpation.</u> Notwithstanding any other provisions of this Agreement, to the fullest extent permitted by the Act, none of the Covered Persons shall be liable to the Company or any Member, Director or other Person who is a party to or bound by this Agreement for any Losses (as defined below) arising out of or relating to any act or omission, or any consent or approval given or withheld, in each case, in relation to the management of the Company or otherwise relating to the conduct of the Company's business and affairs; <u>provided</u> that nothing herein shall limit the liability of any Person for any act or omission by such Covered Person that constitutes fraud, a willful breach of applicable Law, willful misconduct, an intentional breach of such Person's obligations under this Agreement (if any) or a bad-faith violation of the implied contractual covenant of good faith and fair dealing (to the extent applicable).
  - 12.2 <u>Indemnification</u>.

- (a) To the fullest extent permitted by the Act, the Company shall indemnify and hold harmless each current and former (i) Covered Person and (ii) officer of the Company and its Subsidiaries (collectively, the "Indemnified Persons" and each individually an "Indemnified Person") from and against any and all losses, claims, demands, liabilities, reasonable out-of-pocket expenses (including attorneys' fees), judgments, fines, settlements and other amounts ("Losses") incurred by such Person in connection with any and all claims, demands, actions, suits or proceedings, civil, criminal, administrative or investigative in which such Indemnified Person may be involved, or threatened to be involved, as a party, a witness or otherwise, by reason of, or in connection with, any act or omission, or any consent or approval given or withheld, in each case, in relation to the management of the Company or otherwise relating to the conduct of the Company's business and affairs ("Claims"). Notwithstanding the foregoing, no current or former Indemnified Person shall be entitled to indemnification under this Section 12.2 with respect to: (i) any act or omission by such Indemnified Person that constitutes fraud, a willful breach of applicable Law, willful misconduct, or an intentional breach of this Agreement or a bad faith violation of the implied contractual covenant of good faith and fair dealing (to the extent applicable); or (ii) any Claim initiated by such Indemnified Person unless such Claim (or part thereof) was (A) brought to enforce such Indemnified Person's rights to indemnification or advancement of expenses hereunder or (B) authorized or consented to by the Board prior to the initiation of such Claim.
- (b) Reasonable, documented, out-of-pocket expenses (including attorneys' fees) incurred by an Indemnified Person in defending any Claim shall be paid by the Company in advance of the final disposition of such Claim upon receipt by the Company of an undertaking by or on behalf of such Indemnified Person to repay the amount advanced if it shall be ultimately determined in a final, nonappealable determination of a court of competent jurisdiction that such Indemnified Person is not entitled to be indemnified by the Company as authorized by Section 12.2(a). If the Company indemnifies any Indemnified Person or former Indemnified Person against, and shall be entitled to seek contribution from, any third party, including any insurance company, to recover the amount of such indemnification (or such portion thereof as to which the Company shall be entitled to contribution) after the Indemnified Person or former Indemnified Person shall have been fully and completely indemnified (whether pursuant to this Agreement or otherwise) in respect of the Claim which gave rise to such indemnification. Any such Indemnified Person or former Indemnified Person shall fully cooperate with the Company at the Company's expense in its efforts to enforce against any such third party the rights to which it is so subrogated.
- 12.3 <u>Effect of Modification.</u> Any repeal or modification of this <u>Article XII</u> shall not adversely affect any rights of such Indemnified Person pursuant to this <u>Article XII</u>, including the right to indemnification and to the advancement of expenses of an Indemnified Person existing at the time of such repeal or modification with respect to any acts or omissions occurring prior to such repeal or modification.
- 12.4 <u>Nonexclusivity of Rights.</u> The rights conferred on any Indemnified Person by this <u>Article XII</u> shall not be exclusive of any other rights that such Indemnified Person may have or hereafter acquire under any statute, provision of this Agreement, agreement, vote of members or disinterested directors or otherwise.

### ARTICLE XIII AMENDMENT TO AGREEMENT

#### 13.1 Amendments.

- (a) Subject Section 13.1(b) and except as otherwise provided in this Agreement, including Section 4.3 and Section 4.10, amendments, supplements or modifications to this Agreement shall be approved by each of (i) a majority of the entire Board. (ii) Viking, so long as the Investor Member Sell-Down Event in respect of Viking has not occurred, and (iii) Patient Square, so long as the Investor Member Sell-Down Event in respect of Patient Square has not occurred, and any such amendment, supplement or modification shall be binding on the parties hereto, including the Company and the Members; provided, that (i) any amendment, supplement or modification of this Agreement in a manner that would have a material disproportionately adverse impact on any right or obligation of any Member, the consent of such Member shall also be required, (ii) in no event shall any Member's Total Funding Commitment be amended, supplemented or modified without the written consent of such Member, and (iii) in no event shall any Member be obligated to make any cash or non-cash contribution to the Company's capital other than as contemplated by Section 5.1(e)(i) (as in effect on the date hereof) without the written consent of such Member.
- (b) An amendment shall become effective as of the date specified in the applicable approval or, if none is specified, as of the date of such approval or as otherwise provided in the Act. Notwithstanding the provisions set forth in Section 13.1(a), but subject to Section 4.3 and Section 4.10, the Board may amend this Agreement, including the schedules hereto, without the approval of any Members (i) to reflect the issuance, redemption, repurchase or forfeiture of Class B Units in connection with the Applicable Class B Unit Agreements, (ii) to implement the admission of Substitute Members or Additional Members, (iii) to change the name of the Company and (iv) to implement the provisions set forth in Article IX.

# ARTICLE XIV GENERAL PROVISIONS

- 14.1 <u>Expenses.</u> Except as otherwise expressly provided herein or as contemplated by Section 10.1 of the Transaction Agreement, all costs and expenses, including fees and disbursements of legal counsel, financial advisors and accountants, incurred in connection with this Agreement shall be borne by the party incurring such costs and expenses.
- 14.2 <u>Notices.</u> All notices hereunder shall be in writing and shall be deemed to have been duly given or made (i) as of the date delivered if delivered personally, mailed by registered or certified mail (postage prepaid, return receipt requested) or sent by overnight courier (providing proof of delivery) or (ii) as of the date transmitted if sent by electronic transmission to the following electronic mail addresses, in each case, to the addresses below (or at such other contact information for a party hereto as shall be specified by like notice):

If to the Company:

BridgeBioX Pharma, LLC 3160 Porter Drive, Suite 250 Palo Alto, CA 94304 Attention: Legal Department Email: legal@bridgebio.com

If to a Member, to the address of such Member specified on Schedule A hereto.

Such addresses may be changed, from time to time, by means of a notice given in the manner provided in this Section 14.2.

Publicity; Confidentiality. Except as may be required by applicable Disclosure Law, none of the parties hereto 14.3 shall issue a press release or public announcement or otherwise make any disclosure concerning this Agreement or the transactions contemplated hereby or any other information, statements, reports, trade secrets, documents, and other materials concerning the Company, any Member or any of their respective Affiliates that any Member may receive or that may be disclosed, distributed or disseminated (whether in writing, orally, electronically or by other means) by or to such Member or any representative of such Member, or otherwise as a result of such Member's ownership of any Units. Each Member agrees (on behalf of itself and its directors, officers, managers, equityholders, partners, employees, agents and members (collectively, "Representatives")) that it will keep confidential the terms of this Agreement and all information that it receives from the Company in its capacity as a Member (including all confidential or proprietary information concerning the Company and its Subsidiaries and the business of the Company and its Subsidiaries) except that it may disclose such information (a) as may be required in the course of obtaining legal advice or enforcement with respect to the rights and obligations created hereby, (b) to its legal, financial, accounting and other advisors, (c) as may be required by law, regulation, rule, court order, subpoena or other stock exchange or governmental body or in the preparation of tax returns or other governmental or regulatory filings, (d) as and to the extent that such Member's legal counsel determines that such disclosure is required by applicable Law, (e) to the extent required in the good-faith performance of such Member's duties to the Company and its Subsidiaries as an employee thereof, and (f) to the extent reasonably necessary to enforce the terms of this Agreement; provided, however, that each Member may disclose such information to (i) any of its actual and prospective equityholders and lenders, and any equityholder thereof that is an investment fund may disclose such information to its actual and prospective limited partners or other investors (provided in each case that such Persons are subject to reasonable confidentiality obligations with respect to such information), (ii) any Person in connection with financings, acquisitions, divestitures and similar events involving the Company and its Affiliates (provided in each case that such Persons are subject to reasonable confidentiality obligations with respect to such information), (iii) any of its directors, officers, employees or professional advisers to the extent that such disclosure is reasonably necessary in connection with the such Member's investment in the Company for use solely for such purpose and (iv) any Person in connection with such Member's ordinary course internal market, industry and investment analyses, so long as the recipient of such information is subject to a duty of confidentiality to such Member with respect to such information. In the event of any disclosure as permitted above under clauses (a), (b) and (f) hereof to a Member's representatives and advisors, the disclosing party shall direct any Person receiving such information to maintain its confidentiality and shall be responsible for any breach of the confidentiality provisions hereunder by any such Person receiving such information. Notwithstanding anything in this Section 14.3 to the contrary, except in connection with a Change of Control and with the approval of the Board, or except as permitted under clause (e) hereof, no Director or Member shall provide any information that it receives from the Company in its capacity as a Director or Member to any Person (including any officer, director, member, manager, partner, trustee, employee, or agent thereof) that, in the good-faith reasonable judgment of the Board, is an actual or potential competitor of the Company.

- 14.4 Entire Agreement and Waiver. This Agreement, together with all exhibits and schedules hereto and all other agreements referenced herein and therein, shall constitute the entire agreement between the parties hereto relating to the subject matter hereof and supersedes all prior contracts, agreements, discussions and understandings between them. No provision of this Agreement shall be deemed to modify, amend, waive, rescind or otherwise impact the terms of the Transaction Agreement. Any waiver (express or implied) of any default or breach of this Agreement must be in writing and shall not constitute a waiver of any other or subsequent default or breach.
- 14.5 <u>Conflict between This Agreement and Class B Unit Agreement.</u> Notwithstanding anything to the contrary in this Agreement, in the event of a conflict between any term or provision contained herein and a term or provision of any Applicable Class B Unit Agreement or the Management Holdco Operating Agreement, the Board acting in good faith shall be entitled to resolve such conflict. Notwithstanding any other provision of this Agreement to the contrary, no Member who holds only Class B Units shall have any right to receive or review a copy of <u>Schedule A</u> or <u>Schedule B</u> to this Agreement (except for information on <u>Schedule A</u> or <u>Schedule B</u> that relates solely to such Member) or obtain other information about the identities of the other Members or the size or nature of their interests in the Company; provided, however, that any Person may view a copy of <u>Schedule A</u> or <u>Schedule B</u> if the Board determines that it is reasonably necessary for such Person to perform his or her duties in connection with the Company.
- 14.6 <u>Counterparts.</u> This Agreement may be executed in any number of counterparts, each of which, when so executed and delivered, shall be an original, but all of which together shall constitute one agreement binding on the parties hereto. Transmission of an executed counterpart by electronic transmission of this Agreement shall be deemed to constitute due and sufficient delivery of such counterpart, and such signatures shall be deemed original signatures for purposes of the enforcement and construction of this Agreement.
- 14.7 <u>Severability.</u> If any provision of this Agreement is held by a court of competent jurisdiction to be invalid or unenforceable for any reason, such provision shall be ineffective to the extent of such invalidity or unenforceability; provided, however, that the remaining provisions will continue in full force without being impaired or invalidated in any way unless such invalid or unenforceable provision or clause shall be so significant as to materially affect the Members' expectations regarding this Agreement. Otherwise, the Members agree to replace any invalid or unenforceable provision with a valid provision which most closely approximates the intent and economic effect of the invalid or unenforceable provision.
  - 14.8 Governing Law; Venue; Waiver of Jury Trial.

- (a) This Agreement, and all claims, causes of action and disputes (whether in contract or in tort or otherwise, or whether at law (including at common law or by statute) or in equity) that may be based on, arise out of or relate to this Agreement or the negotiation, execution, performance or subject matter hereof, shall be governed by the Laws of the State of Delaware applicable to agreements made and to be performed solely therein, without giving effect to principles of conflicts of law.
- (b) The parties hereto irrevocably submit to the jurisdiction of the Court of Chancery of the State of Delaware and any state appellate court therefrom within the State of Delaware (or, if the Court of Chancery of the State of Delaware declines to accept jurisdiction over a particular matter, any federal court within the State of Delaware (and any appellate court therefrom)) (collectively, the "Chosen Courts") in any action or proceeding that arises in respect of the interpretation and enforcement of the provisions of this Agreement and the documents referred to in this Agreement or in respect of the transactions contemplated herein, and hereby waive, and agree not to assert, as a defense in any action, suit or proceeding for interpretation or enforcement hereof or any such document that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in the Chosen Courts or that venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such Chosen Courts, and the parties hereto irrevocably agree that all claims with respect to such action, suit or proceeding shall be heard and determined exclusively by such Chosen Courts.
- (c) EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY BE BASED UPON, ARISE OUT OF OR RELATE TO THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HERETO HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY PROCEEDING DIRECTLY OR INDIRECTLY BASED UPON, ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREIN. EACH PARTY HERETO CERTIFIES AND ACKNOWLEDGES THAT: (I) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER; (II) SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THE FOREGOING WAIVER; (III) SUCH PARTY MAKES THE FOREGOING WAIVER VOLUNTARILY; AND (IV) SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVER AND CERTIFICATIONS IN THIS SECTION 14.8.
  - 14.9 <u>Binding Effect.</u> This Agreement shall be binding upon, and inure to the benefit of, the Members.
- 14.10 <u>Additional Documents and Acts.</u> The Members shall from time to time execute or cause to be executed all other documents or cause to be done all filing, recording, publishing, or other acts as may be necessary or desirable to comply with the requirements for the operation of a limited liability company under the laws of the State of Delaware and all other jurisdictions in which the Company may from time to time conduct business.

- 14.11 <u>No Third-Party Beneficiary.</u> This Agreement is made solely for the benefit of the parties hereto and no other Person shall have any rights, interest, or claims hereunder or otherwise be entitled to any benefits under or on account of this Agreement as a third-party beneficiary or otherwise, except for the rights of the Member Related Parties pursuant to <u>Section 14.12</u> or any Indemnified Person pursuant to <u>Section 12.2</u>.
- Nonrecourse. Notwithstanding anything to the contrary in this Agreement, except in the case of fraud, this Agreement may only be enforced against, and any action, dispute, claim, suit or other proceeding for breach of this Agreement may only be made against, the entities that are expressly identified herein as parties and none of the former, current and future Affiliates, directors, officers, managers, employees, advisors, representatives, shareholders, members, managers, partners, successors and assigns of a Member or any Affiliate thereof or any former, current and future Affiliate, director, officer, manager, employee, advisor, representative, shareholder, member, manager, partners, successor and assign of any of the foregoing (collectively, "Member Related Parties") shall have any liability for any liabilities or obligations of the parties hereto for any action, dispute, claim, suit or other proceeding (whether in tort, contract or otherwise) for breach of this Agreement or in respect of any oral representations made or alleged to be made in connection herewith, the Company or any other Member or their respective Affiliates shall have no rights of recovery in respect hereof against any Member Related Party and no personal liability shall attach to any Member Related Party through such Member or otherwise, whether by or through attempted piercing of the corporate veil, by or through an action, dispute, claim, suit or other proceeding (whether in tort, contract or otherwise) by or on behalf of a Member against any Member Related Party, by the enforcement of any judgment, fine or penalty or by virtue of any statute, regulation or other applicable Law, or otherwise. The Company agrees that (i) each of Viking, Patient Square, Sequoia, Frazier, Cormorant and Aisling and their respective Affiliates (each, a "Fund Investor") are professional investment funds, and as such invest in numerous portfolio companies, some of which may be deemed competitive with the Company or its Subsidiaries' businesses, and, to the extent permitted by applicable Law, no Fund Investor shall be liable to the Company or any of its subsidiaries for any claim arising out of, or based upon, the investment by any Fund Investor in any entity competitive with the Company or its Subsidiaries and (ii) to the extent permitted by applicable Law, each Fund Investor is permitted to acquire, own and/or dispose of any debt, equity and/or equity-linked securities or other investment in any competitor of the Company or its Subsidiaries without any liability to the Company or its subsidiaries; provided that nothing in the foregoing clauses (i) and (ii) shall relieve any Fund Investor from liability associated with the unauthorized disclosure or use of the Company's confidential information in breach of this Agreement or any other obligation of confidentiality or non-use owed to the Company or any of its Subsidiaries by the Fund Investors.
- 14.13 <u>Successors and Assigns.</u> This Agreement is personal to the parties hereto and shall not be capable of assignment; it being understood that the foregoing shall not be read to limit any Transfer pursuant to and in accordance with <u>Article VIII</u>. All of the terms and provisions of this Agreement shall inure to the benefit of and be binding upon each of the parties hereto and their respective successors and permitted transferees, if any.
- 14.14 <u>Specific Performance</u>. Each of the Members acknowledges and agrees that in the event of any breach or threatened breach of this Agreement, the nonbreaching party or parties would be irreparably harmed and could not be made whole by monetary damages. The Members hereby agree that, in addition to any other remedy to which they may be entitled at law or in equity, they shall be entitled to compel specific performance of this Agreement in any action instituted in any court set forth in <u>Section 14.8</u>.

- 14.15 <u>Representations and Warranties.</u> Each Person that becomes an Additional Member or a holder of Units on or after the date hereof (each a "<u>Warrantor</u>") makes, as of the date such Warrantor becomes an Additional Member or acquires Units, the representations and warranties to the Company set forth on <u>Schedule D</u> hereto.
- 14.16 <u>Survival of Representations and Warranties; Termination of Covenants.</u> All representations and warranties contained in this Agreement or made in writing by any party hereto in connection herewith shall survive the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby. In the event any provision of this Agreement is terminated, such termination shall not relieve any party hereto from liability for such party's prior breach of this Agreement or for failure by such party to perform such party's obligations hereunder.

# 14.17 Agent's Rights.

- (a) Notwithstanding anything contained herein to the contrary, BridgeBio or its Affiliates, each in its capacity as a Member of the Company (each in such capacity, the "Pledging Member"), shall be permitted to pledge or hypothecate any or all of its economic ownership of Units, including all rights to receive the Distributions and allocations of income, gain, loss, deduction, credit and similar items to which the Pledging Member would be entitled, to Blue Owl Capital Corporation, as administrative agent (together with its permitted successors and assigns, in such capacity, the "Agent") under that certain Financing Agreement, dated as of January 17, 2024 (as amended, restated, amended and restated, supplemented or otherwise modified from time to time, the "Financing Agreement"), by and among BBIO, BridgeBio, Agent, the lenders from time to time party thereto and the other parties from time to time party thereto, and any transfer of such economic rights pursuant to the Agent's exercise of remedies in connection with any such pledge or hypothecation shall be permitted under this Agreement with no further action or approval required hereunder. Notwithstanding anything contained herein to the contrary, upon an event of default under the Financing Agreement, the Agent (or its assignee, transferee or designee, as applicable) (i) shall have the right, without further approval of any Member and without becoming a Member, to take economic ownership of the Pledging Member's Units (the "Foreclosed Units") and to receive the Distributions and allocations of income, gain, loss, deduction, credit and similar items to which the Pledging Member would be entitled and (ii) shall not have the right to exercise any of the Pledging Member's voting rights, consent rights, control rights or status rights as a Member. For the avoidance of doubt, upon any such transfer of the economic ownership of Foreclosed Units, the Pledging Member shall cease to have any voting rights, consent rights, control rights or status rights as a Member with respect to the Foreclosed Units.
- (b) The parties hereto further agree that so long as any pledge or hypothecation of the Units in favor of the Agent is in effect, (x) the Company shall not elect that the Units become governed by Article 8 of the Uniform Commercial Code as in effect from time to time in the State of Delaware, (y) this Section 14.17 may not be amended or modified, in each case, without the prior written consent of the Agent and (z) the Agent shall be a third party beneficiary of the provisions of this Section 14.17.

[Signature Pages Follow]							
			77				

IN WITNESS WHEREOF, the parties have duly executed this Agreement as of the day and year first above written.

# **MEMBERS:**

# **BRIDGEBIO PHARMA LLC**

By:/s/Neil Kumar

Name: Neil Kumar Title: Manager

# VIKING GLOBAL OPPORTUNITIES ILLIQUID INVESTMENTS SUB-MASTER LP

By: Viking Global Opportunities Portfolio GP LLC, its general partner

By: /s/ Katerina Novak

Name: Katerina Novak Title: Authorized Signatory

# VIKING GLOBAL OPPORTUNITIES DRAWDOWN (AGGREGATOR) LP

By: Viking Global Opportunities Drawdown Portfolio GP LLC, its general partner

By: /s/ Katerina Novak

Name: Katerina Novak Title: Authorized Signatory

# PATIENT SQUARE BRAVO AGGREGATOR, LP

By: Patient Square Equity Advisors, LP

Its: General Partner

By: Patient Square Capital Holdings, LLC

Its: General Partner

By: /s/ Adam Fliss

Name: Adam Fliss
Its: General Counsel

# SC US/E GROWTH FUND X MANAGEMENT, L.P., as nominee

By: SC US (TTGP), LTD., a Cayman Islands exempted company

Title: General Partner

By:/s/ Roelof F. Botha

Name: Roelof F. Botha Title: Authorized Signatory

# SC US/E VENTURE FUND XVIII MANAGEMENT, L.P., as nominee

By: SC US (TTGP), LTD., a Cayman Islands exempted company

Title: General Partner

By: /s/ Roelof F. Botha

Name: Roelof F. Botha Title: Authorized Signatory

# FRAZIER LIFE SCIENCES XI, L.P.

By: FHMLS XI, L.P. Its general partner

By: FHMLS XI, L.L.C. Its general partner

By: /s/ Patrick Heron
Name: Patrick Heron
Title: Managing Director

# FRAZIER LIFE SCIENCES PUBLIC FUND, L.P.

By: FHMLSP, L.P. Its general partner

By: FHMLSP, L.L.C. Its general partner

By: /s/ Albert Cha
Name: Albert Cha
Title: Managing Director

### FRAZIER LIFE SCIENCES PUBLIC OVERAGE FUND, L.P.

By: FHMLSP Overage, L.P.

Its general partner

By: FHMLSP Overage, L.L.C.

Its general partner

By: /s/ Albert Cha
Name: Albert Cha
Title: Managing Director

# CORMORANT PRIVATE HEALTHCARE FUND IV, LP

By: Cormorant Private Healthcare GP IV, LLC

By: /s/ Bihua Chen

Name: Bihua Chen Title: Managing Member

# CORMORANT PRIVATE HEALTHCARE FUND V, LP

By: Cormorant Private Healthcare GP V,

LLC

By:/s/Bihua Chen

Name: Bihua Chen Title: Managing Member

# CORMORANT GLOBAL HEALTHCARE MASTER FUND,

LP

By: Cormorant Global Healthcare GP, LLC

By: /s/ Bihua Chen

Name: Bihua Chen Title: Managing Member

# AISLING V BRIDGE SPLITTER LP

By: Aisling Capital Partners V, LP

Its: General Partner

By: Aisling Capital Partners V LLC

Its: General Partner

By: /s/ Robert Wenzel

Name: Robert Wenzel

Title: Chief Financial Officer

# KUMAR HALDEA REVOCABLE TRUST

By: /s/ Neil Kumar

Name: Neil Kumar Title: Trustee

# MANAGEMENT HOLDCO: BRIDGEBIOX MANAGEMENT AGGREGATOR, LLC

By: BridgeBioX Pharma, LLC, as Managing Member of Management Holdco

By: /s/ James Li

Name: James Li Title: President

# **Document and Entity Information**

Aug. 16, 2024

**Cover [Abstract]** 

Document Type 8-K
Amendment Flag false

Document Period End Date Aug. 16, 2024
Entity File Number 001-38959

Entity Registrant Name BridgeBio Pharma, Inc.

Entity Central Index Key 0001743881

Entity Incorporation, State or Country Code DE

Entity Tax Identification Number

Entity Address, Address Line One

Entity Address, Address Line Two

Entity Address, City or Town

Palo Alto

Entity Address, State or Province CA
Entity Address, Postal Zip Code 94304
City Area Code 650
Local Phone Number 391-9740

Title of 12(b) Security Common Stock, par value \$0.001 per share

Trading SymbolBBIOSecurity Exchange NameNASDAQEntity Emerging Growth Companyfalse

Written Communications false
Soliciting Material false
Pre-commencement Tender Offer false
Pre-commencement Issuer Tender Offer false

```
"Boba": 1 | "Boba": 1 | "Boba": 1 | "Boba": 1 | "Boba": 2 | "Boba"
                                      |
| "labelLink": {
| "local": [
| "bbio-20240816_lab.xml"
                                           ),
"inline": {
  "local": [
  "ef20034419 8k.htm"
                             "Meydtandard": 23,
"keydtandard": 0,
"keydtatom": 0,
"axisStandard": 0,
"axisStandard": 0,
"memberCustom": 0,
"memberCustom": 0,
"memberCustom": 0,
"memberCustom: 0,
"http://xbx1.sec.gov/de1/2023": 3
"http://xbx1.sec.gov/de1/2023": 3
                             ], "contextCount": 1,
"entivyCount": 1,
"segmentCount": 0,
"elementCount": 0,
"unitCount": 0,
"hatp"Xoronies": {
"http://xbrl.sec.gov/dei/2023": 23
},
                        "MeasTanonnies": [
"thip://dmi.sec.gov/dmi/2027: 23
"thip://dmi.sec.gov/dmi/2027: 23
"thip://dmi.sec.gov/dmi/2027: 23
"thip://dmi.sec.gov/dmi/2027: 23
"thip://dmi.sec.gov/dmi/2027: 25
"report": [
"report": [
"report": [
"relat": http://dmidgato.com/role/DocumentAndEntityInformation",
"Longians": "Document",
"Longians": "Document",
"group/yys": "document",
"group/yys": "document",
"group/yys": "document",
"group/yys": "document",
"group/yys": "document",
"group/yys": "document",
"group/ys": "document",
"gr
                                           "Body",
"httml
"reportCount"; 1,
"reportCount"; 1,
"stateMarin"; 1 true,
"first"; 1 true,
"
                                                     ],
"reportCount": 1,
"baseRef": "ef20034419_8k.htm",
"first": true,
"unique": true
                             ],
"tag": {
  "dei_kmendmentFlag": {
  "mbrltype": "boolsanitenType",
  "mbrltype": "boolsanitenType",
  "nauri": "http://mbrl.aec.gov/dei/2023",
  "localname": "amendmentFlag
  "persentation": {
  "mttp://mbrldyshol.com/role/bocumentAndEntityInformation"
  "mttp://middyshol.com/role/bocumentAndEntityInformation"
                                           "http://www."
'slago": [
"Wen-wall !
"Wen-wall !
"Wen-wall !
"Wen-wall !
"Associated Flago",
"Associated F
                                      "dei_CityAreaCode"; [
"shritype": "normalisedStringItemType",
"shritype": "normalisedStringItemType",
"shritype": "normalisedStringItemType",
"localname": "CityAreaCode",
"localname": "CityAreaCode",
"presentation": [
"http://bridgebio.com/role/DocumentAndEntityInformati]
                                      "http.,...."
"lang":[
"en-us":[
"shell":[city Area Code",
"documentation": "Area code of city"
                                                     "auth_ref": []
                                           l'dai CoverAbstract": {
    "xbritype": "atringitemType",
    "mauri": "mipri/mbrl.ace.gov/dai/2023",
    "localname": "CoverAbstract",
    "lang": {
        "serum": "eserum":
        "lang": "cover [Abstract]",
        "documentation": "Cover page."
    }
                                                     "auth_ref": []
                                      "autr_pet": [1

"set_bocumentFiscalPeriodFocus": [
"shritypet": "fiscalPeriodTestType"
"saurit": "burly/shrisac.gov/ds/7023",
"localname": "bocumentFiscalPeriodFocus",
"presentation": [
"http://bridgebio.com/role/bocumentAndEntityInformati]
                                                                                  erclard of the control of the contro
                                                "auth_ref": []
                                                ),

doi DocumentFiscalYearFocus": [
"xbrltype": "gYearItenffype",
"msur!", "http://xbrl.sec.gov/dei/2023",
"localname": "DocumentFiscalYearFocus",
"presentation": [
"http://bridgebio.com/role/DocumentAndEn ],
```

```
"role": {
   "label": "Document Fiscal Year Focus",
   "documentation": "This is focus fiscal y
           "auth_ref": []
       ],

"dei_DocumentPeriodEndDate": {
    "xbrltype": "dateIntemType",
    "nsuri": "http://xbrl.ace.gov/dei/2023",
    "localname": "DocumentPeriodEndDate",
    "presentation": [
    "http://bridgebio.com/role/DocumentAndE
       ],

"dei_DocumentType": {
    "sknltype": "submissionTypeItemType",
    "nsuris": "http://khrl.sec.gov/dei/2023",
    "localname": "DocumentType",
    "presentation": [
    "http://bridgebio.com/role/DocumentAndS
         "auth_ref": []
"with, pat's []

"Mail point productions of the state of 
       )
)
),
"auth_ref": []
   "auth_ref": [;
],

"del_EntityAddressLine2": [
"skritype": "normallredStringItemType",
"nsuri": "http://zbrl.sec.gov/dsi/2023",
"localname": "EntityAddressAddressline2";
"presentation": [
"http://bridgebio.com/role/DocumentAndEnt]
],
...
       "http://-
| "http://-
| "ang": [
| "ang": [
| "acust: [
| "cols": [
| "habd!" 'Thitty Address, Address line Two",
| "documentation": "Address line 2 such as Street or Suite number
       ),

"dei EntityAddressAddressLine3": (
"xbrltype": "normalizedStringItemType",
"nauri": "http://sbrl.ace.gov/dei/20;"
"localname": "EntityAddressAddressLine3",
"presentation": [
"http://bridgebio.com/role/DocumentAndEn
     "mttp",""

"alang": [
"alang": [
"label": "Entity Address, Address Line Three",
"documentation": "Address Line 3 such as an Office Park"
"
         "auth_ref": []
   www.pefr: []

"doi EntityAddressCityOrTown": {
    Twbritype": "normalizedStringTeamType",
    "masuft" "mitpp://milec.gov/doi/2023",
    "masuft" "mitpp://milec.gov/doi/2023",
    "presentation":
    "http://bridgebio.com/role/DocumentAndEntityIn ],
    "http://bridgebio.com/role/DocumentAndEntityIn ],
     ],

"dei EntityAddressCountry": {
"xbritype": "countryCodeItemType",
"nsur!": "http://xbrl.sec.gov/dei/2023",
"localname": "EntityAddressCountry",
"localname": "EntityAddressCountry",
"presentation": [
"http://bridgebio.com/role/DocumentAnd5]
   ]*
"del EntityAddressPostalZipCode": {
    "xkrltype": "normalizedStringItesfType",
    "naurin' "http://skrl.sec.gov/dei/2022",
    "localname": "EntityAddressPostalZipCode",
    "presentation": [
    "http://bridgsbio.com/role/DocumentAndEnt
     "Right,"
"lang": {
"en-us": {
"cole": {
"label": "Entity Address, Postal Zip Code",
"documentation": "Code for the postal or zip code"
           ),
"auth_ref": []
 "auth_rest: |
del EntipAddressStateOrProvince": |
"absTtype": "stateOrProvinceTtenType",
"nauri": "http://bbl.aec.gov/del/2023",
"localname": "EntipOrther StateOrProvince",
"presentation": |
"tttp://bridgetic.com/role/DocumentAndEntityIn
     "Mang": (
"Genum": (
"Fold": (
"stabel": "Entity Address, State or Province",
"documentation": "Name of the state or province."
         "auth_ref": []
       ],

"doi EntityCentralIndexKey": {
    "xhrltype": "centralIndexKeyTenType",
    "nauri": "http://xhrl.ace.gov/doi/2023",
    "localname": "EntityCentralIndexKey",
    "presentation": [
    "http://bridgebio.com/role/DocumentAndE
         )
"auth_ref": [
"rl"
       ],

"doi EntityEmergingCrowthCompany": {
    "xbrltype": "BooleanItenType",
    "nauri": "http://xbrl.sec.gov/dai/2023",
    "localname": "EntityEmergingCrowthCompany",
    "presentation": [
    "http://bridgebio.com/role/DocumentAndEnti
     "http://www.j.laag*[.[
"en-us".[
"en-us".[
"en-us".[
"decimentation": "Indicate if registrant meets the energing growth company
"documentation": "Indicate if registrant meets the energing growth company
           )
),
"auth_ref": [
"r1"
         ),

"de! EntityFileNumber": {
    "xbrltype": "fileNumberItemType",
    "nsur!": "http://xbrl.sec.gov/dei/2023",
    "localname": "EntityFileNumber",
    "presentation": [
    "http://fidgebic.com/role/DocumentAndEr
```

```
!, "lang": {
  "en-us": {
  "cole": {
  "label": "Entity File Number",
  "documentation": "Commission f'
                  )
},
"auth_ref": []
         ),

"doi.EntityIncorporationStateCountryCode": {
    "xbrltype": "edgarStateCountryComeType",
    "naur!": "http://xbrl.ace.gov/dsi/2023",
    "localname": "EntityIncorporationStateCountryCode",
    "presentation": {
    "thtp://bridgebio.com/role/DocumentAndEntityInforma
    "Risp....
'lang': [
"Gours': [
"Gours': [
"Annie: Thmilty Incorporation, State or Country Code",
"Babal': Thmilty Incorporation, State or Country Code",
"Adcumentation", "Two-character EDGAR code representing the state or country of in-
"auth_ord": []

"all_intiple_". mormaliaedEringTestType_",
"anitype_". mormaliaedEringTestType_",
"mostle_". MormaliaedEringTestType_",
"localmase": "EntityNegistrantName",
"presentation":
"http://midphilo.com/role/DocumentAndEntityInformation
"lamg": [
"entity": [
"entity": [
"entity": [
"entity": [
"entity": [
"documentation": "The exact name of the entity filling
"documentation": "The exact name of the entity filling
              "auth_ref": [
         ],

"dei_EntityTaxIdentificationNumber": {
    "xbritype": "employerIdItenType",
    "nsuri": "http://sbrit.sec.gov/dei/2023",
    "localname": "EntityTaxIdentificationNumber",
    "presentation": [
    "http://bridgebio.com/role/DocumentAndEntity
         "http://-
j.mago":[
"acoust:[
"acous
              "auth_ref": [
"rl"
"sei Local RhomeNumber"; [
"sei Local RhomeNumber"; [
"shirtype"; "normal Leadstring TLearType",
"shortype"; "Local RhomeNumber",
"presentation";
"local RhomeNumber",
"presentation";
"local RhomeNumber",
"local RhomeNumber",
"local RhomeNumber Rh
              )
)
),
"auth_ref": []
    } dei NorradingSymbolFlag": {
    "sbritype": "trueItemType"
    "nauri": "http://kbrl.ase.gov/dsi/2023",
    "localname": "NoTradingSymbolFlag",
    "presentation": [
    "http://bridgebio.com/role/DocumentAndE
         ],
"lang": [
"en-us": [
"role": [
"label": "Wo Trading Symbol Flag",
"documentation": "Boclean flag that is true only for a
              )
),
"auth_ref": []
         ].

"dai, PreCommencementIssuerTenderOffer": {
    "xbritype": "booleanttenType",
    "msuri": "http://bxri.sec.gov/dai/2023",
    "localname": "PreCommencementIssuerTenderOffer",
    "presentation": {
        "http://bridgebio.com/role/DocumentAndEntityIn: 1,
        "http:/
              "auth_ref": [
         )
)
),
"auth_ref": [
"r4"
"http://
"lang": {
    "en-us" {
        "en-us" {
          "en-us" the continue of the conti
              )
),
"auth_ref": [
"r0"
    "Thispy:"--
"lang": [
"Genue": [
"Genue": [
"Annie": Security Exchange Hame",
"Labal": "Security Exchange Hame",
"documentation": "Hame of the Exchange on which a security is registered."
         ]
|,
|auth_ref": [
|"r2"
         "REEST: (NA LANGE AND ADDRESS 
              "auth_ref": [
              },

doi TradingSymbol": {
   "wholtype": "tradingSymbolItenType",
   "mauri": "bhip://whol.sec.gov/doi/2023",
   "localname": "TradingSymbol",
   "presentation";
   "bridgebio.com/role/DocumentAndEntityInformatii.
```

```
"Jack", "Tabling Symbol",
"Schools of the Internance as listed on an enchange."
"School of the Internance as listed on an enchange."
"School of the Internance as listed on an enchange."
"School of the Internance as listed on an enchange."
"School of the Internance as listed on an enchange."
"School of Total of Total
```